

PROBATION DEPARTMENT

REQUEST FOR PROPOSALS FOR

COMPREHENSIVE ELECTRONIC MONITORING SERVICES

FEBRUARY 24, 2025

Prepared By County of Los Angeles

RFP # 6402501

INTR	ODUCTION	1
PURF	POSE-AGREEMENT FOR COMPREHENSIVE ELECTRONIC	3
MON	ITORING SERVICES	1
2.1	Statement of Work	1
2.2	Sample Agreement: County Terms and Conditions	1
PROF	POSER'S MINIMUM MANDATORY QUALIFICATIONS	3
coul	NTY'S RIGHTS AND RESPONSIBILITIES	4
4.1	Representations Made Prior to Contract Execution	4
4.2	Final Contract Award by the Board of Supervisors	4
4.3	County's Option to Reject Proposals	4
4.4	County's Right to Amend Request for Proposals	4
4.5	Background and Security Investigations	5
4.6	County's Quality Assurance Plan	3
PROF	POSER'S REQUIREMENTS AND CERTIFICATIONS	6
5.1	Notice to Proposers Concerning the Public Records Act	3
5.2	Contact with County Personnel	7
5.3	Mandatory Requirement to Register on County's WebVen	3
5.4	Protest Policy Review Process	3
5.5	Injury and Illness Prevention Program	9
5.6	Confidentiality and Independent Contractor Status	9
5.7	Conflict of Interest	9
5.8	Determination of Proposer Responsibility	9
5.9	Proposer Debarment10	Э
5.10	Adherence to County's Child Support Compliance Program	2
5.11	Gratuities1	2
5.12	Notice to Proposers Regarding the County Lobbyist Ordinance1	3
	PURI MONI 2.1 2.2 PROI 4.1 4.2 4.3 4.4 4.5 4.6 PROI 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11	MONITORING SERVICES 2.1 Statement of Work 2.2 Sample Agreement: County Terms and Conditions PROPOSER'S MINIMUM MANDATORY QUALIFICATIONS COUNTY'S RIGHTS AND RESPONSIBILITIES 4.1 Representations Made Prior to Contract Execution 4.2 Final Contract Award by the Board of Supervisors 4.3 County's Option to Reject Proposals 4.4 County's Right to Amend Request for Proposals 4.5 Background and Security Investigations 4.6 County's Quality Assurance Plan PROPOSER'S REQUIREMENTS AND CERTIFICATIONS 5.1 Notice to Proposers Concerning the Public Records Act 5.2 Contact with County Personnel 5.3 Mandatory Requirement to Register on County's WebVen 5.4 Protest Policy Review Process 5.5 Injury and Illness Prevention Program 5.6 Confidentiality and Independent Contractor Status 5.7 Conflict of Interest 5.8 Determination of Proposer Responsibility 5.9 Proposer Debarment 5.10 Adherence to County's Child Support Compliance Program 125.11 Gratuities 126

	5.13	Federal Earned Income Credit	. 14
	5.14	Consideration of GAIN-START Participants for Employment	. 14
	5.15	Recycled Bond Paper	. 14
	5.16	Jury Service Program	. 14
	5.17	Intentionally Omitted	. 16
	5.18	Notification to County of Pending Acquisitions/Mergers by Proposing Company	. 16
	5.19	Proposer's Charitable Contributions Compliance	. 16
	5.20	Defaulted Property Tax Reduction Program	. 18
	5.21	Time Off for Voting	. 18
	5.22	Proposer's Acknowledgement of County's Commitment to Zero Toleran Policy on Human Trafficking	
	5.23	Intentionally Omitted	. 19
	5.24	Default Method of Payment: Direct Deposit or Electronic Funds Transfe (EFT)	
	5.25	Proposer's Acknowledgement of County's Commitment to Fair Chance Employment Hiring Practices	
	5.26	Prohibition from Participation in Future Solicitation(s)	. 20
	5.27	Contribution and Agent Declaration	. 20
6	COUN	NTY'S PREFERENCE PROGRAMS	. 21
	6.1	Overview of County's Preference Programs	. 21
	6.2	Local Small Business Enterprise (LSBE) Preference Program	. 22
	6.3	Local Small Business Enterprise (LSBE) Prompt Payment Program	. 22
	6.4	Social Enterprise (SE) Preference Program	. 23
	6.5	Disabled Veteran Business Enterprise (DVBE) Preference Program	. 23
7	PROF	POSAL SUBMISSION REQUIREMENTS	. 24
	7.1	Proposal Submission Information	. 24

	7.2	Truth and Accuracy of Representations	24
	7.3	RFP Timetable	25
	7.4.	Solicitation Requirements Review	25
	7.5	Proposers' Questions	26
	7.6	Intentionally Omitted	26
	7.7	Virtual Proposers' Conference	27
	7.8	Preparation of the Proposal	27
	7.9	Business Proposal Format	27
	7.10	Cost Proposal Format	35
	7.11	Firm Offer-Withdrawal of Proposal	35
	7.12	Proposal Submission	36
8	SELE	ECTION PROCESS AND EVALUATION CRITERIA	37
	8.1	Selection Process	37
	8.2	Adherence to Minimum Mandatory Requirements (Pass-Fail)	37
	8.3	Disqualification Review	38
	8.4	Business Proposal Evaluation and Criteria (70%)	38
	8.5	Cost Proposal Evaluation Criteria (30%)	40
	8.6	Intentionally Omitted	40
	8.7	Department's Proposed Contractor Selection Review	40
	8.8	County Independent Review Process	42

APPENDICES:

- A Statement of Work: Explains in detail the required services to be performed by the Contractor.
- **B** Statement of Work Technical Exhibits: Attachments which accompany the Statement of Work.
- **C** Sample Contract: Identifies the terms and conditions in the Contract.
- **D** Required Forms: Forms that must be completed and included in the proposal.
- E RFP Transmittal Form to Request a Solicitation Requirements Review: Transmittal sent to Department requesting a Solicitation Requirements Review.
- F County of Los Angeles Policy on Doing Business with Small Business: County Policy
- G Jury Service Ordinance: County Code
- **H** Listing of Contractors Debarred in Los Angeles County: Contractors who are not allowed to Contract with the County for a specific length of time.
- IRS Notice 1015: Provides information on Federal Earned Income Credit.
- **J-L** Intentionally Omitted
- M Background and Resources: California Charities Regulation: An information sheet intended to assist nonprofit agencies with compliance with SB 1262-the Nonprofit Integrity Act of 2004 and identify available resources.
- N Defaulted Property Tax Reduction Program: County Code

1 INTRODUCTION

The Los Angeles County Department of Probation (Probation) is issuing this Request for Proposals (RFP) to solicit proposals for a Contract with an organization who can provide Comprehensive Electronic Monitoring Services (CEMS) on a twenty-four (24) hour continuous basis for Probation's Global Positioning System Monitoring Program (GPSMP), Juvenile Electronic Monitoring Program (JEMP), and Adult Electronic Monitoring Program (Adult EMP).

2 PURPOSE-AGREEMENT FOR COMPREHENSIVE ELECTRONIC MONITORING SERVICES

2.1 Statement of Work

The Contractor shall provide CEMS on a twenty-four (24) hour continuous basis for the County of Los Angeles Probation Department's GPSMP, JEMP, and Adult EMP. The Contractor shall provide all technology, equipment, consumables, systems and related support services, cellular wireless services, accessories, tracking services, data storage, local law enforcement crime scene incident data collection and storage, crime scene correlation mapping analysis and reporting, point pattern analysis, monitoring center support services, training, user manuals, consultation, on-site assistance, and problem analysis not limited to additional support services. The Contractor's administrative business office, monitoring services and data center facilities must be located within the United States.

The Contractor shall be expected to implement the Statement of Work contained in Appendix A (Statement of Work) of this RFP.

2.2 Sample Agreement: County Terms and Conditions

2.2.1 The Contractor shall be expected to implement the requirements outlined in Appendix C (Sample Contract) of this RFP.

2.2.2 Anticipated Contract Term

The resulting Contract term shall be for a five (5) year period, unless terminated or extended in whole or in part, as provided in the Contract. Contingent upon available funding, the Contract may be extended by the Chief Probation Officer and the authorized official of the Contractor, by mutual written agreement, for up to two (2) additional and consecutive thirty (30) month periods for a total maximum Contract term of ten (10) years.

Contingent upon available funding, the term of the Contract may also be extended beyond the stated expiration date on a month-to-month basis, for a period of time not to exceed six (6) months, upon the written request of the Chief Probation Officer and the written concurrence of the Contractor. All terms of the Contract in effect at the time of extending the term shall remain in effect for the duration of the extension.

2.2.3 Contract Rates

The Contractor's rates shall remain firm and fixed for the term of the Contract.

Thereafter, if requested by the Contractor, the contract (hourly, daily, monthly, etc.) amount may, at the sole discretion of the County be increased annually based on the most recently published percentage change in the U.S. Department of Labor, Bureau of Labor Statistics' Consumer Price Index for Urban Consumers (CPI-U) for the Los Angeles-Long Beach-Anaheim Area for the twelve (12) month period preceding the contract anniversary date, which will be the effective date for any Cost of Living Adjustment (COLA). However, any increase must not exceed the general salary movement granted to County employees as determined by the Chief Executive Office as of each July 1 for the prior twelve (12) month period. Furthermore, should fiscal circumstances ultimately prevent the Board from approving any increase in County employee salaries, no COLA will be granted. Where the County decides to grant a COLA pursuant to this Paragraph for living wage contracts, it may, in its sole discretion exclude the cost of labor (including the cost of wages and benefits paid to employees providing services under this Contract) from the base upon which a COLA is calculated, unless the Contractor can show that their labor cost will actually increase. Further, before any COLA increase will take effect and become part of the contract, it will require a written amendment to the contract first, that has been formally approved and executed by the parties.

2.2.4 Days of Operation

Contractor shall be required to provide Comprehensive Electronic Monitoring Services on a twenty-four (24) hours continuous basis.

2.2.5 Indemnification and Insurance

The Contractor shall be required to comply with the provisions contained in Paragraph 8.23 (Indemnification) of Appendix C

(Sample Contract). The Contractor shall procure, maintain, and provide to the County proof of insurance coverage for all the programs of insurance along with associated amounts specified in Paragraph 8.24 (General Provisions for all Insurance Coverage) and Paragraph 8.25 (Insurance Coverage) of Appendix C (Sample Contract).

2.2.6 Intentionally Omitted

3 PROPOSER'S MINIMUM MANDATORY QUALIFICATIONS

- 3.1 Interested and qualified Proposers that can demonstrate their ability and qualifications to successfully provide the required services outlined in Appendix A (Statement of Work) are invited to submit a proposal, provided they meet the following mandatory requirements:
 - 3.1.1 The Proposer must submit a proposal by **April 10, 2025, 12:00** p.m., **PT.**
 - 3.1.2 The Proposer must demonstrate that the agency has a minimum of three (3) years' experience within the last five (5) years providing services in the area of electronic monitoring, including GPS technology, transdermal alcohol monitoring and breath alcohol monitoring for a population of five hundred (500) or more participants.
 - 3.1.3 The Proposer must have an administrative business office located within the United States of America. The address to the administrative business office must be included in Exhibit 1 (Proposer's Organization Questionnaire/Affidavit) of Appendix D (Required Forms).
 - 3.1.4 If the Proposer's compliance with a County Contract has been reviewed by the Department of the Auditor-Controller within the last 10 years, the Proposer must not have unresolved questioned costs identified by the Auditor-Controller, in an amount over \$100,000.00, that are confirmed to be disallowed costs by the contracting County department, and remain unpaid for six months or more from the date of disallowance, unless such disallowed costs are the subject of current good faith negotiations to resolve the disallowed costs, in the opinion of the County.

4 COUNTY'S RIGHTS AND RESPONSIBILITIES

4.1 Representations Made Prior to Contract Execution

4.1.1 The County is not responsible for representations made by any of its officers or employees prior to the execution of the Contract unless such understanding or representation is included in the Contract.

4.2 Final Contract Award by the Board of Supervisors

4.2.1 Notwithstanding a recommendation of a Department, agency, individual, or other, the Board of Supervisors retains the right to exercise its judgment concerning the selection of a proposal and the terms of any resultant agreement, and to determine which proposal best serves the interests of the County. The Board is the ultimate decision-making body and makes the final determinations necessary to arrive at a decision to award, or not award, a Contract.

4.3 County's Option to Reject Proposals

4.3.1 The Proposers are hereby advised that this RFP is a solicitation for proposals only, and is not intended, and is not to be construed as, an offer to enter into a Contract or as a promise to engage in any formal competitive bidding or negotiations pursuant to any statute, ordinance, rule, or regulation. The County may, at its sole discretion, reject any or all proposals submitted in response to this RFP or may, in its sole discretion, reject all proposals and cancel this RFP in its entirety. The County shall not be liable for any costs incurred by the Proposer in connection with the preparation and submission of any proposal. The County reserves the right to waive inconsequential disparities in a submitted proposal.

4.4 County's Right to Amend Request for Proposals

4.4.1 The County has the right to amend the RFP by written addendum. The County is responsible only for that which is expressly stated in the solicitation document and any authorized written addenda thereto. Such addendum shall be made available to each person or organization which County records indicate has received this RFP. Should such addendum require additional information not previously requested, failure to address the requirements of such addendum may result in the proposal being found non-responsive and not being considered, as determined in the sole discretion of the County. The County is not responsible for and shall not be

bound by any representations otherwise made by any individual acting or purporting to act on its behalf.

4.5 Background and Security Investigations

- 4.5.1 Background and security investigations of the Contractor's staff are required as a condition of beginning and continuing work under the resulting Contract. The cost of background checks is the responsibility of the Contractor. The Contractor shall be responsible for the ongoing implementation and monitoring of Subparagraphs 4.5.1.1 through 4.5.1.6. On at least a quarterly basis, the Contractor shall report, in writing, monitoring results to the County, indicating compliance or problem areas. Elements of monitoring report shall receive prior written approval from the County.
 - 4.5.1.1 The Contractor shall submit the names of the Contractor's or subcontractor's employees to the County's Program Manager prior to the employee starting work on this Contract. The County will schedule appointments to conduct background investigation/record checks based on fingerprints of the Contractor's or subcontractor's employees. The County shall have the right to conduct background investigations of the Contractor's or subcontractor's employees at any time. The Contractor's or the subcontractor's employees shall not begin work on this Contract before receiving written notification of clearance from the County.
 - 4.5.1.2 No personnel employed by the Contractor or the subcontractor for this service having access to Probation information or records shall have a criminal conviction record or pending criminal trial unless such information has been fully disclosed to the County and employment of the employee for this service is approved in writing by the County.
 - 4.5.1.3 The County reserves the right, in its sole discretion, to preclude the Contractor or the subcontractor from employment or continued employment of any individual performing services under this Contract.
 - 4.5.1.4 No Contractor or subcontractor staff providing services under this Contract shall be on active probation or parole.

- 4.5.1.5 The Contractor or the subcontractor staff performing services under this Contract shall be under a continuing obligation to disclose any prior or subsequent criminal conviction record or any pending criminal trial to the County.
- 4.5.1.6 Because the County is charged by the State for checking the criminal records of the Contractor's or the subcontractor's employees, the County will bill the Contractor to recover these expenses. The current amount is forty-nine dollars (\$49.00) per record check, which is subject to change by the State.

4.6 County's Quality Assurance Plan

4.6.1 After Contract award, the County or its agent will monitor the Contractor's performance under the Contract on a periodic basis. Such monitoring will include assessing the Contractor's compliance with all terms and conditions in the Contract and performance standards identified in Appendix A (Statement of Work). The Contractor's deficiencies which the County determines are significant or continuing and that may jeopardize performance of the Contract will be reported to the County's Board of Supervisors. The report will include improvement/corrective action measures taken by the County and the Contractor. If improvement does not occur consistent with the corrective action measures, the County may terminate the Contract in whole or in part, or impose other penalties as specified in the Contract.

5 PROPOSER'S REQUIREMENTS AND CERTIFICATIONS

5.1 Notice to Proposers Concerning the Public Records Act

5.1.1 Responses to this solicitation shall become the exclusive property of the County. Absent extraordinary circumstances, the recommended Proposer's proposal will become a matter of public record when 1) Contract negotiations are complete; 2) Probation receives a letter from the recommended Proposer's authorized officer that the negotiated Contract is the firm offer of the recommended Proposer; and 3) Probation releases a copy of the recommended Proposer's proposal in response to a Notice of Intent to Request a Proposed Contractor Selection Review under Board Policy No. 5.055 (Services Contract Solicitation Protest).

Notwithstanding the above, absent extraordinary circumstances, all proposals will become a matter of public record when the

Department's Proposer recommendation appears on the Board agenda.

Exceptions to disclosure are those parts or portions of all proposals that are justifiably defined as business or trade secrets, and plainly marked by the Proposer as "Trade Secret," "Confidential," or "Proprietary."

- 5.1.2 The County shall not, in any way, be liable or responsible for the disclosure of any such record or any parts thereof, if disclosure is required or permitted under the California Public Records Act or otherwise by law. A blanket statement of confidentiality or the marking of each page of the proposal as confidential shall not be deemed sufficient notice of exception. The Proposers must specifically label only those provisions of their respective proposal which are "Trade Secrets," "Confidential," or "Proprietary," in nature.
- 5.1.3 In the event the County is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "Confidential," "Trade Secrets," or "Proprietary," the Proposer agrees to defend and indemnify the County from all costs and expenses, including reasonable attorneys' fees, incurred in connection with any action, proceedings, or liability arising in connection with the Public Records Act request.

5.2 Contact with County Personnel

5.2.1 All contact regarding this RFP or any matter relating thereto must be in writing and may be mailed, e-mailed or faxed as follows:

Oscar Rivas, Contract Analyst
County of Los Angeles Probation Department
Contracts & Grants Management Division
9150 East Imperial Highway, Room D-29
Downey, CA 90242

Email address: Oscar.Rivas@probation.lacounty.gov

Fax#: (562) 658-2307

If it is discovered that the Proposer contacted and received information from any County personnel, other than the person specified above, regarding this solicitation, the County, in its sole determination, may disqualify their proposal from further consideration.

5.3 Mandatory Requirement to Register on County's WebVen

5.3.1 Prior to a Contract award, all potential Contractors must register in the County's WebVen. WebVen contains the vendor's business profile and identifies the goods/services the business provides. Registration can be accomplished online via the Internet by accessing the County's home page at:

http://camisvr.co.la.ca.us/webven/

5.4 Protest Policy Review Process

- 5.4.1 Under Board Policy No. 5.055 (Services Contract Solicitation Protest), any prospective Proposer may request a review of the requirements under a solicitation for a Board-approved services Contract, as described in Subparagraph 5.4.3 (Grounds for Review) below. Additionally, any actual Proposer may request a review of a disqualification or of a proposed Contract award under such a solicitation, as described respectively in the paragraphs below. It is the responsibility of the Proposer challenging the decision of a County Department to demonstrate that the Department committed a sufficiently material error in the solicitation process to justify invalidation of a proposed Contract award.
- 5.4.2 Throughout the review process, the County has no obligation to delay or otherwise postpone an award of Contract based on a Proposer protest. In all cases, the County reserves the right to make an award when it is determined to be in the best interest of the County of Los Angeles to do so.

5.4.3 Grounds for Review

Unless state or federal statutes or regulations otherwise provide, the grounds for review of a solicitation for a Board-approved services Contract provided for under Board Policy No. 5.055 (Services Contract Solicitation Protest) are limited to the following:

- 5.4.3.1 Paragraph 7.4 Solicitation Requirements Review
- 5.4.3.2 Paragraph 8.3 Disqualified Review
- 5.4.3.3 Paragraph 8.7 Department's Proposed Contractor Selection Review

5.5 Injury and Illness Prevention Program

5.5.1 The Contractor shall be required to comply with the State of California's Cal OSHA's regulations. California Code of Regulations Title 8 Section 3203 requires all California employers to have a written, effective Injury and Illness Prevention Program (IIPP) that addresses hazards pertaining to the particular workplace covered by the program.

5.6 Confidentiality and Independent Contractor Status

5.6.1 As appropriate, the Contractor shall be required to comply with Paragraph 7.6 (Confidentiality) and Paragraph 8.22 (Independent Contractor Status), contained in Appendix C (Sample Contract).

5.7 Conflict of Interest

5.7.1 No County employee whose position in the County enables him/her to influence the selection of a Contractor for this RFP, or any competing RFP, nor any spouse of economic dependent of such employees, shall be employed in any capacity by a Proposer or have any other direct or indirect financial interest in the selection of a Contractor. The Proposer shall certify that he/she is aware of and has read Section 2.180.010 of the Los Angeles County Code as stated in Exhibit 5 (Certification of No Conflict of Interest) of Appendix D (Required Forms).

5.8 Determination of Proposer Responsibility

- 5.8.1 A responsible Proposer is a Proposer who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity and experience to satisfactorily perform the Contract. It is the County's policy to conduct business only with responsible Proposers.
- 5.8.2 The Proposers are hereby notified that, in accordance with Chapter 2.202 of the County Code, the County may determine whether the Proposer is responsible based on a review of the Proposer's performance on any Contracts, including but not limited to County Contracts. Particular attention will be given to violations of labor laws related to employee compensation and benefits, and evidence of false claims made by the Proposer against public entities. Labor law violations which are the fault of the subcontractors and of which the Proposer had no knowledge shall not be the basis of a determination that the Proposer is not responsible.

- 5.8.3 The County may declare a Proposer to be non-responsible for purposes of this Contract if the Board of Supervisors, in its discretion, finds that the Proposer has done any of the following: 1) violated a term of a Contract with the County or a nonprofit corporation created by the County; 2) committed an act or omission which negatively reflects on the Proposer's quality, fitness or capacity to perform a Contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same; 3) committed an act or omission which indicates a lack of business integrity or business honesty; or 4) made or submitted a false claim against the County or any other public entity.
- 5.8.4 If there is evidence that the apparent highest ranked Proposer may not be responsible, the Department shall notify the Proposer in writing of the evidence relating to the Proposer's responsibility, and its intention to recommend to the Board of Supervisors that the Proposer be found not responsible. The Department shall provide the Proposer and/or the Proposer's representative with an opportunity to present evidence as to why the Proposer should be found to be responsible and to rebut evidence which is the basis for the Department's recommendation.
- 5.8.5 If the Proposer presents evidence in rebuttal to the Department, the Department shall evaluate the merits of such evidence, and based on that evaluation, make a recommendation to the Board of Supervisors. The final decision concerning the responsibility of the Proposer shall reside with the Board of Supervisors.
- 5.8.6 These terms shall also apply to proposed subcontractors of the Proposers on County Contracts.

5.9 Proposer Debarment

5.9.1 The Proposer is hereby notified that, in accordance with Chapter 2.202 of the County Code, the County may debar the Proposer from bidding or proposing on, or being awarded, and/or performing work on other County Contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and the County may terminate any or all of the Proposer's existing Contracts with the County, if the Board of Supervisors finds, in its discretion, that the Proposer has done any of the following: 1) violated a term of a Contract with the County or a nonprofit corporation created by the County; 2) committed an act or omission which negatively reflects on the Proposer's quality, fitness or

- capacity to perform a Contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same; 3) committed an act or offense which indicates a lack of business integrity or business honesty; or 4) made or submitted a false claim against the County or any other public entity.
- 5.9.2 If there is evidence that the apparent highest ranked Proposer may be subject to debarment, the Department shall notify the Proposer in writing of the evidence which is the basis for the proposed debarment and shall advise the Proposer of the scheduled date for a debarment hearing before the Contractor Hearing Board.
- 5.9.3 The Contractor Hearing Board shall conduct a hearing where evidence on the proposed debarment is presented. The Proposer and/or the Proposer's representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative proposed decision, which shall contain a recommendation regarding whether the Proposer should be debarred, and, if so, the appropriate length of time of the debarment. The Proposer and the Department shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 5.9.4 After consideration of any objections, or if no objections are received, a record of the hearing, the proposed decision and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 5.9.5 If a Proposer has been debarred for a period longer than five (5) years, that Proposer may, after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the Proposer has adequately demonstrated one or more of the following: 1) elimination of the grounds for which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed; or 4) any other reason that is in the best interests of the County.
- 5.9.6 The Contractor Hearing Board will consider requests for review of a debarment determination only where 1) the Proposer has been debarred for a period longer than five (5) years; 2) the debarment

has been in effect for at least five (5) years; and 3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment and includes supporting documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.

- 5.9.7 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 5.9.8 These terms shall also apply to proposed subcontractors of the Proposers on County Contracts.
- 5.9.9 Appendix H (Listing of Contractors Debarred in Los Angeles County) provides a link to the County's website where there is a listing of Contractors that are currently on the Debarment List for Los Angeles County.

5.10 Adherence to County's Child Support Compliance Program

5.10.1 The Proposers shall: 1) fully comply with all applicable State and Federal reporting requirements relating to employment reporting for its employees; and 2) comply with all lawfully served Wage and Earnings Assignment Orders and Notice of Assignment and continue to maintain compliance during the term of any Contract that may be awarded pursuant to this solicitation. Failure to comply may be cause for termination of a Contract or initiation of debarment proceedings against the non-compliant Contractor (County Code Chapter 2.202).

5.11 Gratuities

5.11.1 Attempt to Secure Favorable Treatment

It is improper for any County officer, employee or agent to solicit consideration, in any form, from a Proposer with the implication,

suggestion or statement that the Proposer's provision of the consideration may secure more favorable treatment for the Proposer in the award of the Contract or that the Proposer's failure to provide such consideration may negatively affect the County's consideration of the Proposer's submission. A Proposer shall not offer or give either directly or through an intermediary, consideration, in any form, to a County officer, employee or agent for the purpose of securing favorable treatment with respect to the award of the Contract.

5.11.2 Proposer Notification to County

A Proposer shall immediately report any attempt by a County officer, employee or agent to solicit such improper consideration. The report shall be made either to the County Manager charged with the supervision of the employee or to the Los Angeles County Fraud Hotline at (800) 544-6861 or https://fraud.lacounty.gov/. Failure to report such a solicitation may result in the Proposer's submission being eliminated from consideration.

5.11.3 Form of Improper Consideration

Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

5.12 Notice to Proposers Regarding the County Lobbyist Ordinance

5.12.1 The Board of Supervisors of the County of Los Angeles has enacted an ordinance regulating the activities of persons who lobby County officials. This ordinance, referred to as the "Lobbyist Ordinance", defines a County Lobbyist and imposes certain registration requirements upon individuals meeting the definition. The complete text of the ordinance can be found in County Code Chapter 2.160. In effect, each person, corporation or other entity that seeks a County permit, license, franchise or Contract must certify compliance with the ordinance. As part of this solicitation process, it will be the responsibility of each Proposer to review the ordinance independently as the text of said ordinance is not contained within this RFP. Thereafter, each person, corporation or other entity submitting a response to this solicitation, must certify that each County Lobbyist, as defined by Los Angeles County Code Section 2.160.010, retained by the Proposer is in full compliance with Chapter 2.160 of the Los Angeles County Code and each such County Lobbyist is not on the Executive Office's List of Terminated Registered Lobbyists by completing and submitting Exhibit 6

(Familiarity with the County Lobbyist Ordinance Certification) of Appendix D (Required Forms), as part of their proposal.

5.13 Federal Earned Income Credit

The Contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Appendix I (IRS Notice 1015).

5.14 Consideration of GAIN-START Participants for Employment

As a threshold requirement for consideration for Contract award, the Proposers shall demonstrate a proven record of hiring participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) or Skills and Training to Achieve Readiness for Tomorrow (START) Programs or shall attest to a willingness to consider GAIN-START participants for any future employment openings if they meet the minimum qualifications for that opening. The Proposers shall attest to a willingness to provide employed GAIN-START participants access to the Proposers' employee mentoring program, if available, to assist these individuals in obtaining permanent employment and/or promotional opportunities.

The Proposers who are unable to meet this requirement shall not be considered for Contract award. The Proposers shall submit a completed Exhibit 9 (Attestation of Willingness to Consider GAIN-START Participants) of Appendix D (Required Forms), along with their proposal.

5.15 Recycled Bond Paper

5.15.1 The Proposer shall be required to comply with the County's policy on recycled bond paper as specified in Paragraph 8.39 (Recycled Bond Paper) of Appendix C (Sample Contract).

5.16 Jury Service Program

5.16.1 The prospective Contract is subject to the requirements of the County's Contractor Employee Jury Service Ordinance ("Jury Service Program") (Los Angeles County Code, Chapter 2.203). Prospective Contractors should carefully read Appendix G (Jury Service Ordinance) and Paragraph 8.8 (Compliance with the County's Jury Service Program) of Appendix C (Sample Contract), both of which are incorporated by reference into and made a part of

this RFP. The Jury Service Program applies to both the Contractors and their subcontractors.

Proposals that fail to comply with the requirements of the Jury Service Program will be considered non-responsive and excluded from further consideration.

- 5.16.2 The Jury Service Program requires the Contractors and their subcontractors to have and adhere to a written policy that provides that its employees shall receive from the Contractor, on an annual basis, no less than five (5) days of regular pay for actual jury service. The policy may provide that employees deposit any fees received for such jury service with the Contractor or that the Contractor deducts from the employee's regular pay the fees received for jury service. For purposes of the Jury Service Program, "employee" means any California resident who is a fulltime employee of a Contractor and "full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the County, or 2) the Contractor has a long-standing practice that defines the lesser number of hours as full-time. Therefore, the Jury Service Program applies to all of a Contractor's full-time California employees, even those not working specifically on the Full-time employees providing short-term, County project. temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program.
- 5.16.3 There are two (2) ways in which a Contractor might not be subject to the Jury Service Program. The first is if the Contractor does not fall within the Jury Service Program's definition of "Contractor". The Jury Service Program defines "Contractor" to mean a person, partnership, corporation of other entity which has a Contract with the County or a subcontract with a County Contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County Contracts or subcontracts. The second is if the Contractor meets one of the two exceptions to the Jury Service The first exception concerns small businesses and applies to the Contractors that have 1) ten (10) or fewer employees: and, 2) annual gross revenues in the preceding twelve (12) months which, if added to the annual amount of this Contract is less than five hundred thousand dollars (\$500,000), and, 3) is not an "affiliate or subsidiary of a business dominant in its field of operation." The second exception applies to the Contractors that possess a collective bargaining agreement that expressly supersedes the

provisions of the Jury Service Program. The Contractor is subject to any provision of the Jury Service Program not expressly superseded by the collective bargaining agreement.

5.16.4 If a Contractor does not fall within the Jury Service Program's definition of "Contractor" or if it meets any of the exceptions to the Jury Service Program, then the Contractor must so indicate in Exhibit 10 (Contractor Employee Jury Service Program Certification Form and Application for Exception) of Appendix D (Required Forms), and include with its submission all necessary documentation to support the claim such as tax returns or a collective bargaining agreement, if applicable. Upon reviewing the Contractor's application, the County will determine, in its sole discretion, whether the Contractor falls within the definition of Contractor or meets any of the exceptions to the Jury Service Program. The County's decision will be final.

5.17 Intentionally Omitted

5.18 Notification to County of Pending Acquisitions/Mergers by Proposing Company

5.18.1 The Proposer shall notify the County of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the Proposer is restricted from legally notifying the County of pending acquisitions/mergers, then it should notify the County of the actual acquisitions/mergers as soon as the law allows and provide to the County the legal framework that restricted it from notifying the County prior to the actual acquisitions/mergers. This information shall be provided by the Proposer Exhibit (Proposer's Organization in 1 Questionnaire/Affidavit) of Appendix D (Required Forms). Failure of the Proposer to provide this information may eliminate its proposal from any further consideration. The Proposer shall have a continuing obligation to notify the County and update any changes response in Exhibit 1 (Proposer's Questionnaire/Affidavit) of Appendix D (Required Forms) during the solicitation.

5.19 Proposer's Charitable Contributions Compliance

5.19.1 California's "Supervision of Trustees and Fundraisers for Charitable Purposes Act" regulates receiving and raising charitable contributions. Among other requirements, those subject to the Charitable Purposes Act must register. The 2004 Nonprofit Integrity Act (SB 1262, Chapter 919) increased Charitable

Purposes Act requirements. Prospective Contractors should carefully read Appendix M (Background and Resources: California Charities Regulation). New rules cover California public benefit corporations, unincorporated associations, and trustee entities and may include similar foreign corporations doing business or holding property in California. Key Nonprofit Integrity Act requirements affect executive compensation, fund-raising practices and documentation. Charities with over two million dollars (2,000,000) of revenues (excluding funds that must be accounted for to a governmental entity) have new audit requirements.

- 5.19.2 All Prospective Contractors must determine if they receive or raise charitable contributions which subject them to the Charitable Purposes Act and complete the certification form attached as Exhibit 19 (Charitable Contributions Certification) in Appendix D (Required Forms). A completed Exhibit 19 (Charitable Contributions Certification) of Appendix D (Required Forms) is a required part of any agreement with the County.
- 5.19.3 In Exhibit 19 (Charitable Contributions Certification) of Appendix D (Required Forms), prospective Contractors certify either that:
 - 5.19.3.1 They have determined that they do not now receive or raise charitable contributions regulated under the California Charitable Purposes Act, (including the Nonprofit Integrity Act) but will comply if they become subject to coverage of those laws during the term of a County agreement,

- OR -

- 5.19.3.2 They are currently complying with their obligations under the Charitable Purposes Act, attaching a copy of their most recent filing with the Registry of Charitable Trusts.
- 5.19.4 Prospective County Contractors that do not complete Exhibit 19 (Charitable Contributions Certification) of Appendix D (Required Forms) as part of the solicitation process may, in the County's sole discretion, be disqualified from Contract award. A County Contractor that fails to comply with its obligations under the Charitable Purposes Act is subject to either Contract termination or debarment proceedings or both. (County Code Chapter 2.202)

5.20 Defaulted Property Tax Reduction Program

5.20.1 The prospective Contract is subject to the requirements of the County's Defaulted Property Tax Reduction Program ("Defaulted Tax Program"), (Los Angeles County Code, Chapter 2.206). Prospective Contractors should carefully read Appendix N (Defaulted Property Tax Program Ordinance) and the pertinent provisions in Paragraph 8.51 (Warranty of Compliance with County's Defaulted Property Tax Reduction Program) and 8.52 (Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Tax Reduction Program) of Appendix C (Sample Contract), both of which are incorporated by reference into and made a part of this solicitation. The Defaulted Tax Program applies to both the Contractors and their subcontractors.

The Proposers shall be required to certify that they are in full compliance with the provisions of the Defaulted Tax Program and shall maintain compliance during the term of any Contract that may be awarded pursuant to this solicitation or shall certify that they are exempt from the Defaulted Tax Program by completing Exhibit 20 (Certification of Compliance with the County's Defaulted Property Tax Reduction Program) in Appendix D (Required Forms). Failure to maintain compliance, or to timely cure defects, may be cause for termination of a Contract or initiation of debarment proceedings against the non-compliant Contractor (Los Angeles County Code, Chapter 2.202).

Proposals that fail to comply with the certification requirements of the Defaulted Tax Program will be considered non-responsive and excluded from further consideration.

5.21 Time Off for Voting

5.21.1 The Contractor shall notify its employees and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every Contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

5.22 Proposer's Acknowledgement of County's Commitment to Zero Tolerance Policy on Human Trafficking

5.22.1 On October 4, 2016, the Los Angeles County Board of Supervisors approved a motion taking significant steps to protect victims of human trafficking by establishing a zero-tolerance policy on human trafficking. The policy prohibits the Contractors engaged in human trafficking from receiving Contract awards or performing services under a County Contract.

The Contractors are required to complete Exhibit 21 (Zero Tolerance Policy on Human Trafficking Certification) in Appendix D (Required Forms), certifying that they are in full compliance with the County's Zero Tolerance Policy on Human Trafficking provision as defined in Paragraph 8.54 (Compliance with County's Zero Tolerance Policy on Human Trafficking) of Appendix C (Sample Contract). Further, the Contractors are required to comply with the requirements under said provision for the term of any Contract awarded pursuant to this solicitation.

5.23 Intentionally Omitted

5.24 Default Method of Payment: Direct Deposit or Electronic Funds Transfer (EFT)

- 5.24.1 The County, at its sole discretion, has determined that the most efficient and secure default form of payment for goods and/or services provided under an agreement/contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.24.2 Upon Contract award or at the request of the A-C and/or the contracting department, the Contractor shall submit a direct deposit authorization request with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.
- 5.24.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.24.4 Upon Contract award or at any time during the duration of the agreement/contract, a Contractor may submit a written request for an exemption to this requirement. The A-C, in consultation with the

contracting department(s), shall decide whether to approve exemption requests.

5.25 Proposer's Acknowledgement of County's Commitment to Fair Chance Employment Hiring Practices

On May 29, 2018, the Los Angeles County Board of Supervisors approved a Fair Chance Employment Policy in an effort to remove job barriers for individuals with criminal records. The policy requires businesses that contract with the County to comply with fair chance employment hiring practices set forth in California Government Code Section 12952.

The Contractors are required to complete Exhibit 23 (Compliance with Fair Chance Employment Hiring Practices Certification) in Appendix D (Required Forms), certifying that they, and their subcontractors, are in full compliance with Section 12952, as indicated in the Sample Contract. Further, the Contractors are required to comply with the requirements under Section 12952 for the term of any Contract awarded pursuant to this solicitation.

5.26 Prohibition from Participation in Future Solicitation(s)

A Proposer, or a Contractor or its subsidiary or Subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification of the Proposer/Contractor from participation in the County solicitation or the termination or cancellation of any resultant County contract.

5.27 Contribution and Agent Declaration

Government Code Section 84308 requires a party to a contract proceeding to disclose any contribution of more than \$250 made to a County officer within the preceding twelve (12) months by the party or their agent. State regulations require this disclosure to be made at the time an application is filed, and, if a contribution is made during the contract proceeding, within 30 days of making a contribution or on the date on which the party first appears before or communicates with the agency regarding the proceeding after making the contribution, whichever is earliest. All Proposers are advised that they and all of their Subcontractors must complete and return as part of the proposal, the Contribution and Agent Declaration included in Exhibit 24 (Contribution and Agent

Declaration Form) of Appendix D (Required Forms). Proposers are further advised that they and their Subcontractors must update the Contribution and Agent Declaration Form throughout the pendency of the solicitation if a contribution is made after the initial disclosure when the proposal is submitted, and as requested at any time by the County prior to contract award. Failure by the Proposer or any Subcontractor(s) to complete and submit the required Contribution and Agent Declaration Form in Exhibit 24, and failure by the Proposer or any Subcontractor(s) to update the declaration as required by law or as otherwise requested by the County, may eliminate the proposal from further consideration and/or the Proposer may be disqualified from a contract award, as determined in the County's sole discretion. Further, all Proposers and their Subcontractors are prohibited under Government Code Section 84308 from making a contribution of more than \$250 to a County officer for twelve (12) months after the date a final decision is made in the contract proceeding involving this solicitation.

6 COUNTY'S PREFERENCE PROGRAMS

6.1 Overview of County's Preference Programs

- 6.1.1 The County of Los Angeles has three preference programs. The Local Small Business Enterprise (LSBE), Disabled Veterans Business Enterprise (DVBE), and Social Enterprise (SE). The Board of Supervisors encourages business participation in the County's contracting process by continually streamlining and simplifying our selection process and expanding opportunities for these businesses to compete for County opportunities.
- 6.1.2 The Preference Programs (LSBE, DVBE, and SE) requires that a business must complete certification prior to requesting a preference in a solicitation. This program and how to obtain certification are further explained in Paragraphs 6.2, 6.4, and 6.5 of this solicitation.
- 6.1.3 In no case shall the Preference Programs (LSBE, DVBE, and SE) price or scoring preference be combined with any other County preference program to exceed fifteen percent (15%) in response to any County solicitation.
- 6.1.4 Sanctions and financial penalties may apply to a business that knowingly, and with intent to defraud, seeks to obtain or maintain certification as a certified LSBE, DVBE, or SE when not qualified.
- 6.1.5 The County also has a Policy on Doing Business with Small Business that is stated in Appendix F.

6.2 Local Small Business Enterprise (LSBE) Preference Program

- 6.2.1 The County will give LSBE preference during the solicitation process to businesses that meet the definition of a LSBE, consistent with Chapter 2.204.030C.1 of the Los Angeles County Code. An LSBE is defined as a business: 1) certified by the State of California as a small business and has had its principal place of business located in Los Angeles County for at least one year; or 2) certified as a small business enterprise with other certifying agencies pursuant to the Department of Consumer and Business Affairs' (DCBA) inclusion policy that: a) has its principal place of business located in Los Angeles County, and b) has revenues and employee sizes that meet the State's Department of General Services requirements. The business must be certified by the Department of Consumer and Business Affairs as meeting the requirements set forth above prior to requesting the LSBE Preference in a solicitation.
- 6.2.2 To apply for certification as an LSBE, businesses should contact the Department of Consumer and Business Affairs at http://dcba.lacounty.gov
- 6.2.3 Certified LSBEs may only request the preference if the certification process has been completed and certification is affirmed. Businesses must complete and submit Exhibit 7 (Request for Preference Program Consideration) in Appendix D (Required Forms) and submit a letter of certification from the DCBA with their proposal.
- 6.2.4 Information about the State's small business enterprise certification regulations is in the California Code of Regulations, Title 2, Subchapter 8, Section 1896 et seq., and is also available on the California Department of General Services Office of Small Business Certification and Resources Website at https://caleprocure.ca.gov/pages/sbdvbe-index.aspx

6.3 Local Small Business Enterprise (LSBE) Prompt Payment Program

It is the intent of the County that Certified LSBEs receive prompt payment for services they provide to County Departments. Prompt payment is defined as fifteen (15) calendar days after receipt of an undisputed invoice.

6.4 Social Enterprise (SE) Preference Program

- 6.4.1 The County will give preference during the solicitation process to businesses that meet the definition of a SE, consistent with Chapter 2.205 of the Los Angeles County Code. A SE is defined as:
 - 1) A business that qualifies as a SE and has been in operation for at least one year (1) providing transitional or permanent employment to a Transitional Workforce or providing social, environmental and/or human justice services; and
 - 2) A business certified by the Department of Consumer and Business Affairs (DCBA) as a SE.
- 6.4.2 The DCBA shall certify that a SE meets the criteria set forth in Subparagraph 6.4.1.
- 6.4.3 Certified SEs may only request the preference if the certification has been completed and certification is affirmed. Businesses must complete and submit Exhibit 7 (Request for Preference Program Consideration) in Appendix D (Required Forms) and submit a letter of certification from the DCBA with their proposal.
- 6.4.4 Further information on SEs also available on the DCBA's website at: http://dcba.lacounty.gov

6.5 Disabled Veteran Business Enterprise (DVBE) Preference Program

- 6.5.1 The County will give preference during the solicitation process to businesses that meet the definition of a DVBE, consistent with Chapter 2.211 of the Los Angeles County Code. A DVBE vendor is defined as:
 - A business which is certified by the State of California as a DVBE; or
 - 2) A business which is verified as a service-disabled veteranowned small business (SDVOSB) by the Veterans Administration.
 - 3) A business certified as DVBE with other certifying agencies pursuant to the Department of Consumer and Business Affairs' (DCBA) inclusion policy that meets the criteria set forth by the agencies in 1 and 2 above.

- 6.5.2 The DCBA shall certify that a DVBE is currently certified by the State of California, by the U.S. Department of Veteran Affairs, or is determined by the DCBA's inclusion policy that meets the criteria set forth by the agencies in Section 6.5.1, 1 or 2 above.
- 6.5.3 Certified DVBEs may only request the preference if the certification process has been completed and certification is affirmed. Businesses must complete and submit Exhibit 7 (Request for Preference Program Consideration) in Appendix D (Required Forms) and submit a letter of certification from the DCBA with their proposal.
- 6.5.4 Information about the State's DVBE certification regulations is found in the California Code of Regulations, Title 2, Subchapter 8, Section 1896 et seq., and is also available on the California Department of General Services Office of Disabled Veteran Business Certification and Resources Website at https://caleprocure.ca.gov/pages/sbdvbe-index.aspx
- 6.5.5 Information on the Department of Veteran Affairs SDVOSB certification regulations is found in the Code of Federal Regulations, 38CFR 74 and is also available on the Department of Veterans Affairs Website at: https://www.va.gov/osdbu/

7 PROPOSAL SUBMISSION REQUIREMENTS

7.1 Proposal Submission Information

7.1.1 Section 7 (Proposal Submission Requirements) contains key project dates and activities as well as instructions to the Proposers in how to prepare and submit their proposal.

7.2 Truth and Accuracy of Representations

7.2.1 False, misleading, incomplete, or deceptively unresponsive statements in connection with a proposal shall be sufficient cause for rejection of the proposal. The evaluation and determination in this area shall be at the Director's sole judgment and his/her judgment shall be final. All proposals shall be firm and final offers and may not be withdrawn for a period of one hundred eighty (180) days following the final proposal submission date.

7.3 RFP Timetable

- 7.3.1 The timetable for this RFP is as follows:
 - Release of RFPFebruary 24, 2025.
 - Request for a Solicitation Requirements Review Due
 March 10, 2025, 12:00 p.m., PT.
 - Written Questions Due.......March 7, 2025, 12:00 p.m., PT.

 - Virtual Proposers' Conference...... March 18, 2025, 2:00 p.m., PT.
 - Proposals due by...... April 10, 2025, 12:00 p.m., PT.

7.4. Solicitation Requirements Review

- 7.4.1 Any person or entity may seek a Solicitation Requirements Review by submitting Appendix E (RFP Transmittal Form to Request a Solicitation Requirements Review) to the Department conducting the solicitation as described in Paragraph 7.4 (Solicitation Requirements Review). A request for a Solicitation Requirements Review may be denied, in the Department's sole discretion, if the request does not satisfy all of the following criteria:
 - 1. The request is made within the time frame identified in the solicitation document (generally within ten (10) business days of the issuance of the solicitation document):
 - 2. The request includes documentation (e.g., letterhead, business cards, etc.), which identifies the underlying authority of the person or entity to submit a proposal;
 - 3. The request itemizes in appropriate detail, each matter contested and factual reasons for the requested review; and
 - 4. The request asserts either that:
 - a. application of the minimum requirements, evaluation criteria and/or business requirements unfairly disadvantages the person or entity; or,

 due to unclear instructions, the process may result in the County not receiving the best possible responses from prospective Proposers.

The Solicitation Requirements Review shall be completed, and the Department's determination shall be provided to the requesting person or entity, in writing, within a reasonable time prior to the proposal due date. All Requests for Review should be submitted to:

Tasha Howard, Division Manager
County of Los Angeles Probation Department
Contracts & Grants Management Division
9150 East Imperial Highway, Room D-29
Downey, CA 90242
Fax #: (562) 658-2307

7.5 Proposers' Questions

7.5.1 The Proposers may submit written questions regarding this RFP by mail, fax or e-mail to the individual identified below. All questions must be received by **March 7, 2025, 12:00 p.m., PT.** All questions, without identifying the submitting company, will be compiled with the appropriate answers and issued as an addendum to the RFP.

When submitting questions, please specify the RFP section number, paragraph number, and page number and quote the language that prompted the question. This will ensure that the question can be quickly found in the RFP. The County reserves the right to group similar questions when providing answers.

Questions may address concerns that the application of minimum requirements, evaluation criteria and/or business requirements would unfairly disadvantage Proposers or, due to unclear instructions, may result in the County not receiving the best possible responses from the Proposer.

Questions should be addressed to:

Oscar Rivas, Contract Analyst
County of Los Angeles Probation Department
Contracts & Grants Management Division
9150 East Imperial Highway, Room D-29
Downey, CA 90242

Email address: Oscar.Rivas@probation.lacounty.gov

7.6 Intentionally Omitted

7.7 Virtual Proposers' Conference

- 7.7.1 A Virtual Proposers' Conference will be held to discuss the RFP. The County staff will respond to questions from potential Proposers.
- 7.7.2 Prospective Proposers shall R.S.V.P. to the contract analyst identified in 7.5.1 on or before **March 17, 2025, 12:00 p.m., PT.**The Proposers must provide the name, title, e-mail address, and phone number of the representatives who will be attending the Virtual Proposers' Conference. The County will be using Microsoft Teams live steaming video conferencing platform for this event. Prospective Proposers who R.S.V.P. for the conference will receive an emailed invitation with the Microsoft Teams meeting URL.

7.8 Preparation of the Proposal

Two (2) separate proposals must be submitted - a Business Proposal and a Cost Proposal. All proposals must be bound and submitted in the prescribed format. Any proposal that deviates from this format may be rejected as non-responsive without review at the County's sole discretion.

7.9 Business Proposal Format

7.9.1 The content and sequence of the proposal must be as follows:

- 7.9.1.1 Proposer's Organization Questionnaire/Affidavit and Required Support Documents for Corporations and Limited Liability Companies
- 7.9.1.2 Table of Contents
- 7.9.1.3 Executive Summary (Section A)
- 7.9.1.4 Proposer's Qualifications (Section B)
- 7.9.1.5 Proposer's Approach to Provide Required Services (Section C)
- 7.9.1.6 Proposer's Quality Control Plan (Section D)
- 7.9.1.7 Terms and Conditions in the Sample Contract, and Requirements of the Statement of Work (SOW): Acceptance of/or Exceptions to (Section E)
- 7.9.1.8 Business Proposal Required Forms (Section F)

7.9.1.9 Intentionally Omitted

7.9.2 Proposer's Organization Questionnaire/Affidavit and Required Support Documentation

The Proposer shall complete, sign and date Exhibits 1 and 1a (Proposer's Organization Questionnaire/Affidavit and CBE Information) in Appendix D (Required Forms). The person signing the form must be authorized to sign on behalf of the Proposer and to bind the applicant in a Contract.

Taking into account the structure of the Proposer's organization, the Proposer shall determine which of the below referenced supporting documents the County requires. If the Proposer's organization does not fit into one of these categories, upon receipt of the proposal or at some later time, the County may, in its discretion, request additional documentation regarding the Proposer's business organization and authority of individuals to sign Contracts.

If the below referenced documents are not available at the time of proposal submission, the Proposers must request the appropriate documents from the California Secretary of State and provide a statement on the status of the request.

7.9.3 Required Support Documents:

Corporations or Limited Liability Company (LLC):

The Proposer must submit the following documentation with the proposal:

- 1) A copy of a "Certificate of Good Standing" with the state of incorporation/organization.
- 2) A conformed copy of the most recent "Statement of Information" as filed with the California Secretary of State listing corporate officers or members and managers.

7.9.4 Limited Partnership:

The Proposer must submit a conformed copy of the Certificate of Limited Partnership or Application for Registration of Foreign Limited Partnership as filed with the California Secretary of State, and any amendments.

7.9.5 Table of Contents

List all material included in the proposal. Include a clear definition of the material, identified by sequential page numbers and by section and paragraph reference numbers.

7.9.6 Executive Summary (Section A)

Condense and highlight the contents of the Proposer's Business Proposal to provide the Probation Department with a broad understanding of the Proposer's approach, qualifications, experience, and staffing.

7.9.7 Proposer's Qualifications (Section B)

Demonstrate that the Proposer's organization has the experience and financial capability to perform the required services. The following sections must be included:

7.9.7.1 Proposer's Background and Experience (Section B.1)

Provide a summary of relevant background information to demonstrate that the Proposer meets the minimum requirement(s) stated in Section 3 (Proposer's Minimum Mandatory Qualifications) of this RFP and has the capability to perform the required services as a corporation or other entity.

In addition to the above, include a resume for the Project Director, demonstrating that the Project Director has a minimum of three (3) years' experience within the last five (5) years in the management and operation of electronic monitoring services or similar scope and is a current employee of the agency. The resume must include dates for the positions listed.

7.9.7.2 Proposer's References (Section B.2)

It is the Proposer's sole responsibility to ensure that the firm's name, and point of contact's name, title and phone number for each reference is accurate. The same references may be listed on both Exhibit 2 (Prospective Contractor References) and Exhibit 3 (Prospective Contractor List of Contracts) in Appendix D (Required Forms).

- 1. The County may disqualify a Proposer as non-responsive and/or non-responsible if:
 - a) references fail to substantiate the Proposer's description of the services provided; or
 - b) references fail to support that the Proposer has a continuing pattern of providing capable, productive and skilled personnel; or
 - c) the Department is unable to reach the point of contact with reasonable effort. It is the Proposer's responsibility to inform the point of contact of normal working hours.
- 2. The Proposer must complete and include the following Required Forms:
 - a) Exhibit 2 (Prospective Contractor References) in Appendix D (Required Forms). The Proposer must provide three (3) references where the same or similar scope of services was provided. If more than three (3) references are provided, only the first three (3) listed will be contacted.
 - b) Exhibit 3 (Prospective Contractor List of Contracts) in Appendix D (Required Forms). The listing must include all Public Entities Contracts for the last three (3) years. Use additional sheets if necessary.
 - c) Exhibit 4 (Prospective Contractor List of Terminated Contracts) in Appendix D (Required Forms). Listing must include Contracts terminated within the past three (3) years with a reason for termination.

7.9.7.3 Proposer's Financial Capability (Section B.3)

Provide copies of the company's most current and prior two (2) fiscal years (for example 2023 and 2022) financial statements. Statements should include the company's assets, liabilities and net worth and at a minimum should include the Balance Sheet. Statement of Income, and the

Statement of Cash Flows. It should be noted that depending on the nature of the entity, i.e., for-profit, non-profit, governmental, the title of these statements may differ. For example, for a non-profit entity the Balance Sheet is referred to as the Statement of Financial Position. If audited statements are available, these should be submitted to meet this requirement. Do not submit Income Tax Returns to meet this requirement. Financial statements will be kept confidential if so stamped on each page.

7.9.7.4 Proposer's Pending Litigation and Judgments (Section B.4)

Identify by name, case and court jurisdiction any pending litigation in which the Proposer is involved, or judgments against the Proposer in the past five (5) years. Provide a statement describing the size and scope of any pending or threatening litigation against the Proposer or principals of the Proposer.

7.9.8 Proposer's Approach to Provide Required Services (Section C)

The Proposer's Approach to Provide Required Services section of the Business Proposal shall present a description of the techniques that the firm will employ in meeting the objectives described in Appendix A (Statement of Work). This section shall be a comprehensive plan to show how the Proposer will provide the services requested by the County including, but not limited to the requirements in Section 2.0 (Specific Tasks) of Appendix A, (Statement of Work) of this RFP as follows:

Section C1: Scope of Work (SOW Section 1.0)

Section C1: Monitoring Services (SOW Section 1.1)

Section C2: Monitoring Center Requirements (SOW Section 1.2)

Section C3: System Software Requirements (SOW Section 1.3)

Section C4: Mapping Capability (SOW Section 1.4)

Section C5: Crime Scene Correlation Mapping Analysis

Requirements (SOW Section 1.5)

Section C6: Point Pattern Analysis Requirements (SOW Section 1.6)

Section C7: Reporting Requirements (SOW Section 1.7)

Section C8: Data Documentation Requirements (SOW Section 1.8)

Section C9: Data Support Requirements (SOW Section 1.9)

Section C10: Data Center Requirements (SOW Section 1.10)

Section C11: Data Access/Storage Requirements (SOW Section 1.11)

Section C12: Local Law Enforcement Crime Scene Incident Data Collection Requirements (SOW Section 1.12)

Section C13: Crime Scene Correlation Mapping Analysis Requirements (SOW Section 1.13)

Section C14: Training and Consultation Requirements (SOW Section 1.14)

Section C15: Global Positioning System Monitoring Program (SOW Section 2.1)

Section C16: Juvenile Electronic Monitoring Program (SOW Section 2.2)

Section C17: Adult Electronic Monitoring Program (SOW Section 2.3)

Section C18: Implementation Services (SOW Section 2.4)

Reference to or repetition of scope, objectives, and requirements from this RFP does not constitute a "good understanding" of the project. Complete, yet concise, supplementary procedures, methods, explanations and descriptions are also required to make possible the County's evaluation as to the Proposer's understanding.

7.9.9 Proposer's Quality Control Plan (Section D)

Present a comprehensive Quality Control Plan to be utilized by the Proposer as a self-monitoring tool to ensure the required services

are provided as specified in Section 3.0, Quality Control Plan, Appendix A (Statement of Work) and the Performance Requirements Summary Chart in Appendix B (Statement of Work Technical Exhibits).

The following factors may be included in the plan:

- 7.9.9.1 Activities to be monitored to ensure compliance with all Contract requirements;
- 7.9.9.2 Monitoring methods to be used;
- 7.9.9.3 Frequency of monitoring;
- 7.9.9.4 Samples of forms to be used in monitoring;
- 7.9.9.5 Title/level and qualifications of personnel performing monitoring functions; and
- 7.9.9.6 Documentation methods of all monitoring results, including any corrective action taken.

7.9.10 Intentionally Omitted

- 7.9.11 Terms and Conditions in the Sample Contract, and Requirements of the Statement of Work (SOW): Acceptance of/or Exceptions to (Section E)
 - 7.9.11.1 It is the duty of every Proposer to thoroughly review Sample Contract and Statement of Work to ensure compliance with all terms, conditions and requirements. It is the County's expectation that in submitting a proposal the Proposers will accept, as stated, the County's terms and conditions in the Sample Contract and the County's requirements in Statement of Work. However, the Proposers are provided the opportunity to take exceptions to the County's terms, conditions, and requirements.
 - 7.9.11.2 Section E of the Proposer's response must include:
 - A statement offering the Proposer's acceptance of or exceptions to all terms and conditions listed in Appendix C (Sample Contract);

2. A statement offering the Proposer's acceptance of or exceptions to all requirements listed in Appendix A (Statement of Work); and

For each exception, the Proposer shall provide:

- 1. An explanation of the reason(s) for the exception;
- 2. The proposed alternative language; and
- 3. A description of the impact, if any, to the Proposer's price.
- 7.9.11.3 Indicate all exceptions to the Sample Contract and/or the Statement of Work by providing a 'red-lined' version of the language in question. The County relies on this procedure and any Proposer who fails to make timely exceptions as required herein, may be barred, at the County's sole discretion, from later making such exceptions.

The County reserves the right to determine if the Proposers' exceptions are material enough to deem the proposal non-responsive and not subject to further evaluation.

The County reserves the right to make changes to the Sample Contract and its appendices and exhibits at its sole discretion.

7.9.12 Business Proposal Required Forms (Section F)

Proposal shall include all completed, signed, and dated forms identified in Appendix D (Required Forms).

Exhibit 1	Proposer's Organization Questionnaire /Affidavit		
Exhibit 1a	Community Business Enterprise (CBE) Information		
Exhibit 2	Prospective Contractor References		
Exhibit 3	Prospective Contractor List of Contracts		
Exhibit 4	Prospective Contractor List of Terminated Contracts		
Exhibit 5	Certification of No Conflict of Interest		
Exhibit 6	Familiarity with the County Lobbyist Ordinance Certification		

Exhibit 7	Request for Preference Program Consideration		
Exhibit 8	Proposer's EEO Certification		
Exhibit 9	Attestation of Willingness to Consider GAIN-START Participants		
Exhibit 10	Contractor Employee Jury Service Program Certification Form and Application for Exception		
Exhibits 11-12	Cost Forms, included in the Cost Proposal		
Exhibits 13-18	Intentionally Omitted		
Exhibit 19	Charitable Contributions Certification		
Exhibit 20	Certification of Compliance with the County's Defaulted Property Tax Reduction Program		
Exhibit 21	Zero Tolerance Policy on Human Trafficking Certification		
Exhibit 22	Intentionally Omitted		
Exhibit 23	Compliance with Fair Chance Employment Hiring Practices Certification		
Exhibit 24	Contribution and Agent Declaration Form		

7.9.13 Intentionally Omitted

7.10 Cost Proposal Format

- 7.10.1 The content and sequence of the proposal must be as follows:
 - 7.10.1.1 Cover Page identifying, at a minimum, the RFP and the Proposer's name.
 - 7.10.1.2 Exhibit 11 (Pricing Sheet)
 - 7.10.1.3 Exhibit 12 (Certification of Independent Price Determination and Acknowledgement of RFP Restrictions) in Appendix D (Required Forms)
 - 7.10.1.4 Exhibit 13 Intentionally Omitted
 - 7.10.1.5 Exhibit 14 Intentionally Omitted

7.11 Firm Offer-Withdrawal of Proposal

7.11.1 Until the proposal submission deadline, errors in proposals may be corrected by a request in writing to withdraw the proposal and by submission of another set of proposals with the mistakes corrected.

Corrections will not be accepted once the deadline for submission of proposals has passed.

7.12 Proposal Submission

7.12.1 The original Business Proposal, six (6) paper copies shall be enclosed in a sealed envelope or box, plainly marked in the upper left-hand corner with the name and address of the Proposer and bear the words:

"BUSINESS PROPOSAL FOR COMPREHENSIVE ELECTRONIC MONITORING SERVICES"

The original Cost Proposal, six (6) paper copies must be submitted in a separate sealed package, plainly marked in the upper left-hand corner with the name and address of the Proposer and bear the words:

"COST PROPOSAL FOR COMPREHENSIVE ELECTRONIC MONITORING SERVICES"

The proposal(s) shall be delivered or mailed to:

County of Los Angeles Probation Department 9150 East Imperial Highway, Room D-29 Downey, CA 90242 Attention: Oscar Rivas, Contract Analyst

It is the sole responsibility of the submitting Proposer to ensure that its proposal is received before the submission deadline. Submitting Proposers shall bear all risks associated with delays in delivery by any person or entity, including the U.S. Mail. Any proposals received after the scheduled closing date and time for receipt of proposals, as listed in Paragraph 7.3 (RFP Timetable), will not be accepted and returned to the sender unopened. Timely hand-delivered proposals are acceptable. No facsimile (fax) or electronic mail (e-mail) copies will be accepted.

All proposals shall be firm offers and may not be withdrawn for a period of one hundred eighty (180) days following the last day to submit proposals.

8 SELECTION PROCESS AND EVALUATION CRITERIA

8.1 Selection Process

8.1.1 The County reserves the sole right to judge the contents of the proposals submitted pursuant to this RFP and to review, evaluate and select the successful proposal(s). The selection process will begin with receipt of the proposal on **April 10, 2025, 12:00 p.m., PT.**

Evaluation of the proposals will be made by an Evaluation Committee selected by the Department. The Committee will evaluate the proposals and will use the evaluation approach described herein to select a prospective Contractor. All proposals will be evaluated based on the criteria listed below. All proposals will be scored and ranked in numerical sequence from high to low. The County may also, at its option, invite the Proposers being evaluated to make a verbal presentation or conduct site visits, if appropriate. The Evaluation Committee may utilize the services of appropriate experts to assist in this evaluation.

After a prospective Contractor has been selected, the County and the prospective Contractor(s) will negotiate a Contract for submission to the Board of Supervisors for its consideration and possible approval. If a satisfactory Contract cannot be negotiated, the County may, at its sole discretion, begin the Contract negotiations with the next qualified Proposer who submitted a proposal, as determined by the County.

The recommendation to award a Contract will not bind the Board of Supervisors to award a Contract to the prospective Contractor.

The County retains the right to select a proposal other than the proposal receiving the highest number of points if the County determines, in its sole discretion, another proposal is the most overall qualified, cost-effective, responsive, responsible and in the best interest of the County.

8.2 Adherence to Minimum Mandatory Requirements (Pass-Fail)

8.2.1 The County shall review Exhibit 1 (Proposer's Organization Questionnaire/Affidavit) in Appendix D (Required Forms) and determine if the Proposer meets the minimum mandatory requirements as outlined in Paragraph 3 (Proposer's Minimum Mandatory Qualifications) of this RFP.

Failure of the Proposer to comply with the minimum requirements may eliminate its proposal from any further consideration. The County may elect to waive any informality in a proposal if the sum and substance of the proposal is present.

8.3 Disqualification Review

8.3.1 A proposal may be disqualified from consideration because a Department determined it was non-responsive at any time during the review/evaluation process. If a Department determines that a proposal is disqualified due to non-responsiveness, the Department shall notify the Proposer in writing.

Upon receipt of the written determination of non-responsiveness, the Proposer may submit a written request for a Disqualification Review within the timeframe specified in the written determination.

A request for a Disqualification Review may, in the Department's sole discretion, be denied if the request does not satisfy all of the following criteria:

- 1. The request for a Disqualification Review is submitted timely (i.e., by the date and time specified in the written determination); and
- 2. The request for a Disqualification Review asserts that the Department's determination of disqualification due to non-responsiveness was erroneous (e.g. factual errors, etc.) and provides factual support on each ground asserted as well as copies of all documents and other material that support the assertions.

The Disqualification Review shall be completed, and the determination shall be provided to the requesting Proposer, in writing, prior to the conclusion of the evaluation process.

The Proposer can also be disqualified for Paragraph 5.8 (Determination of Proposer Responsibility).

8.4 Business Proposal Evaluation and Criteria (80%)

- **8.4.1** Any reviews conducted during the evaluation of the proposal may result in a point reduction.
- 8.4.2 Proposer's Qualifications (30%)

Proposer's Background and Experience (20%)

The Proposer will be evaluated on their experience and capacity as a corporation or other entity to perform the required services based on information provided in Subparagraph 7.9.7.1 (Proposer's Background and Experience (Section B.1)) of the proposal.

Proposer's References/ Work History (10%)

The Proposer will be evaluated on the verification of references provided in Subparagraph 7.9.7.2 (Proposer's References (Section B.2)) of the proposal. In addition to the references provided, a review will include the Contractor Alert Reporting Database, if applicable, reflecting past performance history on County or other Contracts. This review may result in point deductions up to one hundred percent (100%) of the total points awarded in this evaluation category. Additionally, a review of terminated Contracts will be conducted which may result in point deductions.

A review will be conducted to evaluate the Proposer's financial capability as provided in Subparagraph 7.9.7.3 (Proposer's Financial Capability (Section B.3)) of the proposal.

A review will be conducted to determine the significance of any litigation or judgments pending against the Proposer as provided in Subparagraph 7.9.7.4 (Proposer's Pending Litigation and Judgment (Section B.4)) of the proposal.

8.4.3 Proposer's Approach to Providing Required Services (40%)

The Proposer will be evaluated on its description of the methodology to be used to meet the County's requirements based on information provided in Subparagraph 7.9.8 (Proposer's Approach to Provide Required Services (Section C)) of the proposal.

8.4.4 Quality Control Plan (10%)

The Proposer will be evaluated on its ability to establish and maintain a complete Quality Control Plan to ensure the requirements of this Contract are provided as specified. Evaluation of the Quality Control Plan shall cover the proposed monitoring system of all services listed on the Performance Requirements Summary (PRS) based on the information provided in Subparagraph 7.9.9 (Proposer's Quality Control Plan (Section D)) of the proposal.

8.4.5 Exceptions to Terms and Conditions of Sample Contract and/or Requirements of the Statement of Work

The Proposer will be evaluated on their willingness to accept the Terms and Conditions outlined in Appendix C (Sample Contract) and the Requirements of the Statement of Work outlined in Appendix A (Statement of Work), as stated in Subparagraph 7.9.11 (Terms and Conditions in the Sample Contract, and Requirements of the Statement of Work (SOW): Acceptance of/or Exceptions to (Section E)) of the proposal. The County may deduct rating points or disqualify the proposal in its entirety if the exceptions are material enough to deem the proposal non-responsive.

The Proposers are further notified that the County may, in its sole determination, disqualify any Proposer with whom the County cannot satisfactorily negotiate a Contract.

8.4.6 Intentionally Omitted

8.5 Cost Proposal Evaluation Criteria (20%)

The maximum number of possible points will be awarded to the lowest cost proposal. All other proposals will be compared to the lowest cost and points awarded accordingly.

However, should one or more of the Proposers request and be granted the preference, the cost component points will be determined as follows:

Fifteen percent (15%) of the lowest cost proposed will be calculated, not to exceed \$150,000, and that amount will be deducted from the cost submitted by all Proposers who requested and were granted the preference.

In no case shall any preference be combined to exceed fifteen percent (15%) of the lowest responsible bid meeting specifications.

8.6 Intentionally Omitted

8.7 Department's Proposed Contractor Selection Review

8.7.1 Departmental Debriefing Process

Upon completion of the evaluation, the Department shall notify the remaining Proposers in writing that the Department is entering negotiations with another Proposer. Upon receipt of the letter, any

non-selected Proposer may submit a written request for a Debriefing within the timeframe specified in the letter. A request for a Debriefing may, in the Department's sole discretion, be denied if the request is not received within the specified timeframe.

The purpose of the Debriefing is to compare the requesting Proposer's response to the solicitation document with the evaluation document. The requesting Proposer shall be debriefed only on its response. Because the Contract negotiations are not yet complete, responses from other Proposers shall not be discussed, although the Department may inform the requesting Proposer of its relative ranking.

During or following the Debriefing, the Department will instruct the requesting Proposer of the manner and timeframe in which the requesting Proposer must notify the Department of its intent to request a Proposed Contractor Selection Review (see Subparagraph 8.7.2 Proposed Contractor Selection Review), if the requesting Proposer is not satisfied with the results of the Debriefing.

8.7.2 Proposed Contractor Selection Review

Any Proposer that has timely submitted a notice of its intent to request a Proposed Contractor Selection Review as described in Subparagraph 8.7.2 (Proposed Contractor Selection Review) may submit a written request for a Proposed Contractor Selection Review, in the manner and timeframe as shall be specified by the Department.

A request for a Proposed Contractor Selection Review may, in the Department's sole discretion, be denied if the request does not satisfy all of the following criteria:

- 1. The request for a Proposed Contractor Selection Review is submitted timely (i.e., by the date and time specified by the Department);
- 2. The person or entity requesting a Proposed Contractor Selection Review asserts in appropriate detail with factual reasons one or more of the following grounds for review:
 - a. The Department materially failed to follow procedures specified in its solicitation document. This includes:

- i. Failure to correctly apply the standards for reviewing the proposal format requirements.
- ii. Failure to correctly apply the standards, and/or follow the prescribed methods, for evaluating the proposals as specified in the solicitation document.
- ii. Use of evaluation criteria that were different from the evaluation criteria disclosed in the solicitation document.
- b. The Department made identifiable mathematical or other errors in evaluating proposals, resulting in the Proposer receiving an incorrect score and not being selected as the recommended Contractor.
- c. A member of the Evaluation Committee demonstrated bias in the conduct of the evaluation.
- d. Another basis for review as provided by state or federal law; and
- 3. The request for a Proposed Contractor Selection Review sets forth sufficient detail to demonstrate that, but for the Department's alleged failure, the Proposer would have been the lowest cost, responsive and responsible bid or the highest-scored proposal, as the case may be.

Upon completing the Proposed Contractor Selection Review, the Department representative shall issue a written decision to the Proposer within a reasonable time following receipt of the request for a Proposed Contractor Selection Review, and always before the date the Contract award recommendation is to be heard by the Board. The written decision shall additionally instruct the Proposer of the manner and timeframe for requesting a County Independent Review. (See Paragraph 8.8 (County Independent Review Process) below.

8.8 County Independent Review Process

8.8.1 Any Proposer that is not satisfied with the results of the Proposed Contractor Selection Review may submit a written request for a County Independent Review in the manner and timeframe specified by the Department in the Department's written decision regarding the Proposed Contractor Selection Review.

A request for County Independent Review may, in the County's sole discretion, be denied if the request does not satisfy all of the following criteria:

- 1. The request for a County Independent Review is submitted timely (i.e., by the date and time specified by the Department); and
- 2. The person or entity requesting review by a County Independent Review has limited the request to items raised in the Proposed Contractor Selection Review as listed in Paragraph 8.7.2 (Proposed Contractor Selection Review) above.

Upon completion of the County Independent Review, Internal Services Department will forward the report to the Department, which will provide a copy to the Proposer.

1

APPENDICES

APPENDIX A: STATEMENT OF WORK

APPENDIX B: STATEMENT OF WORK TECHNICAL EXHIBITS

APPENDIX C: SAMPLE CONTRACT

APPENDIX D: REQUIRED FORMS

APPENDIX E: RFP TRANSMITTAL FORM TO REQUEST A SOLICITATION

REQUIREMENTS REVIEW

APPENDIX F: COUNTY OF LOS ANGELES POLICY ON DOING BUSINESS

WITH SMALL BUSINESS

APPENDIX G: JURY SERVICE ORDINANCE

APPENDIX H: LISTING OF CONTRACTORS DEBARRED IN LOS ANGELES

COUNTY

APPENDIX I: IRS NOTICE 1015

APPENDIX J: INTENTIONALLY OMITTED

APPENDIX K: INTENTIONALLY OMITTED

APPENDIX L: INTENTIONALLY OMITTED

APPENDIX M: BACKGROUND AND RESOURCES: CALIFORNIA CHARITIES

REGULATION

APPENDIX N: DEFAULTED PROPERTY TAX REDUCTION PROGRAM

EXHIBIT A STATEMENT OF WORK

TABLE OF CONTENTS

SECTION		TITLE	PAGE		
1.0	sco	PE OF WORK	1		
2.0	SPE	CIFIC TASKS	16		
	2.1	Global Positioning System Monitoring Program	16		
	2.2	Juvenile Electronic Monitoring Program	20		
	2.3	Adult Electronic Monitoring Program	23		
	2.4	Implementation Services	28		
	2.5	Optional Work	28		
3.0	QUA	QUALITY CONTROL PLAN			
4.0	QUA	UALITY ASSURANCE PLAN2			
5.0	DEF	INITIONS	31		
6.0	RES	PONSIBILITIES	32		
6 <u>C</u> 6 6 6	COU	<u>INTY</u>			
	6.1	Personnel	32		
	CON	<u>ITRACTOR</u>			
	6.2	Contractor's Project Director	32		
	6.3	Contractor Personnel	33		
	6.4	Contractor Furnished Items	34		
	6.5	Contractor's Office	35		
	6.6	Identification Badges	35		
7.0	HOU	HOURS/DAYS OF WORK			
8.0	UNS	UNSCHEDULED WORK			
9.0	PERFORMANCE REQUIREMENTS SUMMARY35				

1.0 SCOPE OF WORK

Contractor will provide Comprehensive Electronic Monitoring Services (CEMS) on a twenty-four (24) hour continuous basis for the County of Los Angeles Probation Department's (County's) Global Positioning System Monitoring Program (GPSMP), Juvenile Electronic Monitoring Program (JEMP), and Adult Electronic Monitoring Program (Adult EMP). Contractor will provide all technology, equipment, consumables, systems and related support services, cellular wireless services, accessories, tracking services, data storage, local law enforcement crime scene incident data collection and storage, crime scene correlation mapping analysis and reporting, point pattern analysis, monitoring center support services, training, user manuals, consultation, on-site assistance, and problem analysis not limited to additional support services. Contractor's administrative office, monitoring services and data center facilities will be located within the United States.

Contractor will provide to County information relating to research findings and new developments in the electronic monitoring industry. Contractor will advise County of any and all innovations and new equipment in the electronic monitoring industry and provide the County with the latest technology. County will be given the opportunity to incorporate improved electronic monitoring technologies as Contractor makes modifications and/or enhancements to its equipment and monitoring capabilities. Contractor will notify the County Program Manager in writing at least thirty (30) days prior to any changes in operations, that include, but are not limited to, new equipment, upgrades to equipment, services, software, or any other changes that could affect the County's Comprehensive Electronic Monitoring Services. Unless expressly stated otherwise, all Comprehensive Electronic Monitoring Services will be provided at no additional cost to County other than the per day, per activated device rate as set forth in the terms of this Contract.

1.1 MONITORING SERVICES

- 1.1.1 Contractor will provide a staffed communication system utilizing a toll free telephone number, a toll free fax number, and email access that is available on a twenty-four (24) hour continuous basis for technical analysis and application assistance.
- 1.1.2 Contractor will provide a toll free telephone number to participants to contact the monitoring center for the resolution of alert notifications.
- 1.1.3 Contractor will provide the capability for every human voice call in and out of the system to be recorded with a transaction record that indicates the called number, calling number, length and the resolution of the call. This information will be made available to the County Program Manager or designee at County's request.

- 1.1.4 Contractor will maintain accurate and concise historical logs of all telephone calls, text message, and emails attempted and completed, including date, time, and the associated incident. Contractor will make these logs available to County upon request. Documentation will be made available as requested by County during the term of the Contract as set forth in Paragraph 8.38 (Record Retention and Inspection/Audit Settlement) of the Contract.
- 1.1.5 Contractor will provide responses to queries from County staff and authorized law enforcement agencies within two (2) minutes of request. Queries may include participant's location by date range and time range.
- 1.1.6 Contractor will provide participant enrollments and scheduling via direct telephone request (password accessible).
- 1.1.7 Contractor will remotely activate or deactivate Global Positioning System (GPS) tracking services for a participant within thirty (30) minutes after receipt of request by telephone or email notification from County.
- 1.1.8 Contractor will have live staff available on a twenty-four (24) hour continuous basis to respond to designated alert notifications.
- 1.1.9 Contractor will triage alerts, and attempt to clear and record all efforts to clear alerts according to County's established protocols.
- 1.1.10 Contractor will provide initial notification via telephone, cellular telephone, text message, and email to designated County staff when an alert notification is generated.
- 1.1.11 Contractor will provide the participant's name, type of violation, time of violation, and the time and location of the participant's last known location to the Deputy Probation Officer (DPO), in the event of an alert violation.
- 1.1.12 Contractor will be able to receive confirmation via a telephone call, email or text message that alert notifications were received/acknowledged by County staff.
- 1.1.13 Contractor will escalate an alert notification to the next County designated contact if the initial County contact does not acknowledge receipt of the alert notification within County specified time.

1.1.14 Contractor will provide the level of alert notification based on protocols established by County. County reserves the right to change alert notification protocol.

1.2 MONITORING CENTER REQUIREMENTS

- 1.2.1 Contractor will provide an in-house, secure, and confidential monitoring service center and help-desk facility with an uninterruptible power source, firewall protections, and a backup disaster recovery plan.
- 1.2.2 The monitoring center and help-desk facility will be operated by Contractor within the United States and shall not be subcontracted.
- 1.2.3 The monitoring center will maintain redundant inbound and outbound communication services, provided by distinct carriers and/or methods, such that the failure of the primary service or method will not adversely affect the secondary (backup) service or method.
- 1.2.4 The monitoring center will perform complete support of all interface hardware and software equipment (within the monitoring center) necessary to ensure provision of the service for the duration of the Contract.
- 1.2.5 The monitoring center will provide County a contact number, accessible on a twenty-four (24) hour continuous basis. In the event any component of the monitoring service becomes inoperable, Contractor will immediately notify the County Program Manager or designee by telephone no later than thirty (30) minutes after service failure.
- 1.2.6 Contractor will have a secondary (backup) monitoring center that provides full operational functions in the event the primary monitoring center is disabled. The secondary monitoring center will be located sufficiently distant from the primary center, such that it is unlikely to be adversely affected by a manmade or natural event or loss of electrical or communications services that could disable the primary monitoring center. In the event of primary monitoring center disruption, the secondary (backup) monitoring center must be activated and fully functional within sixty (60) minutes of initial system failure.
- 1.2.7 Contractor will have documented policies and procedures for network security, application security, data transmission and data security, as well as monitoring center physical security. Contractor

will supply County with a copy of its network security policy and procedures. Contractor shall abide by County's network security policy and procedures, Technical Exhibit 5 (Information Security and Privacy Requirements) of Appendix B (Statement of Work Technical Exhibits).

1.2.8 The monitoring center will be equipped with automated fire detection and suppression equipment.

1.3 SYSTEM SOFTWARE REQUIREMENTS

- 1.3.1 Web-based software (software) will be accessible at no cost to the County, including no licensing fee. The software will support at least seventy-five (75) end users at any one point.
- 1.3.2 The software will have the capability to adjust the frequency at which a GPS tracking device collects a data point and/or makes a cellular call, and transfer the GPS data to Contractor's database.
- 1.3.3 The software will allow for unique alphanumeric identifiers designated by the County that will be used for each participant within Contractor's web-based software. Once the identifier is entered in the appropriate field, only authorized staff will have access to modify the identifier.
- 1.3.4 The software will have the following minimum mandatory fields for initial participant enrollment: name; participant photo; unique alphanumeric identifiers; address; telephone number; offense information; date of enrollment; serial number of device; time zone.
- 1.3.5 The software will be supported by a database that allows for multiple data fields that vary in length. The data fields will be subject to final approval by County.
- 1.3.6 The software will provide the capability for County to download data and reports from the database, through a secured internet access.
- 1.3.7 The software will have the capability to query the database for any/all electronically monitored participants based upon specified dates times and locations.
- 1.3.8 The software will reflect both the alternative tracking points in the absence of GPS versus GPS tracking points and visually differentiate the location indicators on the map between the two.

- 1.3.9 The software will provide the capability for the entry of narrativestyle notes by County and/or Contractor's monitoring center staff.
- 1.3.10 The software will display contact information of the assigned DPO.
- 1.3.11 The software will allow participants to be classified by categories. These Categories may be modified by County.
- 1.3.12 The system will be able to import County provided geographic information system data.
- 1.3.13 The software will be able to assign role-based security levels to users.
- 1.3.14 The system will provide daily backup of data.
- 1.3.15 The software must be able to display a report of a DPO's caseload on one screen and allow access to information on individual participants shown on the reports directly through a hyperlink or similar method.
- 1.3.16 County will have access to a website tracking usage report for each user account to provide a time-stamped listing of date/duration/time of login, a chronological event listing of GPSMP participant profile and mapping information viewed, and a summary/detail report of total time spent in the software per login.
- 1.3.17 The system must be able to export all electronic monitoring data in Microsoft Excel, Adobe PDF, and GIS file format, available on an as-needed basis upon County's request.
- 1.3.18 The software must be able to access archived data.
- 1.3.19 The system must be capable of fully functioning with at least seventy-five (75) administrative users simultaneously during peak periods.
- 1.3.20 The software must provide the ability for the DPO or law enforcement to instantly locate the participant's tracking device to receive the most recent, near real time, location.
- 1.3.21 The software must be capable of quickly changing between information pages for different participants assigned to an individual DPO's caseload, or for supervisors and managers, for all participants on any DPO's caseload.

- 1.3.22 The software must allow for markup data to be added, recorded and displayed for individual and groups of participants.
- 1.3.23 The software must provide a hyperlink to the participant profile information, from any screen associated with the participant.
- 1.3.24 The software system will provide for one hundred (100) percent redundancy to avoid excessive downtime due to hardware or software issues. In the event of application disruption, the software system must be fully functional within sixty (60) minutes of initial system failure. Contractor must notify County immediately in the event of application disruption.

1.4 MAPPING CAPABILITY

- 1.4.1 The software must have the capability to view location information for individuals or groups of participants.
- 1.4.2 The software must have the capability to quickly zoom in to street level including displaying places of interest frequented by participants.
- 1.4.3 The software must be able to display participant location information in a sequenced event and/or at a specific date and time.
- 1.4.4 The software must save/bookmark mapped locations that can be easily accessed by users.
- 1.4.5 The system must provide exporting of data for video capture modes and/or map printing capability at a minimum resolution of 600 dpi.
- 1.4.6 Crime scene correlation mapping must be provided on up to date high resolution maps with aerial photography capability. The software must provide access to a public places of interest layer. These maps must be available at a resolution that provides for easy visibility of streets, buildings, locations, and other points of interest as well as participant locations.
- 1.4.7 The software must allow the user to define and display a variety of zone types including, but not limited to, inclusion, exclusion, each with date and time parameter capability. Maps must also allow users to determine which geographical features or data layers are to be displayed on a particular map.

- 1.4.8 The software must allow users to configure and display polygonal zones.
- 1.4.9 The software must display and hide zones within specified areas of interest.
- 1.4.10 The software must be capable of globally applying specified zones to multiple participants when initiated by County.
- 1.4.11 The software must be capable of quickly loading twenty-four (24) hours of GPS points, subject to a specified date/time selection.
- 1.4.12 The software must display a minimum of seven (7) days of GPS points on an individual participant at one time based on the time the information accessed
- 1.4.13 When multiple days are loaded, the software must distinguish between each calendar day (by color code or some other means) to identify patterns of travel and locations frequented.
- 1.4.14 The software must quickly identify "locations visited" by the participant and identify travel patterns.
- 1.4.15 The software must be able to "play back" a time visualization of GPS points.
- 1.4.16 The software must allow for filtering of GPS points to display time of day, duration of stay, and speed of travel.
- 1.4.17 The software must differentiate GPS points when a violation has occurred and shall define the violation.
- 1.4.18 The software must display various map types that may be selected by the user, including, but not limited to, Satellite, Roads, Birds eye, Street, etc.
- 1.4.19 The software must allow users to specify a geographical zone, and display only points in the define area only.

1.5 CRIME SCENE CORRELATION MAPPING ANALYSIS REQUIREMENTS

1.5.1 The software must have a crime scene correlation mapping component that correlates crime and incident report data collected from local law enforcement agency stakeholders, and associates the crime and incident report data with the tracking data of participants.

- 1.5.2 Crime scene correlation mapping reports will be generated on a daily basis with standard reports generated each in accordance with County specifications.
- 1.5.3 Crime scene correlation mapping will also be available on a near "real time" basis immediately upon request and notification by the County.
- 1.5.4 Crime scene correlation mapping analysis will identify when one or more participants on GPS tracking were in the vicinity of a crime. Mapping analysis will also provide simultaneous locations for all participants assigned to GPS tracking in relation to the location, date and time of the crime and the speed and direction of the participant's travel.
- 1.5.5 Crime scene correlation maps must be easy to navigate both forward and backwards in time for an accurate depiction of a participant's position relative to criminal activity. Mapping information will include participant's time in each area speed and direction of travel, to allow for rapid investigative crime analysis.
- 1.5.6 All crime scene correlation services must be provided at no additional cost to County, and approved law enforcement agencies (via subpoena) other than the per-day, per activated device rate as set forth in the terms of this Contract.

1.6 POINT PATTERN ANALYSIS REQUIREMENTS

Contractor will use Point Pattern Analysis (PPA) software as part of CEMS. County reserves the right to modify requirements provided herein as necessary. The specific capability and functional operation of PPA will be approved by the County Program Manager:

- 1.6.1 The software must analyze locations frequented by a participant to identify places previously visited as well as new locations.
- 1.6.2 The software must filter data points previously visited by participants in order to display data points categorized as new locations.
- 1.6.3 The software must identify locations visited by a participant that had previously been identified or labeled, and list the data in report form.

1.7 REPORTING REQUIREMENTS

Contractor will develop a database system within thirty (30) days from the execution of the Contract, or at the timeframe determined by the County

Program Manager. County reserves the right to modify and format reporting requirements as necessary upon thirty (30) days' written notification to Contractor.

Contractor will provide a database system that allows County to generate the following standardized, preconfigured reports directly from Contractor's database through the secure internet site. To ensure that reports are accurate and timely, the system's database will be updated in real time to ensure all report data is current when viewed and/or downloaded by County personnel. All reports will have the capability of being queried, sorted, or filtered by any field contained in the report or by data parameters as applicable and shall be readable on screen, printable and downloadable into Microsoft Excel format. Report formats will be subject to final approval by the County Program Manager. Reports listed below will be developed in conjunction with and approved by the County Program Manager prior to any equipment deployment/implementation. All requested reports, data, statistical data, and ad hoc reports, will be provided at no additional cost to County.

- 1.7.1 Equipment Report The system will provide a real time report of all assigned active (in use) and inventoried equipment containing a serial number. This report shall include description/type of equipment, serial number, assigned participant, assigned DPO (if applicable), probation office identifier and totals.
- 1.7.2 Offender Alert Report The system will provide County the ability to generate an alert report, queried by individual assigned participant and date parameters that identifies the type of alert, time of alert, method of alert notification (email, telephone call, or text message) and recipient of alert notification.
- 1.7.3 Offender Report The system will provide County the ability to generate a summary report of all participants and/or assigned to a DPO that identifies participant name, probation number and address.
- 1.7.4 <u>Data Fields/Entry Exceptions Report</u> The system will provide County the ability to generate a report that identifies all data fields currently in use and identifies any incomplete or missing data entry.
- 1.7.5 Current Usage Report The system will provide County the ability to generate a current usage report indicating the actual number of daily service devices used to date (real time) for the monthly period. This report should be detailed to reflect participant name, probation number, service type (active) and number of days utilized to date.

- 1.7.6 Monthly Usage Report: The system will provide County the ability to generate a total monthly usage summary report, downloadable on the 5th day of the month, indicating the actual number of active GPS devices utilized during the previous months' time period. This report should be detailed to reflect participant name, probation number, type of service (active) and number of days utilized for the preceding monthly period and will be subtotaled by Probation unit/DPO. Any additions or deletions to the monthly usage report after the 5th of the month will be provided by written notice to the County Program Manager.
- 1.7.7 <u>Unresolved Reports</u> Contractor will provide County the ability to generate a daily report of all alerts that have not been resolved during a specific time period.
- 1.7.8 <u>Events Reviewed</u> Contractor will provide County the ability to generate a daily report that indicates that all events that have not been reviewed during a specific time period.
- 1.7.9 On Demand Report Contractor will provide County the ability to generate an on demand report that lists, times, dates, and participants viewed by any user.
- 1.7.10 <u>Ad Hoc Reports</u> Contractor will provide probation ad hoc reports from the system upon request of the County Program Manager.
- 1.7.11 Notification Report The system will provide County the ability to generate a report by date parameters, sorted by Probation unit/DPO, identifying the number and type of notifications during specified time parameters, such as 1) Monday-Friday, 8:00 am 5:00 pm; after hours, weekends and 2) total number of alerts per Probation unit/DPO; percent of total Probation office alerts per type of alert; and method of alerts notification per month, per participant within the Probation unit.
- 1.7.12 <u>Daily Alert Summary Report</u> Contractor will submit a daily report between the hours of midnight and 6:00 a.m. to County that identifies each alert, type of alert, time and duration of alert, assigned DPO, assigned participant, and totals for the previous day. This report shall be sorted by Probation unit/DPO and will be e-mailed to all designated DPO within each Probation office.
- 1.7.13 <u>Crime Scene Correlation Mapping Report</u> Crime scene correlation mapping reports will be generated regularly on a daily basis with standard reports generated each regular scheduled work day in accordance with County specifications. This requirement is held in

abeyance until such time as the appropriate/necessary information technology infrastructure is in place. Until such automated interfaces are in place, Contractor will provide crime scene correlation mapping reports based upon the County identifying and supplying specific dates, times and locations.

- 1.7.14 Monthly System Performance Report Contractor will submit a monthly report for the preceding month, no later than the 10th day of the following month reporting performance of the system, including any system interruptions, to the County Program Manager or designee. The report will include regular operational metrics such as timing of system updates, failed connections and any system unavailability.
- 1.7.15 <u>Annual Summary Report</u> Contractor will submit an Annual Summary report, on or before the anniversary date of the Contract reporting all issues identified and reported by County. Contractor shall include information on the following: new technology employed improvements to devices and/or service delivery, dates of training and/or on-site technical assistance and court appearances.
- 1.7.16 <u>Customized Report</u> Contractor will provide customized reports as requested by County, including, but not limited to, data downloads generated in Contractor's system such as mapping and location data.

1.8 DATA DOCUMENTATION REQUIREMENTS

Contractor will deliver the data documentation as defined in this section within ninety (90) days from the execution of the Contract, or at the timeframe determined by the County Program Manager. County reserves the right to modify requirements as necessary upon thirty (30) days' written notification to Contractor. The system will be developed in conjunction with and approved by the County Program Manager.

All Contractor documentation must be produced using software compatible with Microsoft Office Suite 2010 and produced in electronic and hardcopy formats.

Contractor must provide comprehensive functional and user documentation, including:

1.8.1 Comprehensive User Operating Instructions, customized to Probation's user environment;

- 1.8.2 System Screen Flow Diagrams;
- 1.8.3 Glossary of system terms & acronyms;
- 1.8.4 User documentation;
- 1.8.5 Available Reports and associated Report Descriptions; and
- 1.8.6 Documentation for each screen.

The documentation must be presented in the following format:

- 1.8.7 Five (5) complete hard copy sets with license to reproduce for County's exclusive usage; and
- 1.8.8 An electronic copy both in Adobe PDF format and in Microsoft format with license to revise and distribute for County's exclusive purpose.

1.9 DATA SUPPORT REQUIREMENTS

Contractor must provide ongoing technical support to County, for technical issues, system enhancements and upgrades.

- 1.9.1 Contractor will ensure that solutions are customized to meet County's business and technical requirements.
- 1.9.2 Contractor will fully test solutions to ensure that the design is functional, stable and free of defects.
- 1.9.3 Contractor will fully test and integrate solutions with all required systems.

1.10 DATA CENTER REQUIREMENTS

- 1.10.1 Contractor will store participant location and alert information in a Data Center and redundant secure secondary Data Center at least two hundred (200) miles apart. All data will be located in USA.
- 1.10.2 Data Centers will have multiple physical security features.
- 1.10.3 Data Centers will have ventilation and temperature control adequate to ensure the proper function of the hardware.
- 1.10.4 Data Centers will have an uninterruptible power supply.

1.10.5 The Data Center system will provide for one hundred (100) percent redundancy to avoid excessive downtime due to hardware or software issues. In the event of data disruption, the secondary data center must be activated and fully functional within sixty (60) minutes of initial system failure.

1.11 DATA ACCESS/STORAGE REQUIREMENTS

- 1.11.1 Contractor will maintain and make available to County unaltered recorded data of participant location information and violations, accessible in original form. Records will be protected from unauthorized access and computerized records will have appropriate safeguards.
- 1.11.2 Contractor will provide affidavit and/or expert witness testimony for prosecution of violations based on CEMS data in court proceedings, violation hearings, or any other proceeding, as needed, at no additional cost to County. The affidavit and/or testimony will address accuracy by a statement of certification of accuracy.
- 1.11.3 Contractor will not release or reveal any data, program information, operation protocols, implementation plans, training material, reports, publications, updates, and/or statistical data to any entity without a legal subpoena or prior written approval from the County Program Manager.
- 1.11.4 All records (automated, electronic or hard copy files that contain any data or information pertaining to the County's GPS Program, including, but not limited to, participant information, system performance, complaints, invoices, technical advice and/or enhancements, system problems and any written documentation on performance or compliance issues and any meeting notes where County and Contractor met) remain the property of County and will be returned within thirty (30) days in the event the Contract is terminated, or expires, at no additional cost to County.
- 1.11.5 Data will be recorded with a historical transaction record and stored/archived for retrieval/backup in a database when requested by County.
- 1.11.6 Contractor will submit written policies and procedures for removal, storage, and destruction of all documents and data files, prior to the removal, storage or destruction of said documents for County's review and approval. A list of all documents being considered for

- removal or destruction will be submitted to County for approval prior to removal or destruction.
- 1.11.7 The storage of any documentation evidencing criminal activity will be maintained in accordance with all applicable federal and state laws, in a manner and for the duration mutually agreed upon by both Contractor and County. Contractor will be responsible for providing a policy for approval by County within fourteen (14) days after the execution of this Contract.

1.12 LOCAL LAW ENFORCEMENT CRIME SCENE INCIDENT DATA COLLECTION REQUIREMENTS

- 1.12.1 The software must have the capability to query participant locations based on date, time and proximity to crime scene locations.
- 1.12.2 Contractor will assist in establishing local law enforcement stakeholders for the purpose of attaining crime and incident report data for GPS crime scene correlation purposes.
- 1.12.3 Contractor will provide and maintain interface capabilities with local law enforcement agency data systems, including, but not limited to, dispatch systems and record management systems, to collect and download crime and incident report data on a daily basis.
- 1.12.4 Contractor will provide all software and hardware system requirements necessary for the complete operation of GPS tracking of participants, the collection of crime and incident data, and the crime scene correlation component.
- 1.12.5 Contractor will ensure that all modalities of information transferred between local law enforcement stakeholders (via subpoena) and Contractor contain and adhere to the highest degree of industry standards for security and confidentiality.
- 1.12.6 Contractor must provide all required access to its secure Central Information System to authorized County staff, DPOs, and local law enforcement agency stakeholders authorized to participate in the GPS Program. System access will only be provided to those persons who have written authorization from the County Program Manager.
- 1.12.7 Contractor will ensure that no probation data can be accessed by any other agency when it is viewing any data or crime scene correlation without authorized consent of the County Program Manager.

1.13 CRIME SCENE CORRELATION MAPPING ANALYSIS REQUIREMENTS

- 1.13.1 All crime scene correlation services will be provided at no additional cost to County and approved law enforcement agencies (via subpoena) other than the per day, per activated device rate as set forth in the terms of this Contract.
- 1.13.2 The software will have a crime scene correlation mapping component that correlates crime and incident report data collected from local law enforcement agency stakeholders and associates the crime data with the tracking data of participants.
- 1.13.3 Crime scene correlation mapping reports will be generated on a daily basis with standard reports generated each scheduled work day in accordance with County specifications. This requirement is held in abeyance until such time as the appropriate/necessary information technology infrastructure is in place. Until such time automated interfaces are in place. Contractor will provide crime scene correlation mapping reports based upon the County identifying and supplying specific dates, times, and location.
- 1.13.4 Crime scene correlation mapping will be available on a near "real time" basis immediately upon request and notification by the County.
- 1.13.5 Crime scene correlation mapping analysis will identify when participants on GPS tracking were in the vicinity of a crime. Mapping analysis will also provide simultaneous locations for all participants assigned to GPS tracking in relation to the location, date and time of the crime and the speed and direction of the participant's travel.
- 1.13.6 Crime scene correlation maps will be easy to navigate both forward and backwards in time for an accurate depiction of a participant's position relative to criminal activity. Mapping information will include participant's time in each area, speed, and direction of travel to allow for rapid investigative crime analysis.
- 1.13.7 All crime scene correlation mapping services will be provided at no additional cost to County and approved law enforcement agencies other than the per-day, per activated device rate as set forth in the terms of this CONTRACT.

1.14 TRAINING AND CONSULTATION REQUIREMENTS

- 1.14.1 Contractor will provide comprehensive initial orientation training. Contractor will develop and provide on-site ongoing training for County staff on the operational use of the system and the use of all associated equipment and services. Any request for initial orientation or subsequent training of County staff must occur within forty-eight (48) hours of a request, as specified by County. All training curriculum must be approved by the County Program Manager and becomes the property of the County upon approval. County reserves the right to develop, change, add, modify, or eliminate components of the training curriculum as deemed necessary.
- 1.14.2 Contractor will provide all materials and equipment necessary to perform training sessions. All training manuals must be hard copies, color print, user friendly, and kept up to date with the most current protocols. Training manuals will be provided to each staff during training and also upon the request of the County Program Manager or designee. All training manuals and all associated training materials will be provided at no cost to County or law enforcement agencies approved by the County Program Manager.
- 1.14.3 Contractor will provide additional web-based seminar or similar web-based online training as specified by County. This training shall cover system, software and hardware updates after initial orientation. Online references or prerecorded training videos or the like will not constitute webinar or similar web-based online training under this section unless authorized in writing by the County Program Manager. All online training curriculums will be approved by the County Program Manager.
- 1.14.4 Contractor will provide crime scene correlation mapping analysis trainings on an ongoing basis, as needed and as specified by County. This training will be performed at designated locations with approval by the County Program Manager, or designee.

2.0 SPECIFIC TASKS

2.1 GLOBAL POSITIONING SYSTEM MONITORING PROGRAM

Contractor will provide services necessary to operate a Global Positioning System Monitoring Program (GPSMP) for participants including, but not limited to, sex offenders; domestic violence/stalking offenders, identified and validated gang members and associates; violent offenders; offenders involved in major crimes as identified by local law enforcement; public

interest/high notoriety cases; and any court ordered cases (GPSMP participant or participant).

Contractor will provide a one-piece electronic monitoring device with Global Positioning System (GPS), Cellular Radio Frequency (CRF) and land line Radio Frequency (RF) capabilities as specified in Technical Exhibit 4 (Equipment Requirements) of Appendix B (Statement of Work Technical Exhibits). Contractor will also make available Transdermal Alcohol Devices. Contractor will provide the central computer hardware and software necessary for the operation and continuous monitoring of the GPS tracking and data storage. County currently has approximately three hundred (300) participants in GPSMP.

- 2.1.1 Contractor shall supply one-piece GPS tracking devices and will be responsible for providing, at a minimum, the following services:
 - 2.1.1.1 Monitor, track, collect and record, in a central information software system, all movement and location data of GPSMP participants assigned to GPS tracking on a twenty-four (24) hour continuous basis.
 - 2.1.1.2 Provide daily reports of each GPSMP participant's GPS tracking activities, including technical violations, to the GPSMP participant's DPO and unit supervisors (via telephone call, text message, and/or email as specified by the County) for all GPSMP participants assigned to GPS monitoring.
 - 2.1.1.3 Provide immediate alert notification to the GPSMP participant's DPO and to the GPSMP participant (via telephone call, text message, and/or email as specified by the County) from Contractor's monitoring service center personnel twenty-four (24) hours per day, including, but not limited to, inclusion zone violations, tamper violations, no GPS notifications, lost cellular communication, and low/critical or shutdown battery notifications.
 - 2.1.1.4 Provide unlimited access to the most up-to-date maps available with most current graphics, aerial photography capabilities, and the flexibility to update maps with public places of interest icon references, including, but not limited to, schools, day-care centers, parks, or other areas of interest specific to each geographical area.
 - 2.1.1.5 Provide County and County-approved local law enforcement agencies unlimited use of the system software

- and database for instant access to the most recent near real-time location of the GPSMP Participant.
- 2.1.1.6 Provide near real-time, user-friendly unlimited access to the data, allowing County staff and local law enforcement agencies, as approved by County, to access and view up to the minute data for investigative, compliance, and criminal behavior analysis of GPSMP participants assigned to GPS tracking
- 2.1.1.7 County must have the ability to query, through a secured web-based software system, Contractor software and all devices assigned to the County. At a minimum the information should include the device number, status of device (operational or inactive) assigned location, GPSMP participant's name.
- 2.1.2 GPS services will be fully supported by monitoring services on a continuous basis and Contractor will provide all systems and equipment required for the service delivery including, but not limited to, the following:
 - 2.1.2.1 A software system with a secure database to monitor GPSMP participant activity;
 - 2.1.2.2 All monitoring devices to communicate location data to Contractor's central information software system; and
 - 2.1.2.3 All labor, materials, equipment, accessories and consumables necessary to perform GPS tracking on a continuous basis.

2.1.3 PROBATION INVENTORY REQUIREMENTS

- 2.1.3.1 County will determine the number of tracking devices, consumables, and installation equipment that will be kept in inventory at each County office.
- 2.1.3.2 Contractor will continually provide and maintain a backup/replacement inventory equal to or exceeding 25 percent of the actual number of GPS devices in use in each County office for the preceding month.
- 2.1.3.3 In the event of any monitoring device failure, County will replace the equipment from the backup/replacement inventory maintained at the assigned County office.

- 2.1.3.4 All devices in transit and/or pending delivery are not considered a part of County inventory. There will be no limit on the number of GPS devices kept in County inventory, and all GPS devices in County possession will be fully operable at all times while in inventory.
- 2.1.3.5 County will not pay Contractor an inventory fee, storage fee, installation equipment fee or any other fee related to inventory devices and equipment. County will only be obligated to pay the per day, per activated device rate.

2.1.4 ONGOING DEMONSTRATION/TESTING OF EQUIPMENT REQUIREMENT

Contractor will provide County unlimited use of two (2) fully operational devices for demonstration and/or test purposes. These devices will not be part of the backup/replacement inventory, will not count toward the per-day, per activated device rate, and will be provided at no additional cost. The devices may be demonstrated/tested at the discretion of the County Program Manager. At the end of Contract, the two (2) devices will be returned to Contractor.

- 2.1.4.1 Contractor will not release or reveal any GPS data, program information, operation protocols, implementation plans, training material, report(s), publication(s), updates, and/or statistical data to any entity, without a legal subpoena or prior written approval from County Program Manager.
- 2.1.4.2 All records (automated, electronic or hard copy files) remain the property of County and will be returned within thirty (30) days in the event the Contract is terminated or expired.
- 2.1.4.3 Data will be recorded with a historical transaction record and centrally stored/archived for retrieval/backup in a database, and made available when requested by County.
- 2.1.4.4 Contractor will submit written policies and procedures for removal, storage, and destruction of documents or data files that contain any data or information pertaining to the County's GPS Program including, but not limited to, participant information, system performance, complaints, invoices, technical advice and/or enhancements, system problems and any written documentation on performance or

compliance issues and any meeting notes where County and Contractor met prior to the removal, storage or destruction of said documents for County's review and approval. A list of all documents being considered for removal or destruction will be submitted to County for approval prior to removal or destruction.

- 2.1.4.5 The storage of any documentation evidencing criminal activity will be maintained in accordance with all applicable Federal and State laws, in a manner and for the duration mutually agreed upon by Contractor and County. Contractor will be responsible for providing a policy for approval by County within fourteen (14) days after the execution of this Contract. Contractor will abide by County's network security policy and procedures, Technical Exhibit 5 (Information Security and Privacy Requirements) of Appendix B (Statement of Work Technical Exhibits).
- 2.1.4.6 Provide appropriate staff to be available, at no cost to the County, to testify regarding a GPSMP participant, if required in any judicial proceeding.

2.2 JUVENILE ELECTRONIC MONITORING PROGRAM

Contractor will provide services necessary to operate a Juvenile Electronic Monitoring Program (JEMP) for specified participants at Probation locations.

Contractor will provide a one-piece electronic monitoring device with Global Positioning System (GPS), Cellular Radio Frequency (CRF) and land line Radio Frequency (RF) capabilities as specified in Technical Exhibit 4 (Equipment Requirements) of Appendix B (Statement of Work Technical Exhibits). Contractor will also make available Transdermal Alcohol Devices. Contractor will provide the central computer hardware and software necessary for the operation and continuous monitoring of the GPS tracking and data storage. County currently has approximately five hundred (200) participants in JEMP.

Devices for JEMP participants will be installed at the homes of JEMP participants and at the following Probation locations:

Barry J. Nidorf Juvenile Hall 16350 Filbert Street Sylmar, CA 91342 1:30pm - 4:00pm M-F Los Padrinos Juvenile Hall 7285 Quill Drive Downey, CA 90010 2:00pm - 7:00pm M - Sun East Los Angeles Area Office 4849 Civic Center Way Los Angeles, Ca 90022 10:00am – 12:00pm M-F

2.2.1 INSTALLATION

- 2.2.1.1 Contractor will provide installation and removal services during the above posted days/hours for each facility. Installation will occur within twenty-four (24) hours of notification by the County.
- 2.2.1.2 Installation by Contractor will consist of strapping the device to the JEMP participant's ankle. Subcontracting will be considered and/or allowed for this requirement of the service only.
- 2.2.1.3 For JEMP CRF switchable devices, Contractor will submit a proposal for the additional costs associated with providing installation and removal services where the device will be installed within twenty-four (24) hours of notification, by the Contractor at the locations specified.
- 2.2.1.4 Contractor will provide repair and/or replacement of equipment within twenty-four (24) hours of discovery or notification of a malfunction.
- 2.2.1.5 The installation of the JEMP devices may be subcontracted. Contractor will ensure that all applicable conditions of employment are in place including, but not limited to, background clearance for all Subcontractors.
- 2.2.2 Contractor will supply one-piece GPS tracking devices and will be responsible for providing, at a minimum, the following services:
 - 2.2.2.1 Monitoring, tracking, collecting and recording, in a software system, all movement and location data of JEMP participants assigned to GPS tracking, on a twenty-four (24) hour continuous basis.
 - 2.2.2.2 Providing daily reports of each JEMP participant's GPS and CRF tracking activities, including technical violations, to the JEMP participant's DPO and unit supervisors (via telephone

- call, text message, and/or e-mail as specified by County) for all JEMP participants assigned to GPS and CRF monitoring.
- 2.2.2.3 Providing immediate alert notification to the JEMP participant's DPO and to the JEMP participant (via telephone, call text message, and/or e-mail as specified by County) from Contractor's monitoring service center personnel twenty-four (24) hours per day, including, but not limited to, inclusion zone violations, tamper violations, no GPS notifications, lost cellular communication, and low/critical or shutdown battery notifications.
- 2.2.2.4 Providing unlimited access to the most up-to-date maps available with most current graphics, aerial photography capabilities, and the flexibility to update maps with public places of interest icon references, including, but not limited to, schools, day-care centers, parks, or other areas of interest specific to each geographical area.
- 2.2.2.5 Providing County and County-approved local law enforcement agencies unlimited use of the software and database for instant access to the most recent near real-time location of the JEMP participants.
- 2.2.2.6 Providing near real-time, user-friendly unlimited access to the data, allowing County staff and local law enforcement agencies, as approved by County, to access and view up to the minute data for investigative, compliance, and criminal behavior analysis of JEMP participants assigned to GPS tracking.
- 2.2.2.7 Providing the central computer hardware and software necessary for the operation of twenty-four (24) hours per day of continuous monitoring of the GPS tracking and data storage, including HMU (Home Monitoring Unit) data related to home detention.
- 2.2.2.8 County will have the ability to query, through a secured webbased system, Contractor software and all devices assigned to the County. At a minimum the information should include the device number, status of device (operational or inactive) assigned location, and JEMP participant's name.
- 2.2.3 GPS services will be fully supported by monitoring services on a continuous basis and Contractor will provide all systems and

equipment required for the service delivery including, but not limited to the following:

- 2.2.3.1 A system with a secure database to monitor JEMP participant activity.
- 2.2.3.2 Monitoring devices, all of which communicate location data to Contractor's central information software system.
- 2.2.3.3 All labor, materials, equipment, accessories and consumables necessary to perform GPS tracking on a continuous basis.
- 2.2.3.4 Appropriate staff, available at no cost to the County, to testify regarding a JEMP participant, if required in any judicial proceeding.

2.3 ADULT ELECTRONIC MONITORING PROGRAM

Contractor will provide services to operate an Adult Electronic Monitoring Program (Adult EMP) to assist in the supervision and tracking of participants that may be placed in Adult EMP including breath and transdermal alcohol monitoring, civil contempt or family law referrals, Los Angeles County Sheriff's Department (LASD) referrals, court ordered misdemeanor or felony pre-conviction, and court ordered misdemeanor or felony post-conviction (Adult EMP participants). The services will include orientation (in-person, enrollment, installation consultation and support services for program participants.

This program is part of an alternative to custody to either alleviate bed space in county jail or as support to the courts in criminal or civil matters. All participants will be supervised by Contractor.

Contractor will provide a one-piece electronic monitoring device with Global Positioning System (GPS), Cellular Radio Frequency (CRF) and land line Radio Frequency (RF) capabilities as specified in Technical Exhibit 4 (Equipment Requirements) of Appendix B (Statement of Work Technical Exhibits). Contractor will also make available Transdermal Alcohol Devices. County currently has approximately one thousand five hundred (1,500) participants in Adult EMP.

2.3.1 Contractor will provide installation and removal of all electronic monitoring equipment within established time frames as specified by County.

- 2.3.2 Contractor will monitor and document the activities of the Adult EMP participants on a continuous basis to determine if they are in compliance with all conditions of the sentencing court and/or Adult EMP requirements.
- 2.3.3 Contractor will notify County and appropriate authorities through standardized reports of an Adult EMP participant's compliance to these requirements within established time frames, as described throughout this section.
- 2.3.4 Contractor will maintain complete and accurate records relating to Adult EMP participants' compliance to orders of the court and Adult EMP requirements, as well as issuance, replacement and maintenance of equipment.
- 2.3.5 Contractor will provide a sufficient number of branch offices located in various geographic locations within the County of Los Angeles. Due to the geographic size of the County, Contractor will provide a minimum of four (4) branch offices, with at least one (1) located in the North, South, East and West region of the County, respectively. All such offices must be located at a reasonable distance away from any residential area, park, school, or other area or structure where children are reasonably likely to congregate as approved by the Los Angeles County Probation Department. County may agree to a lower number of offices based on their geographical locations and Contractor's overall demonstrated ability to provide the required services from the proposed locations.
- 2.3.6 County reserves the right to review and approve all publications and advertisements regarding Contractor's Adult EMP prior to release in accordance with Section 8.37 (Publicity) of the Contract.
- 2.3.7 Contractor will comply with the guidelines and standards of the County's EMP Operational Procedures Manual (Manual), Technical Exhibit 7 of Appendix B (Statement of Work Technical Exhibits). The Manual may be updated based on County needs.

2.3.8 Referral and Assessment Process

2.3.8.1 Participants are identified for Adult EMP by Probation, LASD, or the Courts. Identified participants may be handicapped, have medical restrictions, and/or caretakers of others.

- 2.3.8.2 Probation's Pretrial Services Division (PPSD) conducts a suitability assessment of identified participants prior to enrollment in Adult EMP. Each suitability assessment will consist of an interview; address verification; criminal record analysis, and completion of a validated risk assessment. PPSD will notify the Court, LASD and Contractor of the suitability assessment findings.
- 2.3.8.3 Each assessment will result in a classification finding of suitable, unsuitable or ineligible. A suitable assessment is defined as a participant who receives a low, medium, or high risk score, and meets the program's eligibility requirements.
- 2.3.8.4 An unsuitable assessment is defined as an individual whose risk score indicates a history of violence, threat to public safety or other unfavorable factors and whose risk score exceeds the established high risk threshold. An ineligible assessment is defined as an individual who is disqualified from Adult EMP for not meeting the minimum eligibility requirements.
- 2.3.9 Contractor will provide Adult EMP participants with written program rules, regulations, participant complaint, and equipment instructions at the time of enrollment.
- 2.3.10 Contractor will obtain Adult EMP participant's written consent to both participate in the program and to comply with the rules and regulations of the program at the time of enrollment.
- 2.3.11 Contractor will establish and maintain the Adult EMP participant's case files in compliance with the Manual. All case files will remain under the jurisdiction of Probation.
- 2.3.12 Contractor will assign and install active monitoring equipment, utilizing a random contact back-up system, to all Adult EMP participants within established time frames.
- 2.3.14 Contractor's case manager will meet in person with participants that have been assessed by Probation as specified in the Manual.
- 2.3.15 Contractor will monitor the Adult EMP participant's adherence to Adult EMP rules and regulations, and report compliance to County and appropriate authorities as described in the Manual.

- 2.3.16 Contractor will track and report the Adult EMP participant's adherence to orders of the court or program guidelines.
- 2.3.17 Contractor will provide statistical data as described in the Manual.
- 2.3.18 Contractor will repair and/or replace equipment within twenty-four (24) hours of discovery of a malfunction. Approved procedures must be followed to monitor Adult EMP participant during period of time equipment is not functioning properly.
- 2.3.19 Contractor will provide an immediate, documented telephone call to participant in response to tamper alerts. After the initial telephone call, Contractor will place telephone calls to the participant every two (2) hours until tamper alert is reset or transmitter replaced. Replacement, if necessary, must occur within twenty-four (24) hours of initial tamper alert. Equipment replacements, repairs, resets, and telephone calls must be documented.
- 2.3.20 Contractor will verify all departures to and from the participant's place of confinement.
- 2.3.21 Contractor will notify the County and designated authorities of the probable violation of Adult EMP conditions within the time limits established by the County.
- 2.3.22 Contractor's Adult EMP equipment and services will include the following:
 - 2.3.22.1 All monitoring activity from Adult EMP equipment at participant's residence will be reported to Contractor's monitoring center through a toll free telephone line.
 - 2.3.22.2 Monitoring software must be password protected.
 - 2.3.22.3 Monitoring software must have flexible curfew scheduling capability.
 - 2.3.22.4 Monitoring software must provide for twenty-four (24) hour pager alert for branch managers.
 - 2.3.22.5 Monitoring software must provide expansion capability and internet access.

- 2.3.22.6 Monitoring software must provide custom event printouts.
- 2.3.22.7 Monitoring software must be set to perform an automatic monitor check-in with the field-monitoring device a maximum of every four (4) hours. If check-in is not received within five (5) hours, a documented monitoring response is required by Contractor.
- 2.3.23 Contractor will develop written policy and procedures that specify the methods used for the receiving, safeguarding, disbursing, and recording of funds. Contractor's accounting procedures must minimize loss, mismanagement, and theft and must provide documentation sufficient for County to conduct a forensic audit of the program.
- 2.3.24 Contractor will maintain complete and accurate records regarding:
 - 2.3.24.1 Participant Orientation, and enrollment
 - 2.3.24.2 Participant's compliance
 - 2.3.24.3 Staff training
 - 2.3.24.4 Assignment, Installation, Replacement, and removal of participant's electronic monitoring equipment
- 2.3.25 Contractor will provide a computerized electronic monitoring equipment inventory and maintenance service for each piece of equipment used within this program, including all repairs and/or replaced equipment.
- 2.3.26 Contractor will provide access, at any time, to County for the purpose of inspecting records, offices or facilities being maintained in conjunction with this program.
- 2.3.27 Contractor will provide other services found to be necessary relating to the operation of an Adult EMP, including but not limited to, random contact monitoring equipment and approved transdermal and breath alcohol testing devices, either separately, or in conjunction with electronic monitoring.
- 2.3.28 Contractor will establish procedures to document the receipt and resolution of participant's disputes. Any disputes not resolved through this process within ten (10) business days of the receipt

of the dispute by Contractor will be referred to the County for mediation.

- 2.3.29 Contractor will investigate and respond to County in writing within five (5) business days from receipt of complaint by County. The response will include a statement of the facts, whether the allegation is true or false, corrective action taken or planned, if applicable, and controls to prevent reoccurrence of the problem.
- 2.3.30 Contractor will be available for County on-site inspections and audits to review and monitor all program components, including, but not limited to, participant case files, monitoring records, and other records to ensure Contractor compliance to Contract provisions.
- 2.3.31 Contractor will provide appropriate staff to be available, at no cost to the County, to testify regarding an Adult EMP participant if required in any judicial proceeding.

2.4 IMPLEMENTATION SERVICES

Contractor will provide a description of the methodology, including all applicable tasks and deliverables, for implementing the Comprehensive Electronic Monitoring Services (CEMS), including monitoring center and monitoring services implementation, which must meet all of the requirements set forth in Technical Exhibit 6 (CEMS Implementation Requirements) to Appendix B (Statement of Work Technical Exhibits) of the Contract, among others. Such CEMS implementation tasks and deliverables shall, to the extent applicable, address CEMS center set up, data conversion and migration, CEMS software and interfaces (if any), implementation of CEMS system acceptance testing and transition into operational CEMS environment.

Contractor will correct any and all deficiencies in the CEMS implementation services before County's acceptance and approval of such services and payment by County to Contractor for any services under the resultant Contract.

2.5 Optional Work

The Contractor will inform the County of new technology and methods of electronic monitoring that may be available during the term of this Contract, including, but not limited to, advances in Global Positioning System, and wrist worn devices, and/or other devices. County may determine to implement any of the new technology and methods of optional work under, and in accordance with, the Contract.

Contractor can include the new technologies/methods in the submission.

3.0 QUALITY CONTROL PLAN

Contractor will establish and maintain a Quality Control Plan to assure that the requirements of the Contract are met. A copy must be provided as changes occur. The original plan and any future amendments are subject to County review and approval and shall include, but are not limited to, the following:

- 3.1 An inspection system covering all the services listed on Technical Exhibit 1 (Performance Requirements Summary Chart) of Appendix B (Statement of Work Technical Exhibits). It must specify the activities to be inspected on a scheduled or unscheduled basis, how often inspections will be accomplished, and the title of the individual(s) who will perform the inspection.
- 3.2 The methods for identifying and preventing deficiencies in the quality of service performed before the level of performance becomes unacceptable.
- 3.3 A file of all inspections conducted by Contractor and, if necessary, the corrective action taken. This documentation will be made available as requested by County during the term of the Contract as set forth in Paragraph 8.38 (Record Retention and Inspection/Audit Settlement) of the Contract.
- 3.4 The methods for ensuring uninterrupted service to County in the event of a strike of County's or Contractor's employees or any other unusual occurrence (i.e., power loss or natural disaster) which would result in Contractor being unable to perform the contracted work.
- 3.5 The methods for assuring that confidentiality of adult and juvenile records, and other protected information are maintained while in the care of Contractor's employees.
- 3.6 The methods for maintaining security of records and the methods for preventing the loss or destruction of data.

4.0 QUALITY ASSURANCE PLAN

County or its agent will evaluate Contractor's performance under this Contract on not less than an annual basis. Such evaluation will include assessing Contractor's compliance with all Contract terms and performance standards. Contractor's deficiencies which County determines are severe or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors. The report will include improvement/corrective action measures taken by County and Contractor. If improvement does not occur

consistent with the corrective action measures, County may terminate this Contract or impose other penalties as specified in this Contract.

County will evaluate Contractor's performance under this Contract using the quality assurance procedures specified in Technical Exhibit 1 (Performance Requirements Summary Chart) of Appendix B (Statement of Work Technical Exhibits) or other such procedures as may be necessary to ascertain Contractor's compliance with this Contract.

4.1 Performance Evaluation Meetings

County's Program Manager may meet weekly with Contractor's Project Director during the first three (3) months of the Contract if County's Program Manager determines it necessary. However, a meeting will be held whenever a Contract Discrepancy Report (CDR) is issued. A mutual effort will be made to resolve all problems identified.

- 4.2 After the first three (3) months of operation, regular performance evaluation meetings will be held monthly in accordance with a mutually agreed upon schedule, or as required by County.
- 4.3 County will have the right to remove any Contractor personnel performing services under this Contract who is unsatisfactory in the opinion of County's Program Manager. Contractor personnel will be removed and replaced by Contractor within twenty-four (24) hours when requested to do so by County's Contract Manager.

4.4 Contract Discrepancy Report

Verbal notification of a Contract discrepancy will be made to Contractor's Project Director whenever a Contract discrepancy is identified. The problem will be resolved within a time period mutually agreed upon by County and Contractor.

County's Program Manager will determine whether a formal Contract Discrepancy Report shall be issued as referenced in Technical Exhibit 2 (Contract Discrepancy Report) of Appendix B (Statement of Work Technical Exhibits). Upon receipt of this document, Contractor is required to respond in writing to County's Program Manager within five (5) business days, acknowledging the reported discrepancies or presenting contrary evidence. Contractor will submit a plan for correction of all deficiencies identified in the Contract Discrepancy Report to County's Program Manager within ten (10) business days of receipt of the Contract Discrepancy Report.

5.0 **DEFINITIONS**

- 5.1 <u>Business Day</u> Monday through Friday, 8:00 a.m. to 5:00 p.m., PT, not including any County holidays.
- 5.2 <u>Contract Discrepancy Report (CDR)</u> A report prepared by County's Quality Assurance Evaluator to inform Contractor of faulty service. The CDR requires a response from Contractor explaining the problem and outlining the remedial action being taken to resolve the problem within ten (10) business days after receipt of the CDR.
- 5.3 <u>Contract Start Date</u> The date Contractor begins work in accordance with the terms of the Contract.
- 5.4 <u>Contractor's Project Director</u> Person designated by Contractor to administer the Contract operations after the Contract award.
- 5.5 <u>County's Contract Manager</u> Person designated by County with authority for County on contractual or administrative matters relating to this Contract.
- 5.6 <u>County's Contract Monitor</u> Person designated by County to monitor the Contract and provide reports to County's Contract Manager and County's Program Manager.
- 5.7 <u>County's Program Manager</u> Person designated by County to manage the daily operations under this Contract.
- 5.8 <u>Liquidated Damages</u> The monetary amount deducted from Contractor's payment due to non-compliance with the Contract and/or deficiencies in performance.
- 5.9 <u>Performance Requirements Summary (PRS)</u> The statement that identifies the key performance indicators of the Contract which will be evaluated by County to ensure contract performance standards are met by Contractor as referred to in Technical Exhibit 1 (Performance Requirements Summary Chart) of Appendix B (Statement of Work Technical Exhibits).
- 5.10 Records Personal and social history, including criminal information of adult/juvenile offenders. The records include legal documents and other information, which are confidential. The information is not to be discussed with, or disclosed to, unauthorized persons as defined by the County of Los Angeles Probation Department.
- 5.11 Quality Control Plan All necessary measures taken by Contractor to assure that the quality of service will meet the Contract requirements regarding security, accuracy, timeliness, appearance, completeness, consistency and conformity to

the requirements set forth in Technical Exhibit 1 (Performance Requirements Summary Chart) of Appendix B (Statement of Work Technical Exhibits).

6.0 RESPONSIBILITIES

County's and Contractor's responsibilities are as follows:

COUNTY

6.1 Personnel

County will administer the Contract according to Paragraph 6.0 (Administration of Contract - County) of the Contract. Specific duties will include:

- 6.1.1 Monitoring Contractor's performance in the daily operation of this Contract.
- 6.1.2 Providing direction to Contractor in areas relating to policy, information, and procedural requirements.
- 6.1.3 Preparing Amendments in accordance with the Paragraph 8.1 (Amendments) of the Contract.

CONTRACTOR

6.2 Contractor's Project Director

- 6.2.1 Contractor will provide its own full time officer or employee as Contractor's Project Director. The Contractor's Project Director or an approved alternate will be assigned locally and available for telephone contact 24 hours a day, Monday through Sunday, including all County holidays. The Contractor's Project Director will provide overall management and coordination of the Contract services on Contractor's behalf, and will act as the central point of contact with County.
- 6.2.2 When Contract work is being performed at times other than described above or when the Contractor's Project Director cannot be present, and with prior approval of the County's Program Manager, an equally responsible individual will be designated to act for Contractor's Project Director.
- 6.2.3 The Contractor's Project Director will have a minimum of three (3) years of experience within the last five (5) years in the management and operation of electronic monitoring services or similar scope.

- 6.2.4 The Contractor's Project Director/alternate will have full authority to act for Contractor on all matters relating to the daily operation of the Contract. The Contractor's Project Director/alternate will be able to effectively communicate in English, both orally and in writing.
- 6.2.5 The Contractor's Project Director must be available between 8:00 a.m. to 5:00 p.m., PT, Monday through Friday excluding County holidays, to meet with County personnel designated by County to discuss problem areas.
- 6.2.6 County will have the right of review and approval of the Contractor's Project Director. County will have the right of removal of the Contractor's Project Director and any replacement recommended by Contractor.

6.3 Contractor Personnel

- 6.3.1 Contractor will be responsible for providing competent staff to fulfill the Contract. County will have the right to review and approve potential staff prior to their performing services under this Contract.
- 6.3.2 Contractor will ensure that by the first day of employment, all persons working on this Contract have signed an acknowledgment form regarding confidentiality that meets the standards of the County of Los Angeles Probation Department for County employees having access to confidential Criminal Offender Record Information (CORI) Contractor shall retain the original CORI form and forward a copy to the County's Program Manager within five (5) business days of start of employment. The CORI form is provided at Technical Exhibit 3 (Confidentiality CORI Information) of Appendix B (Statement of Work Technical Exhibits).
- 6.3.3 All personnel must be able to read, write, spell, speak, and understand English.
- 6.3.4 County reserves the right to preclude Contractor staff from performing services under this Contract. Contractor will be responsible for immediately removing and replacing any employee from work on this Contract within twenty-four (24) hours after it is requested to do so by County's Program Manager.
- 6.3.5 County reserves the right to have County's Program Manager or a designated alternate, interview any or all prospective employees of Contractor.

- 6.3.6 Contractor will be required to conduct a background check of their employees as set forth in Paragraph 7.5 (Background and Security Investigations) of the Contract.
- 6.3.7 Contractor must have alternate staff that have successfully passed background clearances pursuant to Paragraph 7.5 (Background and Security Investigations) of the Contract.

6.3.8 <u>CONTRACTOR STAFFING LEVEL REQUIREMENTS AND QUALIFICATIONS</u>

- 6.3.8.1 Contractor will provide training annually, for all Contractor staff working in the monitoring center to ensure that they have sufficient knowledge regarding the system to allow them to effectively respond to questions and to fully utilize the system and equipment. Contractor will provide at least forty (40) hours of training before authorized to provide services. Additionally, Contractor will liaison with, and will maintain a good working relationship with the judiciary, law enforcement, criminal justice system, County staff and the community.
- 6.3.8.2 Contractor will provide a substitute for terminated personnel that are deemed critical by the County in meeting acceptable performance per the contract and where the Contractor has been notified of the critical nature of the specific position by the County and/or positions that are contractually specified positions within forty-eight (48) hours. The noted vacant position must be filled with a permanent employee within thirty (30) days.
- 6.3.8.3 Contractor will provide quality improvement training to their staff and conduct quarterly assurance activities to ensure the data entered in the software is accurate.
- 6.3.8.4 Contractor will develop written operational procedures specific to County's Comprehensive Electronic Monitoring Services for reference by their staff.

6.4 Contractor Furnished Items

6.4.1 Contractor will provide all personnel and supplies necessary to perform all services required by the Statement of Work.

6.5 Contractor's Office

Contractor will maintain an office with a telephone in the company's name where Contractor conducts business. The office must be staffed during the hours of 8:00 a.m. to 5:00 p.m., Monday through Friday, Pacific Time, by at least one employee who can respond to inquiries and complaints which may be received about Contractor's performance of the Contract. When the office is closed, an answering service will be provided to receive calls. Contractor must respond to calls received by the answering service within two (2) hours of receipt of the call.

6.6 Identification Badges

Contractor must ensure their employees are appropriately identified as set forth in Paragraph 7.4 (Contractor's Staff Identification) of the Contract.

7.0 HOURS/DAYS OF WORK

Contractor will be required to provide Comprehensive Electronic Monitoring Services on a twenty-four (24) hours continuous basis.

8.0 UNSCHEDULED WORK

If Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this Contract, the same shall be deemed to be a gratuitous effort on the part of Contractor, and Contractor will have no claim whatsoever against County.

9.0 PERFORMANCE REQUIREMENTS SUMMARY

- 9.1 All listings of services used in the Performance Requirements Summary (PRS) are intended to be completely consistent with the Contract and the Statement of Work (SOW), and are not meant in any case to create, extend, revise, or expand any obligation of Contractor beyond that defined in the Contract and the SOW. In any case of apparent inconsistency between services as stated in the Contract and the SOW and the PRS, the meaning apparent in the Contract and the SOW will prevail. If any service seems to be created in the PRS which is not clearly and forthrightly set forth in the Contract and the SOW, that apparent service will be null and void and place no requirement on Contractor.
- 9.2 A standard level of performance will be required of Contractor for the required services. Technical Exhibit 1 (Performance Requirements Summary Chart) of Appendix B (Statement of Work Technical Exhibits) summarizes the required services, performance standards, maximum allowable deviation from the standards, methods of surveillance to be used

by the County, and liquidated damages to be imposed for unacceptable performance. County will evaluate Contractor's performance under this Contract using the quality assurance procedures specified in Technical Exhibit 1 (Performance Requirements Summary Chart) of Appendix B (Statement of Work Technical Exhibits) or other such procedures as may be necessary to ascertain Contractor compliance with this Contract. Failure of Contractor to achieve this standard can result in an assessment of liquidated damages against Contractor's monthly payment as determined by County.

- 9.3 When Contractor's performance does not conform to the requirements of this Contract, County will have the option to apply the following non-performance remedies:
 - 9.3.1 Require Contractor to implement a formal corrective action plan, subject to approval by County. In the plan, Contractor must include reasons for the unacceptable performance, specific steps to return performance to an acceptable level, and monitoring methods to prevent recurrence.
 - 9.3.2 Reduce payment to Contractor by a computed amount based on the assessment fee(s) in the PRS.
 - 9.3.3 Reduce, suspend or cancel this Contract for systematic, deliberate misrepresentations or unacceptable levels of performance.
 - 9.3.4 Failure of Contractor to comply with or satisfy the request(s) for improvement of performance or to perform the neglected work specified within ten (10) business days shall constitute authorization for County to have the service(s) performed by others. The entire cost of such work performed by others as a consequence of Contractor's failure to perform said service(s), as determined by County, shall be credited to County on Contractor's future invoice.

This Subparagraph does not limit County's right to terminate the Contract upon ten (10) business days' written notice with or without cause, as provided for in Paragraph 8.42 (Termination for Convenience) of the Contract.

/
/

APPENDIX B STATEMENT OF WORK TECHNICAL EXHIBITS

APPENDIX B STATEMENT OF WORK TECHNICAL EXHIBITS

TABLE OF CONTENTS

<u>TE</u>	TECHNICAL EXHIBITS	
1.	PERFORMANCE REQUIREMENTS SUMMARY (PRS) CHART	1
2.	CONTRACT DISCREPANCY REPORT	2
3.	CONFIDENTIALITY OF CORI INFORMATION	3
4	EQUIPMENT REQUIREMENTS	4
5	INFORMATION SECURITY AND PRIVACY REQUIREMENTS	11
6	CEMS IMPLEMENTATION REQUIREMENTS	16
7	PROBATION DEPARTMENT'S ELECTRONIC MONITORING PROGRAM (E OPERATIONAL PROCEDURES MANUAL	

TECHNICAL EXHIBIT 1 PERFORMANCE REQUIREMENTS SUMMARY (PRS) CHART

SERVICE LEVEL REFERENCE	SERVICE LEVEL COMPLIANCE DESCRIPTION	NON-COMPLIANCE DEDUCTIONS/FEES TO BE ASSESSED
SOW 1.0	CEMS system (System) shall be available 99.96% of the time other than planned downtime. System availability for any month shall be defined as the number of minutes the System is accessible to County Designated users via the web divided by the total number of minutes in the month.	For any month in which System availability is less than 99.96% excluding planned downtime, the following fees shall be assessed. • Between 98% and 99.95% fee of \$1,000 • Between 95% and 98% feel of \$2,000 • Between 90% and 95%, fee of \$3,000 • Less than 90%, feel of \$5,000
SOW 1.1.1	Telephone technical support is available seven (7) days a week twenty-four (24) hours per day.	\$100 for each occurrence in which technical support is not available via a telephone call.
SOW 1.7 and 2.2.2.2	Contractor shall deliver standardized preconfigured reports to the County at agreed upon dates and times. All reports shall also have the ability to be queried and generated thought the Contractor's database.	\$100 for each occurrence of a report delivered to the County for each day after the agreed upon day and schedule.
SOW 2.1, 2.2 and 2.3	Contractor shall provide one piece GPS devices that are Radio Frequency (RF) capable along with all software necessary to track offenders as outlined in the SOW	\$100 for each occurrence.
SOW 2.2.1.1, 2.2.1.4, and 2.3.1	Contractor shall provide installation, removal, resets and replacement services of all electronic monitoring equipment within established time frames as specified by the County	\$100 for each occurrence beyond the established time frames.
Contract 7.5.6	Contractor shall reimburse County for criminal record check of Contractor's and Subcontractor's employees.	Up to \$100 per occurrence.

TECHNICAL EXHIBIT 2

CONTRACT DISCREPANCY REPORT

TO:				
FROM:				
DATES:	Prepared:			
	Returned by Contractor:			
	Action Completed:			
DISCREPAN	NCY PROBLEMS:			
Signature of	County Representative	Date		
CONTRACT	OR RESPONSE (Cause and Corrective Action):			
Signature of	Contractor Representative	Date		
COUNTY EV	COUNTY EVALUATION OF CONTRACTOR RESPONSE:			
Signature of	County Representative	Date		
001111777				
COUNTY AC	CTIONS:			
CONTRACT	OR NOTIFIED OF ACTION:			
County Door	resentative's Signature and Dato			
соину керг	resentative's Signature and Date			
Contractor D	Conresentative's Signature and Data			
Contractor R	epresentative's Signature and Date			

TECHNICAL EXHIBIT 3

CONFIDENTIALITY OF CORI INFORMATION

Criminal Offender Record Information (CORI) is that information which is recorded as the result of an arrest, detention or other initiation of criminal proceedings including any consequent proceedings related thereto. As an employee of during the legitimate course of your duties, you may have access to CORI. The Probation Department has a policy of protecting the confidentiality of Criminal Offender Record Information.			
You are required to protect the information contained in case files against disclosure to all ndividuals who do not have a right-to-know or a need-to-know this information.			
The use of any information obtained from case files or other related sources of CORI to make contacts with probationers or their relatives, or to make CORI available to anyone who has no real and proper reason to have access to this information as determined solely by the Probation Department is considered a breach of confidentiality, inappropriate and unauthorized.			
Any employee engaging in such activities is in violation of the Probation Department's confidentiality policy and will be subject to appropriate disciplinary action and/or criminal action pursuant to Section 11142 of the Penal Code.			
I have read and understand the Probation Department's policy concerning the confidentiality of CORI records.			
(Signature)			
Name (Print)			
Title			
Date			
Copy to be forwarded to County Program Manager within five (5) business days of start of employment.			

Page 1 of 7

1.0 EQUIPMENT REQUIREMENTS

- 1.1 Contractor shall provide only equipment that meets the highest levels of ruggedness, durability and performance. The GPS device shall be a one-piece, single body worn device and meet the following minimum requirements:
 - 1.1.1 The device shall be currently registered and approved by the Federal Communications Commission (FCC) including accessories/beacons.
 - 1.1.2 Contractor shall be the owner and Original Equipment Manufacturer (OEM) of the GPS one-piece tracking device and of the related electronic monitoring tracking software and shall have an ISO 9001 certification.
 - 1.1.3 Equipment and systems shall remain the property of Contractor. Contractor shall be responsible for upgrades, maintenance and repair/replacement costs.
 - 1.1.4 The device shall have an identification label identifying Contractor and which displays a central telephone number.
 - 1.1.5 The device shall be the latest proven reliable technology available and the latest in use from the manufacturer and the current product submitted with the Contract for use.
 - 1.1.6 The device shall meet market safety standards and present no health or safety hazards to staff and/or offenders.
 - 1.1.7 The device shall not unduly restrict participants' activities. The device shall be enclosed in a sealed hypoallergenic case that is resistant to shock, water, corrosion, chemical solvents, and detergents. The device shall function reliably under normal environmental conditions.
 - 1.1.8 The device shall not have any sharp edges and shall be designed to cause no excessive chafing or bruising.
 - 1.1.9 The device shall be small, light, and non-restrictive. The device shall attach around the ankle of the offender and weigh no more than ten (10) ounces for a fully installed device. Fully installed device shall include the weight inclusive of all clips, straps, case with all batteries, as required for a fully functioning active GPS one-piece device installed on a participant.

Page 2 of 7

- 1.1.10 The device shall be waterproof.
- 1.1.11 The device shall be shockproof.
- 1.1.12 The device shall be designed to prevent tracing or duplication of the signal by other electronic devices or equipment.
- 1.1.13 The device shall be marked with the model and/or serial identification numbers and will not change colors or lose labeling after being exposed to common cleaning products.
- 1.1.14 The device shall possess non-erasable program identification.

2.0 EQUIPMENT TRACKING REQUIREMENTS

- 2.1 The device shall at a minimum use GPS and the cellular telephone network to track and report the location(s) of offenders both at home and in the community, regardless of location.
- 2.2 The device shall track an offender on a twenty-four (24) hour continuous basis, and shall be able to confirm the date, time and location of the tracking event.
- 2.3 The device shall have the ability to convert between Active GPS, and Enhanced Cellular Radio Frequency mode without any removal of the device from the participant or otherwise requiring a change of equipment.
- 2.4 The Active GPS device configuration shall collect a tracking point at least once every minute, and report the information via the cellular network at least once every ten (10) minutes and shall report tampering, zone, and battery violations immediately.
- 2.5 The device shall be remotely contacted to force the device to instantly locate and call back with its data immediately (showing points each one-minute interval).
- 2.6 The device shall have internal diagnostics that can determine if it is operating properly and the ability to relay the information to the Central Information software system.
- 2.7 The device shall acquire GPS within two (2) minutes when placed in an outdoor environment.
- 2.8 The device shall be programmable to vary the rate of recording and reporting of GPS position data.

Page 3 of 7

- 2.9 The device shall record the "in range" and "out of range" times for CRF devices.
- 2.10 The device shall adjust Reporting and Leave Windows (schedules) for CRF devices.

3.0 EQUIPMENT COMMUNICATION REQUIREMENTS

- 3.1 The device communication shall be encrypted.
- 3.2 The device shall provide Multi-Network Roaming that communicates to the software.
- 3.3 The device shall provide alternative location tracking using the cellular network in the absence of GPS.
- 3.4 The tracking device shall have the option of communicating with a landline telephone based receiver to upload location and alert information to the software for areas not covered by cellular service without removing the device from the offender.
- 3.5 The device shall incorporate non-volatile memory that stores at least ten (10) days' worth of events (with date and time of occurrence) at a collection rate of one (1) point per minute, at times when the cellular service and electrical power may become unavailable and continually attempt to report these events until reporting has been successful. Non-volatile memory shall retain unreported events and report them once power/cellular services have been restored, including date & time of occurrence.

4.0 BATTERY REQUIREMENTS

- 4.1 The battery life of the tracking device must either be sufficient to last throughout the entire term of the participant's sentence or the battery must be replaced within twenty-four (24) hours of a "low battery" alarm received, prior to battery failure.
- 4.2 The battery for the device shall hold a single charge for a minimum period of twenty-four (24) hours while performing one (1) minute GPS acquisition and downloading data to the software at least once every ten (10) minutes without the use of additional accessories such as beacons.
- 4.3 The device shall re-charge the battery to the maximum capacity within two (2) hours from a low battery status.

Page 4 of 7

- 4.4 The device shall re-charge the battery to the maximum capacity within four (4) hours from a complete dead battery status.
- 4.5 The device batteries shall be permanently sealed within the device case, and shall not include any batteries that are removable and replaceable.
- 4.6 The device shall send a signal indicating a low device battery condition at least eight (8) hours prior to battery failure.
- 4.7 The device battery shall have a guaranteed life cycle of not less than twelve (12) months.
- 4.8 The device shall be equipped with an electronic charger device that uses a wall electrical outlet power (alternating current) with a charging cord of a minimum of six (6) feet long and a separate vehicle charger (direct current) source to recharge the battery.

5.0 ALERT/NOTIFICATIONS REQUIREMENTS

The System shall provide a means of notification (alert) to the offender when the participant is late returning home from an approved absence or leaves home when unscheduled. Contractor's monitoring center staff shall call the participant based on protocols determined by County.

- 5.1 The device shall communicate to the offender by vibration of the device for low battery alerts and allow the participant to contact the supervising DPO.
- 5.2 The device shall store up to one hundred fifty (150) zones onboard the device to ensure zone violations are immediately reported regardless of the call in frequency.
- 5.3 The device shall detect, record, and report an alert feature with immediate notification including strap tampers, removal tamper, loss of communication, loss of a GPS signal, low battery, and critical /shutdown battery.
- 5.4 The device shall detect, record, and report an alert feature with immediate alert notification for inclusion and exclusionary zone violations.
- 5.5 The device inclusion and exclusionary zones shall have time zone perimeter capability.
- 5.6 The device shall have an alert feature with an immediate alert notification to detect metallic shielding of the device's ability to receive a GPS signal. The device shall detect, record, and report such an instance of shielding.

Page 5 of 7

5.7 The device shall have an alert feature with an immediate alert notification to detect, record, and report when there is interference or jamming signals present that are disrupting the receipt of GPS signals.

6.0 STRAP REQUIREMENTS

- 6.1 All device straps utilized to attach devices to a participant shall be disposable.
- 6.2 Straps used to attach the devices to a participant shall have exterior surfaces made of hypoallergenic materials.
- 6.3 Straps shall be adjustable in length to fit all participants. All straps shall have the capability to be securely sized and attached to an offender. If straps are pre-sized they shall be immediately available in one-half inch increments or less.
- 6.4 Contractor shall provide straps (clips, snaps, pins, and connectors, etc.), designed to breakaway and become unusable, to prevent damage to the device or participant if the device/strap is removed by the participant.

7.0 ADDITIONAL ACCESSORY REQUIREMENTS

- 7.1 The one-piece GPS device shall be able to utilize additional accessories (e.g., beacons, etc.) for enhanced location verification while in defined impaired environments (such as inside home or buildings).
- 7.2 The Home Based Unit (HBU)/beacon shall detect movement and, shall differentiate momentary movement (such as cleaning around the unit) versus perpetual movement (such as relocation). The HBU/beacon shall be able to communicate an alert of perpetual movement.
- 7.3 The HBU/beacon shall utilize a dedicated wall electrical outlet (alternating current) for its power source, or a battery for its primary source of power.
- 7.4 The HBU/beacon shall detect and report events related to the disconnecting and reconnecting of the electrical power and or telephone line.
- 7.5 In the event of power disconnection or commercial outage, the accessory shall have a minimum of forty-eight (48) hours internal, auto-recharging back-up battery to support completed continued functionality, including but not limited to detecting and reporting information.

Page 6 of 7

- 7.6 The HBU/beacon shall be replaceable and interchangeable without the need to return to Contractor for installation or removal.
- 7.7 The HBU/beacon shall conduct a simple and rapid range test by the DPO to confirm signal coverage between the device and accessory at the participant's home.
- 7.8 The HBU/beacon shall be programmable with adjustable signal range.
- 7.9 The HBU/beacon shall detect when the accessory case has been compromised.

8.0 EQUIPMENT QUALIFICATIONS

- 8.1 Devices with the technology to automatically switch between multiple cellular providers in order to reduce or prevent loss or delay of GPS points due to cellular communication failure that interferes with transmission of data to Contractor's software are highly desirable, but not required.
- 8.2 All devices assigned to the participant shall be manufactured to allow for repeated proper sanitization. Contractor shall provide instructions to sanitize the devices, including recommended cleaning agents and methods.
- 8.3 Contractor shall replace each device at specific intervals, as defined by County, to avoid device failures.

9.0 DELIVERED EQUIPMENT REQUIREMENTS

- 9.1 All equipment delivered to County shall have been tested to ensure it is fully operable and free from any defects or damage prior to shipment to County. County reserves the right to reject any and all equipment it deems unacceptable, inoperable, and/or not in the best interest of public safety.
- 9.2 In the event any equipment becomes inoperable for any reason Contractor shall incur all repair and replacement costs. Contractor shall provide County with replacement equipment within twenty-four (24) hours of request. Contractor shall pay all costs associated with shipping (both delivery and return) of all equipment and supplies.
- 9.3 County shall determine the method, frequency, and location of all equipment deliveries.

Page 7 of 7

- 9.4 Upon approval by the County Program Manager, Contractor shall furnish upgraded equipment with improved technology to County. If it is determined that upgraded equipment is to replace existing equipment, the County Program Manager and Contractor shall develop an agreed upon implementation plan that will include a systematic process to ensure the completion of transition.
- 9.5 County shall not incur any costs for any delivered equipment and supplies that are altered, damaged, stolen, lost, tampered, misplaced, and/or misused.

10.0 INSTALLATION KIT

County shall have the ability to install and remove GPS devices. Devices shall come with an installation and removal kit containing all necessary devices to install, remove, activate or deactivate the device including a quick installation user guide.

/ / / / / / /

Page 1 of 5

This Exhibit (Information Security and Privacy Requirements) is an attachment and addition
to the Risk Management Information Services Agreement dated (the
"Agreement") entered into by and between the County of Los Angeles ("County") and
("Contractor") and is incorporated into the Agreement by reference
hereof. This Exhibit (Information Security and Privacy Requirements) sets forth information
security procedures to be established by Contractor before the effective date of the Agreement
and maintained throughout the term of the Agreement. These procedures are in addition to the
requirements of the Agreement between the Parties. They present a minimum standard only. It
is Contractor's sole obligation to: (i) implement appropriate administrative, physical and
technical measures to secure its systems and data to protect and ensure the privacy,
confidentiality, integrity and availability of County Data as defined in Section the
Agreement (consisting of but not limited to County Confidential Information, Personally
Identifiable Information, and Protected Health Information) against internal and external threats,
vulnerabilities and risks; and (ii) continuously review and revise those measures to address
ongoing threats, vulnerabilities and risks. Failure to comply with the minimum standards set
forth in this Exhibit (Information Security and Privacy Requirements) will constitute a
material, non-curable breach of the Agreement by Contractor, entitling County, in addition to
and cumulative of all other remedies available to it at law, in equity, or under the Agreement, to
immediately terminate the Agreement.
,

- Security Policy. Contractor shall establish and maintain a formal, documented, mandated, company-wide information security program, including security policies, standards and procedures (collectively "Information Security Policy"). The Information Security Policy will be communicated to all Contractor personnel, agents and subcontractors in a relevant, accessible, and understandable form and will be regularly reviewed and evaluated to ensure its operational effectiveness, compliance with all applicable laws and regulations, and to address new threats and risks.
- 2. Personnel and Contractor Protections. Contractor shall screen and conduct background checks on all Contractor personnel exposed to County Confidential Information as defined in Section ____ of the Agreement and require all employees and contractors to sign an appropriate written confidentiality/non-disclosure agreement. All agreements with third-parties involving access to Contractor's systems and data, including all outsourcing arrangements and maintenance and support agreements (including facilities maintenance), shall specifically address security risks, controls, and procedures for information systems. Contractor shall supply each of its Contractor personnel with appropriate, ongoing training regarding information security procedures, risks, vulnerabilities and threats. Contractor shall have an established set of procedures to ensure Contractor personnel promptly report actual and/or suspected breaches of security.
- 3. **Removable Media**. Except in the context of Contractor's routine back-ups or as otherwise specifically authorized by County in writing, Contractor shall institute strict administrative, physical and logical security controls to prevent transfer of County information to any form of Removable Media. For purposes of this Exhibit ___ (Information Security and Privacy Requirements), "Removable Media" means portable or removable hard disks, floppy disks, USB memory drives, zip disks, optical disks, CDs, DVDs, digital film, digital cameras, memory cards (e.g., Secure Digital (SD), Memory Sticks (MS), CompactFlash (CF), Smart Media (SM), Multimedia Card (MMC), and xD-Picture Card (xD)), magnetic tape, and all other removable data storage media.

Page 2 of 5

- Storage, Transmission, and Destruction of Personally Identifiable Information and 4. Protected Health Information. All Personally Identifiable Information and Protected Health Information shall be rendered unusable, unreadable, or indecipherable to unauthorized individuals in accordance with HIPAA, as amended and supplemented by the HITECH Act and the California Civil Code section 1798 et seg. Without limiting the generality of the foregoing, Contractor shall encrypt (i.e., National Institute of Standards and Technology (NIST) Special Publication (SP) 800-111 Guide to Storage Encryption Technologies for End User Devices¹) all Personally Identifiable Information and electronic Protected Health Information (stored and during transmission) in accordance with HIPAA and the HITECH Act, as implemented by the U.S. Department of Health and Human Services. If Personally Identifiable Information and Protected Health Information is no longer required to be retained by Contractor under the Agreement and applicable law, Contractor shall destroy such Personally Identifiable Information and Protected Health Information by: (a) shredding or otherwise destroying paper, film, or other hard copy media so that the Personally Identifiable Information and Protected Health Information cannot be read or otherwise cannot be reconstructed; and (b) clearing, purging, or destroying electronic media containing Personally Identifiable Information and Protected Health Information consistent with National Institute of Standards and Technology (NIST) Special Publication (SP) 800-88, Guidelines for Media Sanitization 2 and US Department of Defense (DOD) 5220.22-M data sanitization and clearing directive³ such that the Personally Identifiable Information and Protected Health Information cannot be retrieved.
- Data Control, Media Disposal and Servicing. Subject to and without limiting the requirements 5. under Section 4 (Storage, Transmission and Destruction of Personally Identifiable Information and Protected Health Information), County Data (i) may only be made available and accessible to those parties explicitly authorized under the Agreement or otherwise expressly Approved by County in writing; (ii) if transferred across the Internet, any wireless network (e.g., cellular, 802.11x, or similar technology), or other public or shared networks, must be protected using industry standard encryption technology in accordance with the NIST SP 800-52 Guidelines for the Selection and use of Transport Layer Security Implementations⁴; and (iii) if transferred using Removable Media (as defined above) must be sent via a bonded courier or protected using industry standard encryption technology in accordance with NIST SP 800-111 Guide to Storage Encryption Technologies for End User Devices⁵. The foregoing requirements shall apply to backup data stored by Contractor at off-site facilities. In the event any hardware, storage media, or Removable Media must be disposed of or sent off-site for servicing, Contractor shall ensure all County Confidential Information, including Personally Identifiable Information and Protected Health Information, has been cleared, purged, or scrubbed from such hardware and/or media using industry best practices in accordance with NIST SP 800-88, Guidelines for Media Sanitization⁶).
- 6. **Hardware Return**. Upon termination or expiration of the Agreement or at any time upon County's request, Contractor will return all hardware, if any, provided by County containing Personally Identifiable Information, Protected Health Information, or County Confidential Information to County. The Personally Identifiable Information, Protected Health Information, and County Confidential Information shall not be removed or altered in any way. The hardware should be physically sealed and returned via a bonded courier or as otherwise directed by County. In the event the hardware containing County Confidential Information or Personally Identifiable Information is owned by Contractor or a third-party, a notarized statement, detailing the

¹ Available at http://www.csrc.nist.gov/

² Available at http://www.csrc.nist.gov/

³ Available at http://www.dtic.mil/whs/directives/corres/pdf/522022MSup1.pdf

⁴ Available at http://www.csrc.nist.gov/

⁵ Available at http://www.csrc.nist.gov/

⁶ Available at http://www.csrc.nist.gov/

Page 3 of 5

destruction method used and the data sets involved, the date of destruction, and the company or individual who performed the destruction will be sent to a designated County security representative within fifteen (15) days of termination or expiration of the Agreement or at any time upon County's request. Contractor's destruction or erasure of Personal Information and Protected Health Information pursuant to this Section shall be in compliance with industry Best Practices (e.g., NIST Special Publication 800-88, Guidelines for Media Sanitization⁷).

- 7. **Physical and Environmental Security**. Contractor facilities that process County Data will be housed in secure areas and protected by perimeter security such as barrier access controls (e.g., the use of guards and entry badges) that provide a physically secure environment from unauthorized access, damage, and interference.
- 8. **Communications and Operational Management**. Contractor shall: (i) monitor and manage all of its information processing facilities, including, without limitation, implementing operational procedures, change management and incident response procedures; and (ii) deploy adequate anti-viral software and adequate back-up facilities to ensure essential business information can be promptly recovered in the event of a disaster or media failure; and (iii) ensure its operating procedures will be adequately documented and designed to protect information, computer media, and data from theft and unauthorized access.
- 9. **Access Control**. Contractor shall implement formal procedures to control access to its systems, services, and data, including, but not limited to, user account management procedures and the following controls:
 - a. Network access to both internal and external networked services shall be controlled, including, but not limited to, the use of properly configured firewalls;
 - b. Operating systems will be used to enforce access controls to computer resources including, but not limited to, authentication, authorization, and event logging;
 - c. Applications will include access control to limit user access to information and application system functions; and
 - d. All systems will be monitored to detect deviation from access control policies and identify suspicious activity. Contractor shall record, review and act upon all events in accordance with incident response policies set forth below.
- 10. **Security Incident.** A "Security Incident" shall have the meaning given to such term in 45 C.F.R. § 164.304.
 - a. Contractor will promptly notify (but in no event more than twenty-four (24) hours after the detection of a Security Incident) the designated County security contact by telephone and subsequently via written letter of any potential or actual security attacks or Security Incidents.
 - b. The notice shall include the approximate date and time of the occurrence and a summary of the relevant facts, including a description of measures being taken to address the occurrence. A Security Incident includes instances in which internal

⁷ Available at http://www.csrc.nist.gov/

Page 4 of 5

personnel access systems in excess of their user rights or use the systems inappropriately.

- c. Contractor will provide a monthly report of all Security Incidents noting the actions taken. This will be provided via a written letter to the County security representative on or before the first (1st) week of each calendar month. County or its third-party designee may, but is not obligated, perform audits and security tests of Contractor's environment that may include, but are not limited to, interviews of relevant personnel, review of policies, procedures and guidelines, and other documentation, or technical inspection of systems, as they relate to the receipt, maintenance, use, retention, and authorized destruction of County Data.
- d. In the event County desires to conduct an unannounced penetration test, County shall provide contemporaneous notice to Contractor's Vice President of Audit, or such equivalent position. Any of County's regulators shall have the same right upon request. Contractor shall provide all information reasonably requested by County in connection with any such audits and shall provide reasonable access and assistance to County or its regulators upon request. Contractor agrees to comply with all reasonable recommendations that result from such inspections, tests, and audits within reasonable timeframes. County reserves the right to view, upon request, any original security reports that Contractor has undertaken on its behalf to assess Contractor's own network security. If requested, copies of these reports will be sent via bonded courier to the County security contact. Contractor will notify County of any new assessments.
- 11. **Contractor Self Audit.** Contractor will provide to County a summary of: (1) the results of any security audits, security reviews, or other relevant audits listed below, conducted by Contractor or a third-party as applicable; and (2) the corrective actions or modifications, if any, Contractor will implement in response to such audits.

Relevant audits conducted by Contractor as of the Effective Date include:

- a. ISO 27001:2013 (Information Security Management) or FDA's Quality System Regulation, etc. – Contractor-Wide. A full recertification is conducted every three (3) years with surveillance audits annually.
 - External Audit Audit conducted by non-Contractor personnel, to assess Contractor's level of compliance to applicable regulations, standards, and contractual requirements.
 - ii. **Internal Audit –** Audit conducted by qualified Contractor Personnel (or contracted designee) not responsible for the area of review, of Contractor organizations, operations, processes, and procedures, to assess compliance to and effectiveness of Contractor's Quality System ("**CQS**") in support of applicable regulations, standards, and requirements.
 - iii. **Supplier Audit –** Quality audit conducted by qualified Contractor Personnel (or contracted designee) of product and service suppliers contracted by Contractor for internal or Contractor client use.

Page 5 of 5

- iv. **Detailed findings** are not published externally, but a summary of the report findings, and corrective actions, if any, will be made available to County as provided above and the ISO certificate is published on Contractor's website.
- b. SSAE-16 (formerly known as SAS -70 II) As to the Hosting Services only:
 - i. Audit spans a full twelve (12) months of operation and is produced every six (6) months (end of June, end of December) to keep it "fresh".
 - ii. The resulting detailed report is available to County.

Detailed findings are not published externally, but a summary of the report findings, and corrective actions, if any, will be made available to County as provided above.

12. Security Audits. In addition to the audits described in Section 10 (Contractor Self Audit), during the term of the Agreement, County or its third-party designee may annually, or more frequently as agreed in writing by the Parties, request a security audit (e.g., attestation of security controls) of Contractor's data center and systems. The audit will take place at a time mutually agreed to by the Parties, but in no event on a date more than ninety (90) days from the date of the request by County. County's request for security audit will specify the areas (e.g., Administrative, Physical and Technical) that are subject to the audit and may include but not limited to physical controls inspection, process reviews, policy reviews, evidence of external and internal vulnerability scans, penetration tests results, evidence of code reviews, and evidence of system configuration and audit log reviews. County shall pay for all third-party costs associated with the audit. It is understood that summary data of the results may filtered to remove the specific information of other Contractor customers such as IP address, server names, and others. Contractor shall cooperate with County in the development of the scope and methodology for the audit, and the timing and implementation of the audit. Any of the County's regulators shall have the same right upon request, to request an audit as described above. Contractor agrees to comply with all reasonable recommendations that result from such inspections, tests, and audits within reasonable timeframes.

> / / / / / /

TECHNICAL EXHIBIT 6 CEMS IMPLEMENTATION REQUIREMENTS

1. CEMS Center Software and Hardware Setup:

CEMS center system environment setup, including system and hardware configuration.

2. CEMS Software and Hardware Installation:

Installation and configuration of CEMS software, hardware, and interfaces, if any.

3. Data Conversion and Migration:

Conversion and migration of data from any existing systems/databases, as specified by the Department.

4. Systems Acceptance Testing:

a. Integration Testing

Testing of CEMS system's integration with other County systems, if any, remote terminals, and all monitoring equipment.

b. User Testing

Testing by County and Contractor users of the applicable components of CEMS.

5. Transition to Operational Environment:

Transition services for cutover from the existing system/database to the CEMS operational/production environment._

6. Final Acceptance:

Contractor shall maintain CEMS in operational environment for forty-five (45) days following the cutover ("Warranty Period"). Contractor shall correct any and all deficiencies discovered during such Warranty Period before CEMS can achieve Final Acceptance and before Contractor may be paid by County for services provided hereunder.

TECHNICAL EXHIBIT 7 PROBATION DEPARTMENT'S ELECTRONIC MONITORING (EM) PROGRAM OPERATIONAL PROCEDURES MANUAL



County of Los Angeles Probation Department Pretrial Services Bureau 450 Bauchet Street Los Angeles, CA 90012 (213) 893-5369



COUNTY OF LOS ANGELES PROBATION DEPARTMENT ADULT ELECTRONIC MONITORING PROGRAM

OPERATIONAL PROCEDURES MANUAL

TECHNICAL EXHIBIT 7 PROBATION DEPARTMENT'S ELECTRONIC MONITORING (EM) PROGRAM OPERATIONAL PROCEDURES MANUAL

TABLE OF CONTENTS

INTROD	OCTION	IV
PART I -	PRETRIAL SERVICES CASE PROCESSING	
	Referral Process	2
	Application Types	2
	Risk Assessment	3
	Contractor Notification	3
	Program Marketing/Awareness	3
	Court Referrals	4
	County Jail Referrals	4
	Civil Referrals	5
	Assessment Process	6
PART II	- CONTRACTOR CASE PROCESSING	
	Enrollment Date Requirements	9
	Participant Enrollment	9
	Participant Orientation	10
	Setting Up Case File	10
	Setting Curfew Schedules	12
	Work Hours	12
	Court authorized activities	13
	Installation and Hook-up	14
	Daily Contact List	14

TECHNICAL EXHIBIT 7 PROBATION DEPARTMENT'S ELECTRONIC MONITORING (EM) PROGRAM OPERATIONAL PROCEDURES MANUAL

TABLE OF CONTENTS

Schedule Changes	15
Type of Equipment	15
Replacement/Repair of Equipment	16
Tamper Alarms	16
Reporting Guidelines	16
Types of Reports	17
Distribution	19
Monitoring Activity Reports	20
New Arrests	20
Participant Voluntarily Terminates	21
Defendant Walk-ins	21
Enrollments after Ineligible Recommendation	22
Participant Change of Address	22
Documentation	23
Office Meetings	24
Case Notes	24
Quality Control	25
Daily Reports	26
Definitions	27

APPENDICES

APPENDIX A:	EM Referral/Monitoring Forms	.29
APPENDIX B:	EM Assessment Report	.32
APPENDIX C:	Excluded Charges - Court Program	.33
APPENDIX D:	Excluded Charges – Sheriff's Program	.37
APPENDIX E:	Probation Approved Forms	.40
APPENDIX F:	EM Participant Contract	.47

INTRODUCTION

Contractor(s) providing electronic monitoring services for Los Angeles County on a daily twenty-four (24) hour continuous basis shall comply with the Los Angeles County Probation Department's Electronic Monitoring Program's Operational Procedures Manual (OPM).

The purpose of the standards and procedures presented in this manual, along with the Probation Department's screening process of referred participants and its auditing component, is to provide the highest level of service to the courts and maintain a high regard for community protection.

This manual will explain standards and procedures for the processing and the monitoring of program participants and the maintenance of documentation in case files.

There shall be no deviation from the standards and/or procedures contained in this manual without the prior written approval of the Electronic Monitoring Program's Steering Committee, which will be chaired by the Probation Department's Project Director.

The County will evaluate the contractor(s) contract performance using various quality assurance measurements. Compliance with the OPM's standards and procedures will be one of those measurements.

PART I

PRETRIAL SERVICES CASE PROCESSING

REFERRAL AND ASSESSMENT PROCESS

REFERRAL PROCESS

Participants are referred to Pretrial Services (PTS) to be evaluated for participation in the Electronic Monitoring (EM) Program by agencies or persons within the justice community.

PTS is responsible for evaluating participants to determine eligibility/suitability for program participation. This evaluation process will include:

- -- Interviewing defendant
- -- Accessing, interpreting and evaluating criminal history information
- Completing risk assessment scale
- -- Verifying the address, telephone and other information provided by the defendant
- -- Determination of eligibility/suitability for program participation
- Reporting defendant's eligibility/suitability for program participation to the Court,
 Deputy Probation Officer (DPO), Sheriff's Department, and to the Contractor

APPLICATION TYPES

There are three (3) types of EM Program referrals. The following is a list of codes used by PTS to identify each application type:

- ES Electronic Monitoring; referred by Sheriff (i.e., County Jail CJ cases)
- EC Electronic Monitoring; referred by court post-sentence
- ET Electronic Monitoring; referred by court pre-sentence

RISK ASSESSMENT

Once the evaluation has been completed, a risk level is assigned based on the point outcome of the EM Risk Assessment Scale:

- 0 13 <u>Standard Monitoring</u>
- 14 22 Enhanced Monitoring

CONTRACTOR NOTIFICATION

An Assessment is completed by PTS on all participants referred to EM and suitable Assessments are faxed or emailed to the Contractor. This Notification provides the following information:

- -- EM application number
- -- Court case number or booking number
- -- Offense(s)
- -- Defendant's name, address and telephone number, if applicable
- Assessment findings
- -- Risk assessment level (criminal offenders only)
- -- Probation status (includes "X" number, DPO, Probation Area Office and telephone number)
- -- Driver's license status (valid, suspended, revoked, restricted, or none)
- -- Primary language

PROGRAM MARKETING/AWARENESS

Marketing the EM Program is the responsibility of the contractors. The Probation Department will support the EM Program by educating relevant County agencies about the program's policies and procedures.

COURT REFERRALS

Prior to sentencing where the defendant has been convicted, pled guilty or found in violation of probation, the Court refers the defendant to be evaluated by PTS for suitability for the EMProgram.

The Court completes the Court Referral Form and faxes or emails it to EM staff at (213) 633-4684 or contacts PTS staff by telephone at (213) 893-5369 and provides the information listed on the Court Referral Form. The Court then instructs the defendant to contact PTS staff at (213) 893-5369 by the end of the next business day to complete an application. Instead of providing verbal instructions, the Court has the option to give the defendant a copy of the "EM Instructions to Defendant" form.

Participants with multiple cases referred by the court must have a separate EM APPLICATION NUMBER assigned to each court case number.

Criteria for referral

- -- Post- conviction misdemeanors and felonies
- Pre-sentenced misdemeanors and felonies
- -- Civil Contempt cases

Contractor Notification

PTS staff will fax or email the completed suitable Assessment to the Contractor. The Contractor will contact PTS regarding any enrollments or scheduled enrollments that do not have a completed suitable Assessments.

COUNTY JAIL REFERRALS

The Sheriff's Department's Community Based Alternatives to Custody (CBAC) personnel refer county jail inmates to be evaluated by PTS for determination of suitability for electronic monitoring release. CBAC Sheriff personnel have two (2) methods of referral to EM. The first method is by providing, each morning, an automated fully sentenced inmate list identified as potential candidates for EM consideration. The second method is by providing individual cases, such as special classified inmates, for EM evaluation. Beginning October 12th, 2004, LASD identified fully sentenced inmates with PC273.5 and PC243(e)1 convictions as a population potentially acceptable for EM. With concurrence between LASD Administration, and Probation Department Administration, PTS staff began screening fully sentenced PC273.5 and PC243(e)1 inmates for EM.

Only suitable assessed inmates by PTS will be released on electronic monitoring. All inmates assessed and determined to be ineligible or unsuitable will remain in custody. Sheriff personnel retain final release authority and may disqualify a suitable EM candidate if deemed necessary. The case turnaround time for completing County jail referrals is approximately forty-eight (48) hours. In some instances, cases may require additional time with clearance from the County Program Manager or designee.

Criteria for referral

- County jail inmates fully sentenced to county jail time on misdemeanor or felony cases
- -- Inmates cannot be in custody on any excluded conviction charges
- -- Inmates can have no pending felony cases
- Inmates can have no outstanding warrants
- -- Inmates can have no custody holds
- Inmates can have no rehabilitation orders
- -- Inmates can have no disqualified custody special handling restrictions

Contractor Notification

PTS staff will provide a copy of the completed suitable Assessment to the Contractor and Sheriff CBAC personnel. This will be done daily without exception. As inmates are approved, the names are given to the Contractor staff assigned to the jail location periodically throughout the day.

All suitable Assessments are faxed or emailed to the Contractor's enrollment center on a daily basis. There will be required communication from PTS staff and the Contractor's enrollment staff to verify proper notification on all suitable EM candidates.

CIVIL REFERRALS

On occasion, Superior Court will refer Family Law matters (civil contempt) to the EM Program. In most cases, the bench officers do not want an assessment. Referrals should be made from the court to PTS and will be handled in the same manner as criminal cases, except that there will be no criminal record check. In the event a Contractor should receive a referral directly from the court on a civil case, Contractor shall instruct the defendant to call PTS first. PTS will complete a partial application and will check the

address and telephone information only. The defendant will be given an EM application number and will be told to call the Contractor for enrollment.

ASSESSMENT PROCESS – COURT REFERRALS

I. REFERRAL

- A. Court completes the Court Referral Form and faxes or emails it to EM staff at (213) 633-4684,
- B. Court instructs offender to contact PTS staff at (213) 893-5369 by the end of the next business day to complete an application.

II. ASSESSMENT

- A. PTS staff completes an application when the defendant telephones.
- B. PTS staff accesses complete criminal record.
- C. PTS staff verifies address, telephone and other information provided by the defendant.
- D. PTS staff completes Risk Assessment scale.
- E. PTS staff evaluates information and completes an EM Assessment Report, which is sent to the Court by the next court date.

III. JUDICIAL DECISION FOR EM

- A. Defendant returns to court and court instructs the defendant regarding terms and conditions of EM and completes a Court Ordered Monitoring Conditions form (Appendix A).
- B. Court instructs offender to contact PTS at the number listed on the bottom of the Court Ordered Monitoring Conditions form (Appendix A).
- C. PTS and the Contractor notifies the court of defendant enrollment or failure to enroll in the program.

ASSESSMENT PROCESS – COUNTY JAIL REFERRALS

I. REFERRAL

A. The Sheriff's CBAC personnel refer county jail inmates to be evaluated by PTS.

Inmate's family or friends contact PTS by telephone at (213) 893-5369 and provide the booking or case information.

II. <u>ASSESSMENT</u>

- A. PTS staff completes an application after the defendant is interviewed by video-conference, telephone or in-person at the jail facility.
- B. PTS staff accesses complete criminal record.
- C. PTS staff verifies reference, address, telephone and other information provided by the defendant.
- D. PTS staff completes Risk Assessment scale.
- E. PTS staff evaluates information and completes an EM Assessment Report. Suitable Assessments are sent to the Contractor and the Sheriff's CBAC personnel.

PART II

CONTRACTOR CASE PROCESSING

ENROLLMENT DATE REQUIREMENTS

Court Component

The court will complete a Court Ordered Monitoring Conditions form indicating, among other things, the date by which a participant must be enrolled. Contractor must make every effort to enroll a participant by the date ordered. If this is not possible because, for example, the participant waited until the last day to be enrolled before contacting the Contractor, the Contractor will call the court, explain the circumstances, and ask for an extension.

County Jail Component

Inmates approved for participation by PTS are processed for release from custody into EM with the assistance of Sheriff and Contractor personnel assigned to work at the jail location. Inmates released for EM participation must be enrolled the same day they are released from jail. County jail inmates released to EM retain custody status until their sentence expires.

- A. Contractor personnel assigned to the jail will explain the program rules and regulations (Appendix F) to the inmate. EM installation instructions and the consequences of non-compliance are explained to the inmate. If the inmate agrees to abide by the program guidelines, he will sign a consent form; which becomes a part of the inmate's custody jacket and participant case file.
- B. Sheriff's Deputies will process the EM release, which includes a custody record and commitment document check. The inmate will be instructed to go directly home and report to a designated branch office for orientationthe following day.

PARTICIPANT ENROLLMENT

The Contractor will instruct the court ordered participant to bring in a copy of the Court Ordered Monitoring Conditions form, signed by the judge or court clerk. All participants will be instructed to provide picture identification, verification of employment, and a recent telephone bill verifying the telephone number and address of record. The enrollment process consists of program orientation, equipment installation and receipt of a "good hook-up" message. Note: County Jail participants will have GPS attached before release from custody; court participants not in-custody will have GPS attached by Contractor staff at enrollment appointment. Contractor staff will provide all participants with installation instructions.

PARTICIPANT ORIENTATION

1. **INTERVIEW**

During the initial enrollment interview the case manager will establish a case file (as described in paragraph 2 below); familiarize the participant with program requirements; set curfew schedule; explain, in detail, the participant's responsibility to provide documentation to verify <u>all</u> time away from the residence. The orientation component of the enrollment process must include the following steps:

- a. Photocopy participant's picture ID and telephone bill for case file. Utility bills, bank statements, and other forms of address verification are also acceptable forms in lieu of a telephone bill.
- b. Provide written rules and regulations to participant, explain each point, including grievance procedures and have participant initial each page, section and sign the last page acknowledging that he/she has read, understood and received a copy of the program guidelines and agrees to comply.
- c. Schedule next office meeting according to risk level assessed by PTS:
 - STANDARD MONITORING a minimum of once every two
 (2) weeks. "Standard" participants who fail to comply with
 curfew schedules or other program guidelines will be changed
 to weekly meetings until such time as compliance improves
 substantially.
 - ENHANCED MONITORING a minimum of once every week for the first thirty (30) days of program participation; thereafter, "enhanced" participants who comply with all program guidelines may be seen once every 2 weeks.

Note: Participants sentenced on excluded charges will always be monitored at the ENHANCED level.

SETTING UP CASE FILE

Case files are established to provide a documented record of all EM participant activity. Participant case files are official Probation Department records and the information contained therein is confidential. The information contained in a case file is occasionally called into evidence at a court hearing. Should this occur, case

managers are authorized to testify as to specific information contained in the file, but are precluded from releasing any documents or copies of any documents without authorization from the Probation Department.

Approved case file contents include, but are not limited to the following:

- EM Assessment Report
- Court Ordered Monitoring Conditions form (and other court orders)
- Phone Bill (or other forms of address verification)
- Photo ID
- Orientation Forms, or Participant Contract, must be dated and signed by participant and case manager.
- Employment Verification
- Chronological case notes
- Enrollment/Failure to Enroll Notice
- All other Notices
- Completion/Termination Notice
- Original Schedule
- First transmit or good hook-up message
- Daily electronic monitoring activity reports
- Schedule changes
- Activity verification/documentation

SETTING CURFEW SCHEDULE

Set curfew schedule based on the activities allowed by either the court or the Sheriff's Department and by information provided by the participant on the orientation forms. Under no circumstances shall a participant be allowed more time away from his or her residence than is needed to complete these authorized activities.

<u>Court Component</u> - The Court Ordered Monitoring Conditions form lists all activities ordered or allowed by the court. The Court must approve any requests for additional activities. If Contractor does not have a copy of the Court Ordered Monitoring Conditions during the initial interview, call the court immediately to request a faxed or a digital-copy. The participant is to be on a "closed schedule" status that does not allow any activities until the Contractor can verify the court's order regarding permitted activities.

<u>County Jail Component</u> - The Sheriff's Department generally allows a participant to work or seek employment. Activities generally allowed may include court appointments, doctor/dental appointments, DMV, Social Services, Probation/Parole meetings, Narcotics Anonymous (NA) or Alcoholics Anonymous (AA) meetings. Other activities allowed under certain circumstances are shopping/laundry (when participant lives alone) and haircuts (if necessary for employment). All other activities must be verbally approved by a Sheriff Deputy assigned to the EM Program.

Work Hours - Conventional Employment

It is expected that all participants will be allowed to work. Participant must provide his or her regular work schedule and the approximate travel time. Case manager will enter this information on the original scheduling form; which will be maintained in the participant's case file. Any changes to the permanent work schedule must be submitted at least twenty-four (24) hours in advance must be noted in the case file.

- Set permanent curfew schedule to include work hours and travel time each way.
- b. Change schedule, as needed, for prior approved overtime, scheduled medical appointments, etc., noting reason for change in participant's case file. Instruct participant to provide appropriate documentation of the activity at the next office meeting.
- c. Maintain copy of participant's original schedule and any subsequent schedule changes in case file.

Work Hours – Unconventional Employment

Many participants are either self-employed or do not report to a specific job every day. Setting curfew schedules for these cases must be handled on an individual basis according to the participant's type of employment. Participant is expected to provide case manager with a work schedule and be able to verify his/her activities.

Participant's work curfew schedule is not to be so broad as to allow the participant to come and go for 8, 10 or 12 hours throughout the day with no way to verify employment, unless the court is aware of the situation and authorizes participant to work with minimal or no verification. It is sometimes best, when a participant is self-employed, does "odd jobs" or "day work", to "close" his/her schedule (no time away from home) until he/she reports a specific day and time period which employment is scheduled.

Unemployed - Court authorizes job search

Participants will not generally be allowed an arbitrary eight (8) to twelve (12) hours, five (5) to seven (7) days per week to be away from their residence to seek employment.

- a. Set a maximum three (3) to four (4) hours per day, two (2) or three (3) days per week to allow participant to look for employment.
- b. Instruct participant, whenever possible, to provide a written list of prospective employers he/she plans to see, including the addresses. Participant is to provide a means of verifying his or her job search activities. Verification may include, but not be limited to, a duplicate job application, company business card signed by a representative of the company, or a signed attendance sheet. Discuss with the participant their responsibility to verify their whereabouts.

Note: If the participant is on formal probation, the DPO can authorize more time to search for employment.

Other court authorized activities

 Participant must provide the Contractor with a schedule, including the date, time and location of any court authorized activities, such as drug or alcohol treatment meetings, counseling sessions, medical appointments, etc. It is desirable, for scheduling purposes, to allow a participant to attend meetings on the same days and times each week.

Note: Because most NA or AA evening meetings are held after 7:00 p.m., it is usually a good practice to require the participant to go directly home from work and then set a separate schedule to allow for attendance at a scheduled meeting.

Example: Participant works from 8:00 a.m. to 5:00 p.m. and he is allowed 1/2 hour travel time each way; he has a scheduled AA meeting from 8:00 p.m. to 9:30 p.m. and is allowed 15 minutes travel time each way. The schedule should be set to allow participant to leave at 07:30 and enter at 17:30; leave again at 19:45 and enter at 21:15. In most cases, the participant should not be allowed to be away from his residence for almost 14 hours (07:30 until 21:15) in one increment.

2. Participant is to be given an attendance sheet to be signed by authorized personnel and submitted for inclusion into case file.

INSTALLATION AND HOOK-UP

- 1. GPS and strap are attached to participant's ankle, by an installer either at a branch office, Inmate Reception Center, or the participant's residence.
- 2. Record serial numbers of each piece of electronic monitoring equipment assigned to participant on computerized equipment log.
- 3. When the receiving unit is plugged in, if it is working properly, the computer will receive a "good hook-up" message. If this is done at a branch office, the participant is instructed on how and where to install the equipment in the home.

DAILY CONTACT LIST

Contractor is required to notify PTS of all enrollments, completions and terminations on a daily basis. Although the Daily Contact List was developed for this purpose, it is permissible for Contractor to utilize a computerized version of the form instead, provided it contains all the pertinent information.

A. Fax or email contact list to PTS daily at the end of the day, whether or not there was any activity.

Note: Contractor will retain the absconder's name on active status (i.e., daily count list) for a period of up to fourteen (14) calendar days. After fourteen (14) calendar days, the absconders will be removed from the daily count but will remain on the Contractor's data base indefinitely, until a final disposition is received.

- B. List must be legible and shall include all enrollments, completions and terminations.
- C. If an enrollment or completion occurred after the list was faxed or emailed, i.e., weekends, evenings, include the information on the next day's list.
- D. All follow-up documents (Enrollment, Completion, Termination Notices) must bear the same date shown on the list.
- E. Proofread documents for accuracy in case name, number of days ordered, dates.

SCHEDULE CHANGES

- A. Schedule changes shall not be approved for activities not authorized by the court. Schedule changes must be approved by authorized Contractor personnel in advance of the activity and a schedule change form must be maintained in the case file after appropriate case notes stating the reason for the change have been entered into the case file
- B. Enter schedule changes into the monitoring computer prior to the activity; in the case of "one time only" changes, the original schedule must be reinstated into the computer upon participant's return.

TYPE OF EQUIPMENT

All program participants will be monitored by a twenty-four (24) active/continuous system, utilizing a random contact back-up system. Monitoring equipment must be equipped with a tamper alarm system. The monitoring system must have a very high degree of reliability and dependability. It is the Contractors' responsibility to ensure that the monitoring equipment is working properly.

- A. The battery life of the transmitter and receiver must be sufficient to last throughout the term in between required charging periods. The participant is required to charge the device twice per day for one hour. If the device fails to hold a charge within this period, the device, charging unit, or both must be addressed by the contractor within one business day.
- B. Contractor's software system must be set to perform an automatic check of GPS device contact and tracking. System to report immediately if either contact or tracking from device has not been received in over 4 hours.

REPLACEMENT/REPAIR OF EQUIPMENT

If there appears to be any equipment problem; which compromises monitoring accuracy, Contractor(s) will provide for prompt repair or replacement of the equipment.

- A. All equipment malfunctions must be replaced or repaired within twenty-four (24) hours of discovery of the malfunction.
- B. Place documented telephone calls to the participant every two (2) hours until equipment is repaired or replaced.

TAMPER ALARMS

Contractors must provide immediate documented telephonic response to tamper alarms.

- A. After initial telephone contact, place documented telephone calls to the participant every two (2) hours until equipment is inspected or replaced by the Contractor.
- B. Replacement, if necessary, must occur within twenty-four (24) hours of initial tamper alarm.
- C. Replacement, repairs and resets must be documented in case file and in equipment logs.

REPORTING GUIDELINES

It is the Contractor's responsibility to track and report participant's adherence to program guidelines. Contractors will notify designated authorities of the probable violation of conditions within the time limits established by the Probation Department (below) using only Probation Department approved forms (Appendix E). When sending notification to the court, the Contractor is to complete one notice, as applicable, for each application number corresponding to court case number.

A. TYPES OF REPORTS

- 1. **ABSCOND NOTICE** Reports any time a participant is out during curfew and unable to be monitored for longer than four (4) hours; Abscond Notices shall be sent to Pretrial Services, and, as applicable, to the court, Sheriff's Department, and DPO within the following time frames:
 - a. <u>During court hours</u>: If a participant is away from his or her residence without authorization for four (4) hours during business hours, the Contractor will immediately fax or email an Abscond Notice to PTS, and, as applicable the court, Sheriff's Department, and DPO.
 - b. <u>After court hours/weekends</u>: If a participant is away from his or her residence without authorization for four (4) hours after business hours or on weekends, the monitoring center will alert the branch manager by pager and continue monitoring the participant's activity.
 - i. If the participant returns prior to the start of the next business day, the Contractor will attempt to obtain the participant's statement regarding the violation. By 11:00 a.m., fax or email a Non-Compliance Notice to PTS and, as applicable, the court, Sheriff's Department and DPO, detailing the incident.
 - ii. If the participant has not returned by the start of the next business day, the Contractor will telephonically notify PTS, the court or Sheriff's Department, if applicable, by 9:00 a.m. Follow up with an Abscond Notice faxed or emailed to PTS and, as applicable, the court, Sheriff's Department and DPO by 11:00 a.m. (if participant still has not returned).
- 2. **ALCOHOL VIOLATIONS** Mail, email or fax a Non-Compliance Notice within twenty-four (24) hours to PTS and, as applicable, the court, Sheriff's Department, and the DPO if, after three (3) tries, breath test results show any measurable amount of alcohol.

If a participant tests at a level of .08 or higher on a breath alcohol test, a Non-Compliance Notice shall be faxed or emailed to PTS and, as applicable, the court, Sheriff's Department and DPO, within the following time frames:

- a. <u>During court hours:</u> Call the court or the Sheriff's Department and fax or email immediately.
- b. <u>After court hours/weekends:</u> Fax or email by 9:00 a.m. next business day.
- 3. **COMPLETION NOTICE** Reports the last day participant is monitored after successfully completing the term specified by the court or Sheriff in the EM program.
- 4. **ENROLLMENT NOTICE** Reports the date the participant actually enrolls with the Contractor; which includes equipment installation and good hook-up message.
- 5. **FAILURE TO ENROLL NOTICE** Reports that participant has failed to enroll by the date ordered by the court, or after release from custody
- 6. **NON-COMPLIANCE REPORT** Reports any action by the participant; which is in violation of the court's order or program guidelines. Incidents of non-compliance shall be reported to PTS, the court and DPO, if applicable, within the time frames indicated below, applicable to both standard and enhanced cases:
 - a. <u>During court hours:</u> All incidents of non-compliance must be faxed or emailed to PTS, the court and DPO, if applicable, by 5:00 p.m. the next business day.
 - b. <u>After court hours/weekends:</u> Fax or email current status to PTS, the court and DPO, if applicable, by 5:00 p.m. the next business day.
- 7. **STATUS REPORT** Reports routine status of participant's performance on the program or is sent after any other report, if the status changes.
- 8. **TERMINATION NOTICE** Reports that participant has been removed from the EM program, for non-compliance, prior to expected completion date.

B. **DISTRIBUTION**

1.

ourt	Court
------	-------

- b. (1) Sheriff Department, if CJ participant
- c. (1) PTS

Non-Compliance Notice:

- d. (1) DPO, if on formal probation
- e. (1) Program participant
- f. (1) Case file

Note: Contractor is not required to send Non-Compliance Notices to Sheriff's Department unless the violation is a high breath alcohol test (BAT). PTS will track Non-Compliance Reports on County Jail participants, report potential problems and make recommendations to Sheriff's Department.

2. All other notices:

- a. (1) Court, if a court participant
- b. (1) Sheriff's Department, if CJ participant
- c. (1) PTS
- d. (1) DPO, if on formal probation
- e. (1) Program participant
- f. (1) Case file

MONITORING ACTIVITY REPORTS

Daily Exception Reports

Case managers will review the daily monitoring activity "exception report" first thing each business day. This will bring to their attention routine curfew violations or units that were late to test for the previous day. The daily report will include mention of any telephone contacts between the twenty-four (24) hour monitoring center staff and the person to whom they spoke at the participant's residence.

The monitoring center staff will notify Branch Managers, as needed, with immediate notification of serious violations, such as active tamper alarms, unresolved late-to-test, and knowledge of situations which may compromise public safety. Non-Compliance Reports and/or Abscond Notices for these incidents will be faxed or emailed by 11:00 a.m. the next business day to designated authorities. Non-Compliance Reports on less serious program violations, such as failure to provide documentation, curfew violations, unresolved or actual transmitter tampers can be emailed, mailed or faxed by the end of the business day.

All-activity Reports

Reports of all monitoring activity, including but not limited to, enters, leaves, monitor checkins, late-to-test messages for the period between office visits will be printed for the file prior to each office visit. The report will be reviewed with the participant and documentation or verification of activities will be collected during the office visit. Non-Compliance Reports will be sent on any activity which was unauthorized or undocumented.

NEW ARRESTS

In the event that the participant is arrested while on the court component of the EM Program, the court will determine whether the participant will remain on electronic monitoring. The Contractor shall follow the steps outlined below:

- A. Fax, email or mail a Non-Compliance Notice to the court, PTS and DPO, if applicable, that a new arrest has been verified.
- B. Fax, email or mail an Abscond Notice to PTS and as applicable, to the court, Sheriff's Department, and DPO, if the participant has not returned home within four (4) hours and the participant's arrest has not yet been verified.
- C. Fax, email or mail a Status Report once it has been verified that the participant is in custody or has returned home.

In the event that the participant is arrested while on the jail component of the EM Program, the Sheriff's Department will arrange for the participant to be returned to custody. A Termination Notice will be faxed, emailed or mailed to PTS, the Sheriff, and the DPO, if applicable.

PARTICIPANT VOLUNTARILY TERMINATES

- A. If a court participant voluntarily removes him or herself from the court component of the EM Program and turns in their monitoring equipment, an ABSCOND NOTICE shall be sent to the court. PTS will follow up on court action.
- B. If a participant appears in court to requests to be removed from the EM program and the court grants the request, for other than non-compliance, a COMPLETION NOTICE shall be sent.
- C. If a jail participant voluntarily returns to custody, not due to program violations, he or she will not be penalized but will remain in custody until his or her early release date. In this instance a COMPLETION NOTICE will be sent because this is not a program failure.
- D. If a CJ participant is remanded for non-compliance, it is considered a program failure; the inmate will remain in custody for the term of the sentence and is no longer eligible for any future release on EM. In this instance a TERMINATION NOTICE shall be sent.

DEFENDANT WALK-INS

If a defendant comes in with Court Ordered Monitoring Conditions from the court and Contractor has not received an Assessment Report, the Contractor will fax or email a copy of the Court Ordered Monitoring Conditions to PTS, obtain an EM application number, and instruct the defendant to call PTS from the office so that an application can be taken.

Note: Participants may not be enrolled without an EM application number. However, it is not necessary that the assessment be completed prior to enrollment. A number can be assigned once PTS has received Court Ordered Monitoring Conditions. Participant must be monitored at the Enhanced level until the Assessment Report is received and contractor is notified otherwise.

ENROLLMENTS AFTER INELIGIBLE RECOMMENDATION

Participants are required to have both a permanent residence and a telephone in order to participate in this program. Occasionally, a participant who does meet one or both of these requirements is referred by the court and is found ineligible for program participation. If the court subsequently orders the participant into the program, the contractor shall instruct the participant to call PTS during the enrollment interview. PTS will complete an application by telephone. The contractor will include the participant's address and telephone number on the ENROLLMENT NOTICE.

PARTICIPANT CHANGE OF ADDRESS

Court Component

A court sentenced participant may not move without prior approval of the Court. Telephonic approval is acceptable, provided the phone call is documented in the case file and includes the date, time, name of the person spoken to, the name of the judge authorizing the change, the new address, and telephone number, if applicable.

County Jail Component

Contractor has the permission of the Sheriff's Department to approve a change of address for CJ participants.

The participant's new address and phone number must be reported to PTS on a STATUS REPORT within twenty-four (24) hours. (Court or CJ component)

- A. Contractor will verify the new location within twenty-four (24) hours of the monitor re-connect at the new address.
- B. Case manager will send a STATUS REPORT to PTS and, as applicable, the court, the Sheriff's Department, and the DPO.
- C. Contractor will maintain a copy in the case file of participant's first monitoring activity report showing a successful "good hook-up" message.

DOCUMENTATION

It is the Contractors' responsibility to account for the participant's whereabouts twenty-four (24) hours a day. Make it clear to the participant at the time of enrollment that certain documentation is required to account for his/her work schedule and other court approved activity. Probation Department approved documentation must provide **independent** verification by a third party that the participant was where he/she claims to have been on any given date and time. The documentation provided must correspond to the participant's daily monitoring activity report showing "enters" and "leaves". Acceptable forms of documentation include but are not limited to:

A. Conventional employment

- 1. Time card, signed by supervisor, is preferred
- 2. Paycheck stubs showing hours and time period worked, including any vacation and/or sick time taken and/or overtime worked
- 3. Note or letter detailing hours and days worked during a given week, signed by supervisor

B. Unconventional employment (self-employed, independent contractor, sales, field work, etc.).

Enrollment Notices for these participants will include a statement detailing the type of documentation to be provided. If participant subsequently obtains employment, a Status Report will be sent to appropriate agencies.

When a participant does not have a regular "9-5" job, it is sometimes difficult to find a way to verify his/her time at work. During the initial interview, make it clear to the participant that it is his/her responsibility to prove where he/she is when he/she is not at home. Discuss with the participant different methods of verifying his/her away time. For example:

- 1. Provide participant with an activity log, which requires a name, address, time, date, signature and telephone number of someone who can confirm the participant's whereabouts.
- 2. Instruct participant to provide a daily or weekly work schedule, including appointments, with contact name, address and telephone numbers.

- 3. Instruct participant to provide invoices, which include date of a job, hours spent on a job, job location, contact person, amount charged and/or paid.
- 4. Instruct participant to provide daily itineraries, mileage claim forms or similar records used for billing purposes.
- 5. Have participant explain how he/she plans to verify his/her activities and work with him/her to find an acceptable method.

OFFICE MEETINGS

To ensure program compliance, the steps listed below are minimum requirements of a meeting with a participant:

- A. Check GPS strap
- B. Collect verification of work and other approved activities
- C. Review activity reports since last office meeting and discuss any irregularities with participant
- D. Set next appointment according to risk level:
 - 1. STANDARD a minimum of every two (2) weeks
 - 2. ENHANCED every week. A minimum of once every week for the first thirty (30) days of program participation. Thereafter, enhanced participants who comply with all program guidelines may be seen once every two (2) weeks.
- E. Enter all of the above in case notes

CASE NOTES

A correct entry in the case notes must include the <u>time</u> and <u>date</u> of the entry, as well as the first initial and last name, or the signature of the person making the entry.

- A. Report any contact with participant and any information pertinent to monitoring participant's activities
- B. Report all schedule changes, reason, date, etc.
- C. Report all equipment problems, including the disposition

QUALITY CONTROL

The Contractor shall have a written quality control plan to ensure that the requirements of the contract are met. The original plan and any future amendments are subject to County review and approval and shall include, but not limited to:

- A. A functional performance test and evaluation of the electronic monitoring equipment, with documented results, each time the equipment is issued to and returned by a program participant; the Contractor will provide a written plan describing how its systems will be tested and how performance standards will be met.
- B. An inspection system assuring ongoing delivery of services. The inspection system must specify the activities to be audited/inspected on either a scheduled or unscheduled basis, how often audits/inspections will be accomplished, the title of the individual(s) who will perform and record the audits/inspections and the methods for identifying and preventing deficiencies in the quality of the system. All audits/inspection results must be documented and available for review by the County during normal business hours.
- C. A computerized method of tracking equipment inventory, maintenance, battery life, and service records specific to each piece of equipment in accordance with suggested manufacturers maintenance specifications
- D. A method for ensuring uninterrupted service to Probation in the event of a strike of Contractor's employees
- E. A method for ensuring that offender record confidentiality is maintained

DAILY REPORTS

Contractor will provide a Master List of all EM cases using format provided by the County. The report may include, but not be limited to, the following:

A. MASTER LIST OF ALL ACTIVE EM CASES SORTED BY BRANCH OFFICE

Information to be reported on each case:

1.	EM #	8.	CASE MANAGER
2.	LAST NAME	9.	BRANCH OFFICE
3.	FIRST NAME	10.	ENROLLMENT DATE
4.	COURT	11.	FINAL DISPO DATE
5.	COURT CASE NUMBER	12.	# DAYS MONITORED
6.	CHARGE		
7.	RISK LEVEL		

DEFINITIONS

<u>Abscond</u> - A violation of curfew guidelines and inability to be electronically monitored for more than four (4) hours.

<u>Active Equipment</u> - Electronic monitoring device utilizing a one – piece electronic monitoring device with Global Positioning System (GPS).

<u>Assessment</u> - An evaluation of a participant referred to Pretrial Services to determine suitability for participation in the electronic monitoring program.

BAT - A breath alcohol-testing device approved by the Department of Transportation.

<u>Case Manager</u> - An employee of the Contractor whose responsibility is to provide the highest level of service to the courts and maintain a high regard for community safety. A case manager's duties include, but are not limited to, providing the participant with program rules and regulations, ongoing monitoring and documenting the participants compliance to court orders, and reporting incidents of non-compliance to the appropriate authorities.

<u>Completion</u> - Fulfillment of all requirements of the court as listed on the Court Order Monitoring Conditions, or a disposition as directed by the court or Sheriff's Department for other than program non-compliance

<u>Concurrent</u> - Applies to separate periods of confinement imposed for separate offenses which, by court order, may be served simultaneously.

<u>Consecutive</u> - Applies to separate penalties imposed for separate offenses which, by court order, must be served one after the other

<u>Contract Discrepancy Report (CDR)</u> - A report prepared by the Probation Department's Project Director to inform the Contractor of faulty service. The CDR requires a response from the contractor within ten (10) days, or as otherwise specified by the Project Director, explaining the problem and outlining the remedial action being taken to resolve the problem.

<u>Contractor</u> - Any vendor, service provider, or company contracting with the Probation Department to provide electronic monitoring services.

<u>Court Order</u> - Any notice in writing or orally presented by a judge specifying conditions of participation for EM.

<u>Curfew</u> - Hours during which a participant is required remain within the interior premises of the designated place of residence.

<u>Day(s)</u> - Calendar days, not business or working days.

Enhanced monitoring - Refers to monitoring guidelines established for program participants who are convicted of an excluded charge or those with a risk assessment determined by Pretrial Services to be at a level between 14 - 22 on the EM risk assessment scale.

Enrollment - The process of instructing a participant in the program guidelines, , obtaining signed participation agreement, attaching GPS to the participant and installing the equipment in the participant's home. Enrollment is not complete until a successful "transmit" message is received from the equipment in the participant's home by the contractor's monitoring center.

Excluded charge - Any charge found on the Court and Sheriff's Excluded Charge lists.

<u>Failure to enroll</u> - Participant did not enroll in EM by the date specified on the Court Order Monitoring Conditions form or other court order.

<u>Hook-up</u> - That part of the enrollment process where the GPS is fitted to the participant's ankle and activated.

Hours - Sequential sixty (60) minute time frames.

<u>Installation</u> - That part of the enrollment process where the monitoring equipment is installed in the participant's home, equipment is switched on and "good hook-up" message is received at the monitoring center.

<u>Installer</u> - An equipment technician employed by the contractor to install and service participants' monitoring equipment.

Low-Risk Offender - An individual whose criminal record, including the offense for which he has been referred to EM, reflects no history of violence, drug sales or child molestation.

<u>Monitoring Staff</u> - Contractor employees assigned to the Monitoring Center who track and report the transmissions of a participant's electronic monitoring equipment.

<u>Non-Compliance</u> - Any violation of program rules, guidelines or any conditions listed on the Court Ordered Monitoring Conditions form.

<u>Orientation</u> - That part of the enrollment process where the participant is instructed in the program guidelines, participant signs required documents to be accepted into the program, and arrangements are made for the installation of the monitoring equipment.

<u>Participant</u> - Any pre-sentenced or convicted offender who is participating in EM. In some instances, a witness is also placed as a participant upon program eligibility.

<u>Project Director</u> - Chief Probation Officer or his/her designee responsible for overall operations of EM.

<u>Project Manager</u> - Probation Department approved contractor's representative responsible for overall operations by the Contractor in compliance with the Contract and its procedures.

<u>Standard Monitoring</u> - Refers to monitoring guidelines established for program participants with a risk assessment determined by PTS at a level between 0 - 13 on the EM risk assessment scale.

<u>Termination</u> - Discontinuance of the participant as directed by the court or Sheriff for non-compliance.



PROBATION DEPARTMENT PRETRIAL SERVICES BUREAU ELECTRONIC MONITORING PROGRAM



Phone: (213) 893-5369 • Fax: (213) 633-4684 • Email: EMprogram@probation.lacounty.gov

Send completed form to the Electronic Monitoring Program

The charges listed on the Courts List of Excluded Charges are generally excluded from participation in the Electronic Monitoring Program. However, in the interest of justice, participants referred on excluded charges will be assessed for electronic monitoring if requested by the court.

(Rev. 05/2021)

TECHNICAL EXHIBIT 7

Appendix A-II Page 1 of 2

EM Application #:

COURT - ORIGINAL



PRETRIAL SERVICES BUREAU ELECTRONIC MONITORING PROGRAM



Court Ordered Monitoring Conditions

Defendant: Last Name First Name Attorney's Name:	Middle Initial	Defendant's Phone #: Offense: Attorney's Phone #:
	Concurrent/Consec	utive with Case #:
Judge:	ectronic monitoring in lie	
PERMIT	TED ACTIVITIES (home co	onfinement with restrictions)
Employment Prearranged Overtime at work Secondary Employment Laundry – prearranged Grocery Shopping – prearranged Religious Activities (Church/Temple) Community Service	AA or NA Mee	al Appointments etings Frequency: opointment:
OPTIO	ONAL CONDITIONS	
Defendant is ordered alcohol monitoring uni [] Breath Alcohol Remote Testing Drug Testing Frequency:	til	 cohol Monitoring
Additional Restrictions/Conditions: Enrollment requires two (2) business days to com Defendant will be required to contact the Electronic M		-
Judge/Clerk Signature:		Date:
Send completed forr Phone: (213) 893-5369 Fax: (213) 6	m to the Electronic Monit 633-4684 E-mail: EM	oring Program Program@probation.lacounty.gov Revised 05/25/2021

PRETRIAL SERVICES - COPY

DEFENDANT - COPY

TECHNICAL EXHIBIT 7

Appendix A-II Page 2 of 2

INSTRUCTIONS ON HOW TO ENROLL ONTO THE ELECTRONIC MONITORING PROGRAM

- Upon receiving the Court Ordered Monitoring Conditions form from the Judge, the defendant must fax or email the form
 to the Electronic Monitoring Program (page 1) and to CorrectiveSolutions at fax (800) 742-3254. The defendant must
 enroll within 48 hours of the order date or the date specified by the court.
- 2. The defendant must contact the Electronic Monitoring Program for an interview at (213) 893-5369.
- Defendants will then be referred to CorrectiveSolutions to schedule an appointment for orientation. At the appointment, defendant will be issued the equipment and will be assigned a case manager.

THE FOLLOWING DOCUMENTS MUST BE PROVIDED BY THE DEFENDANT AT THE ORIENTATION APPOINTMENT:

- · Court papers regarding this case, including the Court Ordered Monitoring Conditions form
- · Driver's license or picture ID
- · Address verification (i.e., recent utility bill, bank statement, or any other forms of address verification)
- · Proof of employment (i.e., recent paystub, letter of employment, etc.)
- · For self-employed, tax returns, income statements, account's receivable, and business licenses should be provided

ADDITIONALLY, THE FOLLOWING IS REQUIRED:

- 1. You must have electricity during the duration of the program.
- 2. If a landline device is required for monitoring equipment, the line cannot
 - · Have an answering machine
 - · Be a cordless phone
 - · Have any custom phone services such as call waiting, voicemail, call forwarding, caller ID, computer modems, etc.
- If the defendant does not have a residential telephone, it must be disclosed to CorrectiveSolutions at time of appointment scheduling.

INSTRUCCIONES PARA INSCRIBIRSE AL PROGRAMA DE MONITOREO ELECTRONICO

- Al recibir la forma de condiciones (Court Ordered Monitoring Conditions) de parte del Juez, debe mandar la forma al programa de monitoreo electronico (pagina 1) y CorrectiveSolutions al fax (800) 742-3254. Tendra 48 horas del dia de la orden o dia determinado por el Juez para matricula.
- 2. Debe llamar al programa de monitoreo electronico al (213) 893-5369 para una entrevista.
- Al terminar la entrevista, sera referido a CorrectiveSolutions para una cita de orientacion donde recibira el equipo electronico y sera dirigido a un trabajador de caso.

LOS SIGUIENTES DOCUMENTOS DEBERAN SER PRESENTADOS EL DIA SE SU CITA:

- Los papeles de la corte, incluyendo su Court Ordered Monitoring Conditions (papel color rosa)
- · Licensia de manejo o identificacion
- Verificacion de domicilio, por ejemplo factura de utilidades o otro documento confirmando el domicilio
- · Verificacion de empleo, por ejemplo talones de cheques recientes o carta de empleo
- Si empleado por si mismo, presente papeles de income tax, recibos de cuentas, o licencia de negocio.

ADICIONALMENTE, LO SIGUIENTE ES REQUERIDO:

- 1. Debe mantener electricidad en su casa durante el programa.
- 2. Si su equipo requiere linea telefonica donde su equipo electronico se conectara, no puede tener
 - Maquina de recibir mensajes
 - · Telefonos inalambrico
 - Servicios especiales como llamada de espera, doble linea, modems de computadoras
- 3. Si no tiene linea telefonica en su casa, tiene que informarle a CorrectiveSolutions cuando confirme su cita de inscripcion.

COURT – ORIGINAL PRETRIAL SERVICES – COPY DEFENDANT – COPY

County of Los Angeles Probation Department Pretrial Services Division, Electronic Monitoring Program Assessment Report

Case Filing								Applicati	on No.:		
Filed Name			Court/Sheriff Facility Court Date			t Date					
Arrest											
Booked name Arresting Agency			ng Agency		Booking Number			Arrest Date			
Case Number	r	Charge		Level	Description	1					
Defendant I	nforma	ation									
Sex	Race		Birthdate	Age	!				SS Numbe	er	
Address	•			•		Residence I	n:		Born		
						County:					
						State			Primary L	angua	ge
Telephone						USA:					
()											
Lives With			Relations	nip		ID Number		Туре	E:	xpirati	on Date
Marital Status		How Long	Supp	ort	Children 0						
Employment/Sup	pport Stat	us	How Long		Income 0	Education					
Employer						School					
Defendant J	Defendant Justice Identifies Codes/Criminal History										
			· ·					t l	D. d. d.	V NI	- L
Main Number		CII Number		FB	l Number		LAPDI	lumber	Probation	ı x ıvur	nber
Comments											
Assessment Stat	us									_	
Interviewed Poir	nts	Verified Points	Ris	sk Asses	sment	Completion :	Status				Initials
Signature:			Possi Date		olment		Possi	ible Completior	n Date:	_	

TECHNICAL EXHIBIT 7

County of Los Angeles Probation Department

ELECTRONIC MONITORING PROGRAM COURTS LIST OF EXCLUDED CHARGES

Page 1 of 4

The charges listed in this document are generally excluded from participation in the Electronic Monitoring Program. However, Judicial Officers may assign participants, with charges listed in the exclusionary list, to the program when the interest of justice may be served.

HS11154	Prescribing, administering or furnishing controlled substances
HS11155	Physician surrendering controlled substance
HS11351	Possession or purchase for sale of certain specific or classified
	controlled substances.
HS11351.5	Possession or purchase for sale of cocaine base.
HS11352 ALL	Illegal transportation, sale, furnishing of certain specific or classified
	controlled substances
HS11353	Person 18 years or over using minor in sale, transportation, or giving
	to a minor certain specific or classified controlled substances
HS11353.5	Adult preparing to sell to minor, drugs or alcohol
HS11354 A	Person under 18 years using minor in sale, transportation, or giving
	to a minor certain specific or classified substances
HS11359	Marijuana possession for sale
HS11360 A	Marijuana transportation, sale or furnishing
HS11361 ALL	Marijuana person 18 years or over using minor in sale,
	transportation, giving to minor
HS11366	Maintaining place for selling, giving, using of certain specific or
	classified controlled substances
HS11366.5 B	Allowing location to be fort for sale
HS11366.6	Using fort location for sale
HS11378	Possession of controlled substances for sale
HS11378.5	Possession or purchase for sale of PCP
HS11379 -	Transportation, sale of controlled substances and related offenses
HS11379.9 ALL	
HS11379.5 ALL	Transportation, sale of PCP
HS11379.6 ALL	Manufacture of PCP and other controlled substances
HS11380	Soliciting minor controlled substances
HS11383 A	Possession of precursors with intent to manufacture PCP
HS11390	Cultivation of mushrooms
HS11391	Transportation of mushrooms
PC128	Perjury resulting in capital punishment
PC139A	Threat to use force upon witness or their family
PC140	Threatening witness, victim or informant
PC148 ALL	Resist, delay or obstruct public officer
PC181	Slavery
PC186.26	Criminal Street Gang
PC187 A ALL	Murder

TECHNICAL EXHIBIT 7

County of Los Angeles Probation DepartmentELECTRONIC MONITORING PROGRAM

COURTS LIST OF EXCLUDED CHARGES

Page 2 of 4

PC189	First or second degree murder			
PC192	Voluntary manslaughter			
PC192 B	Involuntary manslaughter			
PC203	Mayhem			
PC205	Aggravated mayhem			
PC207 ALL	Kidnapping			
PC209 ALL	Kidnapping of ransom or extortion			
PC210.5	False imprisonment – kidnap to evade arrest			
PC211 ALL	Robbery			
PC212.5 A	Robbery in an inhabited dwelling, vessel, or building if person			
	charged with using a deadly or dangerous weapon PC12022			
PC213.5	Robbery, inhabited house or trailer			
PC215 ALL	Carjacking			
PC217.1	Attempt to kill specified public officials			
PC218	Train wrecking			
PC219	Train derailing			
PC220	Assault with intent to commit mayhem, rape, sodomy, oral copulation			
	or any violation of PC264.1, 288 or 289			
PC236	False imprisonment			
PC240 ALL	Assault			
PC241.1	Assault			
PC242	Battery			
PC243 ALL	Battery			
PC244 ALL	Battery with chemicals			
PC245 ALL	Assault with deadly weapon or force likely to produce a great bodily injury			
PC246 ALL	Shooting at inhabited dwelling, building, occupied motor vehicle			
PC261 ALL	Rape			
PC262 ALL	Spousal rape			
PC264	Rape in concert with force and violence			
PC266 ALL	Enticement of female under 18 for prostitution			
PC267	Abduction of person under 18 for prostitution			
PC271	Desertion of child under 14			
PC272	Contributing to delinquency of person under 18			
PC273 A ALL	Willful cruelty of a child			
PC273 D ALL	Corporal punishment of a child			
PC273.5 A	Corporal injury of spouse, cohabitant of opposite sex, or mother or father of his/her child			
PC285	Incest			
PC286 ALL	Sodomy			

County of Los Angeles Probation DepartmentELECTRONIC MONITORING PROGRAM

COURTS LIST OF EXCLUDED CHARGES

Page 3 of 4

PC288	Lewd or lascivious involving children
PC 288 (a) ALL	Lewd or lascivious acts upon a child under 14
PC288 A (a)	Oral copulation
ALL	'
PC288.5	Three or more acts of substantial sexual conduct with child under 14
PC289 ALL	Penetration of foreign object
PC289.5	Rape and sodomy
PC368 ALL	GBI and mental suffering of dependent adult
PC404.6	Incitement to riot
PC405 B	Lynching
PC417 ALL	Exhibit deadly weapon
PC417.1	Exhibit firearm in presence of officer
PC417.25 ALL	Exhibit laser scope at a person threatening manner
PC417.3	Exhibit firearm in presence of vehicle occupant
PC417.4	Exhibit imitation firearm in threatening manner
PC417.6 ALL	Intentionally inflicting GBI while in the commission of PC 417,
	PC417.2 or PC 417.8
PC417.8	Exhibit a firearm with the intent to resist arrest
PC422	Terrorist Threats
PC451 ALL	Arson
PC452 ALL	Unlawfully causing fire
PC453 ALL	Possession of explosives or flammable matter
PC454 ALL	Arson
PC455	Attempted Arson
PC464	Burglary with explosives
PC487 C	Grand Theft Person
PC519 ALL	Extortion
PC646.9 ALL	Stalking
PC647.6	Annoying/molest child with prior felony conviction
PC653 ALL	Solicit commission of certain offenses
PC664	Any attempt of any violation on the EMO 1 List
PC667.10	Sex crimes (repeat offenders) against someone under 14
PC667.6 A	Sex conviction with prior sex conviction
PC667.6 B	Sex conviction with two or more prior sex convictions
PC667.7	GBI with prior violent felony conviction
PC667.8 A	Kidnapping to commit sex offense
PC667.8 B	Kidnapping to commit sex offense, victim under 14
PC1203.09	Violent felonies against aged
PC12021.5	Street gang crimes with firearms

Appendix C

County of Los Angeles Probation Department ELECTRONIC MONITORING PROGRAM

COURTS LIST OF EXCLUDED CHARGES

Page 4 of 4

PC12022 ALL	Use of firearm in commission of a felony
PC12022.4 ALL	Furnish a firearm to another of raiding and abetting another to
	commit a felony
PC12022.55	Intentional infliction of GBI or death by discharging firearm from
	vehicle
PC12022.5 ALL	Use of firearm in commission of a felony
PC12022.7	Infliction of GBI in commission of felony
PC12022.75	Administer controlled substance against victim's will
PC12022.8	Inflict GBI in commission of certain sex offenses
PC12022.85	Sexual offense with knowledge that he/she has AIDS or HIV
PC12022.9 ALL	GBI causing termination of pregnancy
PC12034 C	Personally shooting from a motor vehicle.
PC12303.1	Carrying or placing explosive on common carrier
PC12303.2	Possession of destructive devices/explosives in public places.
PC12303.3	Possession of exploding or destructive device or explosive with
	intent to injure
PC12308	Explosion of destructive device with intent to commit murder
PC12309	Unlawful explosion causing bodily injury
PC12310	Explosive device which causes mayhem.
PC12312	Possession of materials with intent to make destructive device or
	explosive.
PC12355 ALL	Booby-trap causing GBI
VC2800.1	Evading a Peace Officer
VC 2800.2	Evading a Peace Officer Reckless Driving
•	

COMMUNITY BASED ALTERNATIVES TO CUSTODY (CBAC) LIST OF EXCLUDED CHARGES FOR CBAC PROGRAMS - 2017 **** CONVICTIONS ONLY****

Page 1 of 3

Asia (Alapa) Resisting a Peace Officer resulting in Death / serious Injury to Peace Officer never 166 PC Criminal contempt with gang injunction/domestic violence never 166 PC Criminal contempt with gang injunction/domestic violence never 168.0.22[A)PC Crime of active gang member never 168.0.22[A)PC Gross vehicular manslaughter never 169.0.00 PC Murder never 169.0.00 PC Murder never 169.0.00 PC Murder never 169.0.00 PC Manslaughter never 169.0.00 PC Mayham never 169.0.00 PC Kidnapping with rape, robbery, ransom, or extortion never 169.0.00 PC Kidnapping with rape, robbery, ransom, or extortion never 169.0.00 PC Kidnapping with rape, robbery, ransom, or extortion never 169.0.00 PC Kidnapping with rape, robbery, ransom, or extortion never 169.0.00 PC Kidnapping with rape, robbery, ransom, or extortion never 169.0.00 PC Kidnapping with rape, robbery, ransom, or extortion never 169.0.00 PC Kidnapping with rape, robbery 15.0.00 PC FC Kidnapping PC FC K			Post
Assault with intent to commit rape 136PC Assault with intent to commit rape 136PC Carjacking 136PC Resisting a Peace Officer resulting in Death / serious Injury to Peace Officer 136PC Cirlimol Contempt with gang injunction/domestic violence 136PC Cirlimol Cartive gang member 136PC Cirlimol Cartive gang member 136PC Cirlimol Cartive gang member 136PC Murder 139PC Murder 139PC Murder 139PC Manslaughter 139PC Manslaughter 139PC Manslaughter 139PC Manslaughter 139PC Manslaughter 139PC Kidnapping 139PC Kidnapping 139PC Kidnapping 139PC Kidnapping 139PC Kidnapping with rape, robbery, ransom, or extortion 139PC Kidnapping with rape, robbery, ransom, or extortion 139PC Kidnapping 139PC Kidnapping 139PC Kidnapping with rape, robbery, ransom, or extortion 139PC Kidnapping 139PC Kidnapping 139PC Kidnapping with rape, robbery, ransom, or extortion 139PC Kidnapping with rape, robbery for exterior for e	CHARGE	DESCRIPTION	Incarceration
33 33 33 33 34 35 34 35 34 35 34 35 36 36 36 36 36 36 36			Consideration
483.10(A)PC Resisting a Peace Officer resulting in Death / serious injury to Peace Officer never 166.PC Criminal contempt with gang injunction/domestic violence never 186.22(A)PC Crime of active gang member never 186.22(B)(I)PC Enhancement crime to benefit street gang never 18.7PC Murder never 91.5PC Gross vehicular manslaughter never 91.5PC Manslaughter never 03PC Mayhem never 03PC Mayhem never 03PC Aggravated mayhem never 03PC Kidnapping with rape, robbery, ransom, or extortion never 03PC Kidnapping during a carjacking never 10PC Robbery 5 11PC Robbery (home invasion) never 12PC Robbery (home invasion) never 13PC Assault with intent to commit rape never 12PC Assault with intent to commit rape never 13PC Battery on Police Officer 3 43I(DPC	136PC	Preventing /dissuade a witness from testimony	
166 PC			
186.22(B)C Crime of active gang member never 87PC Murder never 87PC Murder never 93.5PC Gross vehicular manslaughter never 93PC Manslaughter never 03PC Mayhem never 03PC Mayhem never 03PC Algaravated mayhem never 03PC Kidnapping with rape, robbery, ransom, or extortion never 03PC Kidnapping with rape, robbery, ransom, or extortion never 10PC Kidnapping with rape, robbery, ransom, or extortion never 10PC Kidnapping with rape, robbery, ransom, or extortion never 10PSPC Kidnapping with rape, robbery, ransom, or extortion never 10PSPC Kidnapping with rape, robbery, ransom, or extortion never 10PSPC Kidnapping with rape, robbery, ransom, or extortion never 10PSPC Kidnapping with rape, robbery, ransom, or extortion never 13PSPC Caljacking never 13PSPC Caljacking never 13PSPC Assault with intent to commit rape never <td></td> <td>· · · · · · · · · · · · · · · · · · ·</td> <td></td>		· · · · · · · · · · · · · · · · · · ·	
RED_ZED_TIPECT Charactement crime to benefit street gang never			
187PC Murder never 1915.PC Gross welicular manslaughter never 192PC Manslaughter never 192PC Manslaughter never 192PC Mayhem never 193PC Mayhem never 197PC Kidnapping never 197PC Kidnapping with rape, robbery, ransom, or extortion never 197PC Kidnapping with rape, robbery, ransom, or extortion never 197PC Kidnapping with rape, robbery, ransom, or extortion never 197PC Kidnapping with rape, robbery 5			
191.5PC Gross vehicular manslaughter never 192PC Manslaughter never 192PC Manslaughter never 192PC Manslaughter never 192PC Maybem never 193PC Aggravated mayhem never 193PC Kidnapping mever 193PC Kidnapping during a carjacking never 193PC Kidnapping during a carjacking never 193PC Kidnapping during a carjacking never 112PC Robbery (193PC Robbery 112PC Robbery 112P			
Manslaughter never 103PC Manslaughter never 103PC Mayhem never 103PC Aggravated mayhem never 103PC Kidnapping with rape, robbery, ransom, or extortion never 103PC Kidnapping with rape, robbery, ransom, or extortion never 103PC Kidnapping with rape, robbery 15PC Robbery 15PC Robbery 15PC Robbery 15PC Robbery 15PC Robbery 16PC Rob			
Mayhem never DSPC Aggravated mayhem never DSPC Aggravated mayhem never DOPC Kidnapping with rape, robbery, ransom, or extortion never DS-SPC Kidnapping with rape, robbery, ransom, or extortion never 11PC Robbery 5 11PC Robbery (more invasion) never 11SPC Carjacking 10 20PC Assault with intent to commit rape never 36PC False imprisonment 5 343(D)PC Battery on Police Officer 3 434(DPC Battery on Peace Officer/femergency personnel never 43.4 PC Sexual battery 5 45(A)(2)PC Assault with ferearm never 445(B)PC Assault with semi-auto firearm never 45(B)PC Assault with semi-auto firearm never 46.9PC Shoot at occupied dwelling never 46.9PC Shoot at occupied dwelling never 46.9PC Shoot at occupied dwelling never <t< td=""><td></td><td></td><td></td></t<>			
DOSPC			
Mighapping Never		·	
Mighapping with rape, robbery, ransom, or extortion			
11PC Robbery S Sobery Robbery S Sobery Robbery S Sobery Robbery Robbery S Sobery Robbery Robbery Sobery Sobery Robbery			
S 11PC Robbery Rob			
112PC Robbery (home invasion)			
15PC Carjacking 10 10 10 10 10 10 10 1		,	
Description		··	
False imprisonment 5 3			
A3(B)PC Battery on Police Officer 3 143(C)PC Battery on Peace Officer/emergency personnel never 143(A)PC Sexual battery 5 5 145(A)(2)PC Assault with firearm never 145(B)PC Assault with semi-auto firearm never 145(B)PC Assault on Peace Officer firearm never 145(C)PC Assault on Peace Officer firearm never 145(C)PC Assault on Peace Officer firearm never 145(B)PC Assault on Peace Officer firearm never 145(B)PC Shoot at occupied dwelling never 146(B)PC Discharge firearm / cause Injury - Death never 146(B)PC Rape never 146(B)PC Rape never 146(B)PC Rape never 146(B)PC Rape never 146(B)PC Willful cruelty to child never 146(B)PC Willful cruelty to child never 146(B)PC Sodomy never 146(B)PC Sodomy never 146(B)PC Forcible penetration by foreign object never 146(B)PC Registration of sex offenders never 147(B)PC Registration of sex offenders 147(B)PC Registration 147(B)PC Registration 147(B)PC Registration 147(B)PC Registration 147(B)PC Registration 1			
1434C)PC Battery on Peace Officer/emergency personnel never 1434APC Sexual battery 5 1434S(A)(2)PC Assault with firearm never 144S(A)(2)PC Assault with semi-auto firearm never 144S(B)PC Assault with semi-auto firearm never 144S(PC)PC Assault on Peace Officer firearm never 144S(PC)PC Assault on Peace Officer firearm never 146PC Shoot at occupied dwelling never 146PC Shoot at occupied dwelling never 146PC Rape never 146PC Rape never 146PPC Rape never 146PPC Williful cruelty to child never 1473(A)PC Williful cruelty to child never 1473(A)PC Williful cruelty to child never 148PPC Sodomy never 148PPC Sodomy never 148PPC Forcible penetration by foreign object never 148PPC Forcible penetration by foreign object never 149PPC Registration of sex offenders never 141PPC Lewd or obscene conduct / indecent exposure 10 148PPC Lewd or obscene conduct / indecent exposure 10 149PPC Riot (County Jail only) 5 141PPC Firearms 3 141PPC Firearms 3 141PPC Firearms 3 142PPC Criminal threats 10 150-1PPC Battery by prisoner never 151PPC Arson never 151PPC Arson never 151PPC Arson never 151PPC Arson never 151PPC Grand theft of a firearm never 151PPC Arson never 151PPC Grand theft of a firearm never 151PPC Arson never 151PPC Arson never 151PPC Grand theft of a firearm never 151PPC Arson never 151PPC Arson never 151PPC Arson never 151PPC Grand theft of a firearm never 151PPC Arson never 151PPC Arson never 151PPC Arson never 151PPC Arson never 151PPC Grand theft of a firearm never 151PPC Arson never 152PPC Grand theft of a firearm never 152PPC Grand theft of a firearm never 152PPC Grand theft of a firearm never 152PPC Stalking never never 152PPC Stalking never ne			
243.4PC Sexual battery 5 245.[A](2)PC Assault with firearm never 145.[B]PC Assault with semi-auto firearm never 145.[B]PC Assault with semi-auto firearm never 145.[C]PC Assault with semi-auto firearm never 145.[C]PC Assault on Peace Officer firearm never 146.BPC Shoot at occupied dwelling never 146.BPC Discharge firearm /cause Injury - Death never 146.BPC Rape never 146.BPC Rape never 146.BPC Nod at occupied dwelling never 146.BPC Nod at occupied dwelling never 146.BPC Nod Aggravated sexual assault on a child never 146.BPC Wilful cruelty to child never 146.BPC Wilful cruelty to child never 146.BPC Sodomy never 146.BPC Sodomy never 146.BPC Sodomy never 146.BPC Nod	· '	·	
Assault with firearm never	\ /		
424S(B)PC Assault with semi-auto firearm never 425(C)PC Assault on Peace Officer firearm never 426(DPC Shoot at occupied dwelling never 426(DPC Shoot at occupied dwelling never 61(DPC Rape never 61(DPC Rape never 62(DPC) Rape never 42(DPC) Willful cruelty to child never 48(DPC) Sodomy never 48(DPC) Registration of sex offenders never 49(DPC) Registration of sex offenders never 41(DPC) Lewd or obscene conduct / indecent exposure 10 44(DPC) Lewd or obscene conduct		·	
Assault on Peace Officer firearm Assault on a child Assau	, ,, ,		
ASPC Discharge firearm /cause Injury - Death never	` '		
246.3PCDischarge firearm /cause Injury - Deathnever161PCRapenever169 PC (All subs)Aggravated sexual assault on a childnever173(A)PCWillful cruelty to childnever186PCSodomynever188PCSodomynever189PCForcible penetration by foreign objectnever190PCRegistration of sex offendersnever190PCRegistration of sex offendersnever191PCChild pornographynever191PCLewd or obscene conduct / indecent exposure10105PCRiot (County Jail only)5197PCFirearms3192PCCriminal threats10192PCCriminal threats10193PCArsonnever194PCBattery by prisonernever195PCArsonnever195PCArsonnever194PCGrand theft of a firearmnever195PCArsonnever197CGrand theft of a firearmnever198PCGun at school10194PCStalkingnever194PCStalkingnever194PCAnnoying or molesting a child under 18never198PCAnnoying arrest5198PCEvading arrest5 <td>` '</td> <td></td> <td></td>	` '		
261 PCRapenever269 PC (All subs)Aggravated sexual assault on a childnever273 (A)PCWillful cruelty to childnever286 PCSodomynever288 PC (All subs)Lewd or lascivious acts with a child under 14never289 PCForcible penetration by foreign objectnever290 PCRegistration of sex offendersnever311 PCChild pornographynever314 PCLewd or obscene conduct / indecent exposure10368 PCCruelty to an adult dependentnever369 PCRiot (County Jail only)537 PCFirearms3312 PCCriminal threats10350.1 PCBattery by prisonernever387 (d)(2) PCGrand theft of a firearmnever387 (d)(2) PCGrand theft of a firearmnever347.6 PCAnnoying or molesting a child under 18never370 PC / 2684 PCMentally incompetent/Criminally insanenever3800.2 (A)VCEvading arrest53800.2 (A)VCEvading Peace Officer, disregard safety10300.08 PCParole violationnever3056 PCParole violationnever3056 PCParole violationnever3056 PCFlash incarceration (current arrest)never3057 PCEscapenever			
269 PC (All subs)Aggravated sexual assault on a childnever273 (A)PCWillful cruelty to childnever286 PCSodomynever287 (All subs)Lewd or lascivious acts with a child under 14never289 PCForcible penetration by foreign objectnever289 PCRegistration of sex offendersnever290 PCRegistration of sex offendersnever211 PCChild pornographynever214 PCLewd or obscene conduct / indecent exposure10268 PCCruelty to an adult dependentnever205 PCRiot (County Jail only)5217 PCFirearms3122 PCCriminal threats10250.1 PCBattery by prisonernever157 PCArsonnever157 PCArsonnever157 PCArsonnever157 PCGrand theft of a firearmnever157 PCGrand theft of a firearmnever266.9 PCGun at school10246.9 PCStalkingnever247.6 PCAnnoying or molesting a child under 18never2800.1 (A) VCEvading arrest52800.2 (A) VCEvading Peace Officer, disregard safety102800.2 (A) VCEvading Peace Officer, disregard safety10200.0 AB PCParole warrant - revoke parole supervisionnever2056 PCParole violation32454 (B)/3455 PCParole warrant - revoke parole supervisionnever25		9 , , , ,	never
273 (A)PCWillful cruelty to childnever186PCSodomynever188PC (All subs)Lewd or lascivious acts with a child under 14never188PC (All subs)Lewd or lascivious acts with a child under 14never189PCForcible penetration by foreign objectnever190PCRegistration of sex offendersnever311PCChild pornographynever314PCLewd or obscene conduct / indecent exposure10105PCRiot (County Jail only)5117PCFirearms3122PCCriminal threats10150.1PCBattery by prisonernever151PCArsonnever152PCArsonnever151PCArsonnever152C.9PCGun at school10154C.9PCStalkingnever154C.9PCStalkingnever137OPC / 2684 PCMentally incompetent/Criminally insanenever137OPC / 2684 PCMentally incompetent/Criminally insanenever1800.1(A)VCEvading arrest51800.2(A)VCEvading reace Officer, disregard safety10100.00.8 PCParole warrant - revoke parole supervisionnever105454 (B)/3455 PCFlash incarceration (current arrest)never105452(C)PCWork Release failure51532PCEscapenever		· · · · · · · · · · · · · · · · · · ·	never
286PCSodomynever288PC (All subs)Lewd or lascivious acts with a child under 14never289PCForcible penetration by foreign objectnever290PCRegistration of sex offendersnever291PCChild pornographynever291PCLewd or obscene conduct / indecent exposure102068PCCruelty to an adult dependentnever205PCRiot (County Jail only)517PCFirearms3122PCCriminal threats10150.1PCBattery by prisonernever151PCArsonnever152.6.9PCGrand theft of a firearmnever262.6.9PCGun at school10347.6PCAnnoying or molesting a child under 18never347.6PCAnnoying or molesting a child under 18never2800.1(A)VCEvading arrest52800.1(A)VCEvading resce Officer, disregard safety10300.00.08 PCParole warrant - revoke parole supervisionnever3056PCParole violation34054(B)/3455 PCFlash incarceration (current arrest)never40532PCEscapenever		•	never
Lewd or lascivious acts with a child under 14 189PC Forcible penetration by foreign object never 189PC Registration of sex offenders never 181PC Child pornography never 181PC Lewd or obscene conduct / indecent exposure 10 168PC Cruelty to an adult dependent never 165PC Riot (County Jail only) 5 17PC Firearms 3 1822PC Criminal threats 10 1850.1PC Battery by prisoner never 1851PC Arson never 187 (d)(2) PC Grand theft of a firearm never 187 (d)(2) PC Grand theft of a firearm never 187 (d)(2) PC Gun at school 10 1846.9PC Stalking never 187 (d)(2) PC Wentally incompetent/Criminally insane never 188 (d)(2) PC Evading arrest 5 188 (d)(2) PC Evading Peace Officer, disregard safety 10 10 (d)(2) PC Evading Peace Officer, disregard safety 10 10 (d)(3) PC Evading Peace Officer, disregard safety 10 10 (d)(4) PC Evading Peace Officer, disregard safety 10 10 (d)(4) PC Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE Parole warrant - revoke parole supervision never 10 10 (d)(4) PATOLE PAROLE PAR		·	never
289PCForcible penetration by foreign objectnever190PCRegistration of sex offendersnever111PCChild pornographynever114PCLewd or obscene conduct / indecent exposure10168PCCruelty to an adult dependentnever105PCRiot (County Jail only)5117PCFirearms3122PCCriminal threats10150.1PCBattery by prisonernever157PCArsonnever187 (d)(2) PCGrand theft of a firearmnever187 (d)(2) PCGrand theft of a firearmnever147.6PCAnnoying or molesting a child under 18never147.6PCAnnoying or molesting a child under 18never1800.1(A)VCEvading arrest52800.2(A)VCEvading Peace Officer, disregard safety10100.00.8 PCParole warrant - revoke parole supervisionnever105454 (B)/3455 PCFlash incarceration (current arrest)never105452PCWork Release failure510532PCEscapenever			never
290PCRegistration of sex offendersnever311PCChild pornographynever314PCLewd or obscene conduct / indecent exposure10368PCCruelty to an adult dependentnever405PCRiot (County Jail only)5117PCFirearms3122PCCriminal threats10150.1PCBattery by prisonernever151PCArsonnever187 (d)(2) PCGrand theft of a firearmnever364.9PCStalkingnever347.6PCAnnoying or molesting a child under 18never330PC / 2684 PCMentally incompetent/Criminally insanenever2800.1(A)VCEvading arrest52800.2(A)VCEvading Peace Officer, disregard safety103056PCParole warrant - revoke parole supervisionnever3056PCParole violation34054 (B)/3455 PCFlash incarceration (current arrest)never4052.2(C)PCWork Release failure540532PCEscapenever			never
STIPC Child pornography never	289PC	Forcible penetration by foreign object	never
114PC Lewd or obscene conduct / indecent exposure 10 168PC Cruelty to an adult dependent never 105PC Riot (County Jail only) 5 117PC Firearms 3 122PC Criminal threats 10 150.1PC Battery by prisoner never 187 (d)(2) PC Grand theft of a firearm never 187 (d)(2) PC Grand theft of a firearm never 187 (d)(2) PC Grand theft of a firearm never 187 (d)(2) PC Grand theft of a firearm never 197 (d)(2) PC Gun at school 10 1646.9PC Stalking never 197 (d)(2) PC Gun at school 10 1646.9PC Mentally incompetent/Criminally insane never 1870PC / 2684 PC Mentally incompetent/Criminally insane never 18800.1(A)VC Evading arrest 5 1800.2(A)VC Evading Peace Officer, disregard safety 10 1000.08 PC Parole warrant - revoke parole supervision never 197 (d)(2) PC Parole violation 3 18454 (B)/3455 PC Flash incarceration (current arrest) never 197 (d)(2) PC Work Release failure 198 (d)(2) PC Parole violation 198 (d)(290PC	Registration of sex offenders	never
668PCCruelty to an adult dependentnever105PCRiot (County Jail only)5117PCFirearms3122PCCriminal threats10150.1PCBattery by prisonernever187 (d)(2) PCArsonnever187 (d)(2) PCGrand theft of a firearmnever187 (d)(2) PCGun at school10146.9PCStalkingnever147.6PCAnnoying or molesting a child under 18never1370PC / 2684 PCMentally incompetent/Criminally insanenever1800.1(A)VCEvading arrest51800.2(A)VCEvading Peace Officer, disregard safety101000.08 PCParole warrant - revoke parole supervisionnever1056PCParole violation31454 (B)/3455 PCFlash incarceration (current arrest)never1024.2(C)PCWork Release failure51532PCEscapenever	311PC	Child pornography	never
Signature Sign	314PC	Lewd or obscene conduct / indecent exposure	10
117PC Firearms 3 122PC Criminal threats 10 150.1PC Battery by prisoner never 151PC Arson never 187 (d)(2) PC Grand theft of a firearm never 187 (d)(2) PC Grand theft of a firearm 10 146.9PC Stalking never 147.6PC Annoying or molesting a child under 18 never 1370PC / 2684 PC Mentally incompetent/Criminally insane never 1800.1(A)VC Evading arrest 5 1800.2(A)VC Evading Peace Officer, disregard safety 10 1000.08 PC Parole warrant - revoke parole supervision never 10454 (B)/3455 PC Flash incarceration (current arrest) never 1024.2(C)PC Work Release failure 5 1532PC Escape never	868PC	Cruelty to an adult dependent	never
122PCCriminal threats10150.1PCBattery by prisonernever151PCArsonnever187 (d)(2) PCGrand theft of a firearmnever186.9PCGun at school10147.6PCAnnoying or molesting a child under 18never1370PC / 2684 PCMentally incompetent/Criminally insanenever1800.1(A)VCEvading arrest51800.2(A)VCEvading Peace Officer, disregard safety101000.08 PCParole warrant - revoke parole supervisionnever1056PCParole violation3105454 (B)/3455 PCFlash incarceration (current arrest)never1052.2PCWork Release failure51532PCEscapenever	105PC	Riot (County Jail only)	5
So.1PC Battery by prisoner never So.1PC Arson never So.1PC Arson never So.1PC Grand theft of a firearm never So.1PC Grand theft of a firearm never So.1PC Gun at school 10 So.1PC Stalking never So.1PC Annoying or molesting a child under 18 never So.1PC Annoying or molesting a child under 18 never So.1PC Z684 PC Mentally incompetent/Criminally insane never So.1PC Evading arrest 5 So.0.2(A)VC Evading arrest 5 So.0.2(A)VC Evading Peace Officer, disregard safety 10 So.0.0.8 PC Parole warrant - revoke parole supervision never So.1PC Parole violation 3 So.1PC So.1PC Flash incarceration (current arrest) never So.2PC Escape never So.2PC Escape never So.2PC So.2PC So.2PC So.2PC Never Never So.2PC Never Never	17PC	Firearms	3
STPC		Criminal threats	10
187 (d)(2) PCGrand theft of a firearmnever526.9PCGun at school10546.9PCStalkingnever547.6PCAnnoying or molesting a child under 18never1370PC / 2684 PCMentally incompetent/Criminally insanenever2800.1(A)VCEvading arrest52800.2(A)VCEvading Peace Officer, disregard safety103000.08 PCParole warrant - revoke parole supervisionnever3056PCParole violation33454 (B)/3455 PCFlash incarceration (current arrest)never4024.2(C)PCWork Release failure54532PCEscapenever	150.1PC	Battery by prisoner	never
526.9PCGun at school10646.9PCStalkingnever647.6PCAnnoying or molesting a child under 18never3370PC / 2684 PCMentally incompetent/Criminally insanenever2800.1(A)VCEvading arrest52800.2(A)VCEvading Peace Officer, disregard safety10000.08 PCParole warrant - revoke parole supervisionnever0056PCParole violation33454 (B)/3455 PCFlash incarceration (current arrest)never024.2(C)PCWork Release failure54532PCEscapenever		Arson	never
346.9PCStalkingnever347.6PCAnnoying or molesting a child under 18never.370PC / 2684 PCMentally incompetent/Criminally insanenever.800.1(A)VCEvading arrest5.800.2(A)VCEvading Peace Officer, disregard safety10.000.08 PCParole warrant - revoke parole supervisionnever.0556PCParole violation3.454 (B)/3455 PCFlash incarceration (current arrest)never.024.2(C)PCWork Release failure5.532PCEscapenever	187 (d)(2) PC	Grand theft of a firearm	never
647.6PC Annoying or molesting a child under 18 never .370PC / 2684 PC Mentally incompetent/Criminally insane never .800.1(A)VC Evading arrest 5 .800.2(A)VC Evading Peace Officer, disregard safety 10 .000.08 PC Parole warrant - revoke parole supervision never .056PC Parole violation 3 .454 (B)/3455 PC Flash incarceration (current arrest) never .002.4.2(C)PC Work Release failure 5 .532PC Escape never		Gun at school	10
370PC / 2684 PCMentally incompetent/Criminally insanenever2800.1(A)VCEvading arrest52800.2(A)VCEvading Peace Officer, disregard safety101000.08 PCParole warrant - revoke parole supervisionnever1056PCParole violation34554 (B)/3455 PCFlash incarceration (current arrest)never1024.2(C)PCWork Release failure51532PCEscapenever	646.9PC	Stalking	never
.370PC / 2684 PC Mentally incompetent/Criminally insane never .800.1(A)VC Evading arrest 5 .800.2(A)VC Evading Peace Officer, disregard safety 10 .000.08 PC Parole warrant - revoke parole supervision never .056PC Parole violation 3 .454 (B)/3455 PC Flash incarceration (current arrest) never .024.2(C)PC Work Release failure 5 .532PC Escape never	47.6PC	Annoying or molesting a child under 18	never
2800.1(A)VC Evading arrest 5 2800.2(A)VC Evading Peace Officer, disregard safety 10 3000.08 PC Parole warrant - revoke parole supervision never 3056PC Parole violation 3 4454 (B)/3455 PC Flash incarceration (current arrest) never 4024.2(C)PC Work Release failure 5 4532PC Escape never	370PC / 2684 PC	<u> </u>	
1800.2(A)VC Evading Peace Officer, disregard safety 10 1000.08 PC Parole warrant - revoke parole supervision never 1056PC Parole violation 3 1454 (B)/3455 PC Flash incarceration (current arrest) never 1024.2(C)PC Work Release failure 5 1532PC Escape never		, , , , ,	
000.08 PC Parole warrant - revoke parole supervision never 056PC Parole violation 3 454 (B)/3455 PC Flash incarceration (current arrest) never 024.2(C)PC Work Release failure 5 4532PC Escape never			
056PC Parole violation 3 454 (B)/3455 PC Flash incarceration (current arrest) never 024.2(C)PC Work Release failure 5 532PC Escape never			
454 (B)/3455 PC Flash incarceration (current arrest) never 1024.2 (C)PC Work Release failure 5 1532PC Escape never			
1024.2(C)PC Work Release failure 5 1532PC Escape never			
532PC Escape never			
	20001(A)VC	Hit and run causing GBI or Death	never

Revised: May 10, 2017 (JZ) Printed on 5/10/2017 at 9:33 AM

COMMUNITY BASED ALTERNATIVES TO CUSTODY (CBAC) LIST OF EXCLUDED CHARGES FOR CBAC PROGRAMS - 2017 **** CONVICTIONS ONLY****

Page 2 of 3

		Post
CHARGE	DESCRIPTION	Incarceration
		Consideration
	DOMESTIC VIOLENCE	
243(E)(1)PC	Non-cohabitating domestic violence/qualifies for Work Release if not current case	*
	*If current case - may qualify for EMP with DV review by LACO Probation personnel	*
243(f)(10) PC	Domestic violence dating relationship/qualifies for WRP if not current case	*
	*If current case - may qualify for EMP with DV review by LACO Probation personnel	*
273(D)PC	Corporal injury to child	never
273.5(A)PC	Corporal injury of spouse/cohabit - May qualify EMP with DV review by LACO Prob	**
	**If not current case, no case in 12 months approved for WRP	
273.6PC	Violation of court order to prevent domestic violence/EMP w/DV review LACO Prob	**
	**If not current case, no case in 12 months approved for WRP	
	·	
23152(a)&(b) VC	DUI Misd qualifies for WRP/three convictions in three yrs requires EMP w/BAT	conditional
23153(A) VC	Felony DUI with injury	never
23153(f) VC	DUI causing bodily injury to person other than driver	never
23175 VC	DUI with priors felony - 3+ convictions in 3yrs (high risk review by CBAC Suprvr)	conditional
	DISQUALIFYING WEAPONS CHARGES	
11370.1A HS	Possession of a controlled substance while armed W/loaded firearm	never
12020(A) PC	Unlawful carrying or possession of weapons (*known gang members only)	never
12021PC (all subs)	Possess firearm	never
12022.1 PC	Commit crime while on Bail or O.R.	never
12022.53 PC	Use of a firearm in the commission of specified felonies (sentence enhancement)	never
12022.9/.95 PC	Felony against a pregnant victim/child death	never
12023 PC	Carry a loaded firearm intent to commit assault	never
12024 PC	Deadly weapon intent assault	never
12025 PC (all subs)	Carry a concealed weapon	never
12028.5 PC	Weapons at the scene of domestic violence	never
12031 PC (all subs)	Carry loaded firearm in vehicle /public place /public street	never
12040 PC	Criminal possession of firearm while wearing a mask	never
12070 PC	Illegal sale of firearms	never
12280 PC	Possession of assault weapon	never
12303 PC (all subs)	Possession of destructive device	never
12308 PC	Use of destructive device with intent to commit murder	never
12309 PC	Use destructive device causing GBI/Death	never
12310 PC	Use destructive device causing Death Possession of materials w/intent to make a destructive device	never
12312 PC 12316 PC		never
12355(a)&(b) PC	Gang injunction - possession of ammunition Bobby trap assembly/possession with intent	
12590 PC (all subs)	Picketing with firearm	never never
25850(A)&(C) PC	Carry a loaded firearm	never
23030(A)&(C) T C	Carry a loaded meanin	licvei
	NEW WEAPONS LAWS	
17500 PC	Deadly weapon intent to assault	never
17510(a) PC	Carry a concealed /loaded firearm while picketing	never
18250 PC	Weapons at scene of domestic violence	never
18710/20 PC	Possession of assault weapon/materials to make destructive device	never
18715/25 PC	Placing a destructive device/transport a destructive device	never
18740/45 PC	Use of destructive device	never
18750/55 PC	Use of destructive device causing GBI/Death	never
19100 PC	Carry concealed explosive	never
20110 PC	Bobby trap assembly/possession with intent to commit assault	never
23920 PC	Possession of an altered firearm	never
25300 PC	Criminal possession of firearm while wearing a mask	never
25400 PC (all subs)	Carry a concealed weapon	never
25800 (A) PC	Use of a firearm in the commission of a felony	never
25850 PC (all subs)	Carry loaded firearm /vehicle/ public place or on public street	never
26350 PC (all subs)	Openly carry unloaded gun on person/veh/public place (*discretionary review)	*
26500 PC	Illegal sale of firearm	never

COMMUNITY BASED ALTERNATIVES TO CUSTODY (CBAC) LIST OF EXCLUDED CHARGES FOR CBAC PROGRAMS - 2017 **** CONVICTIONS ONLY****

Page 3 of 3

CHARGE	DESCRIPTION	Post Incarceration
		Consideration
27235 PC	Concealed firearm	never
29800 (b) PC	Possess firearm adult court conviction while minor	never
29815(a) PC	Possession of firearm/probation violation	never
29900 (a)&(b) PC	Possession firearm violent felon/adult conviction while minor	never
30305 PC	Possession ammunition by felon/gang member	never
30605 PC (all subs)	Possession of assault weapon	never
	PRIOR CONVICTION ENHANCEMENTS	
667.5PC	Prior imprisonment in State Prison	never
667(A)(1)PC	Prior conviction of serious or violent felony	never
1170.12(A)-(D)PC	Prior conviction of serious or violent felony	never
11353HS	Furnish minor controlled substance for use/transport	never
11370.2HS	Prior conviction of narcotic sales or possession for sales	never

The inmate must:

- A. Be fully sentenced (status III)
- B. Be classified with a security level 7 or below
- C. All inmates are subject to discretionary review by Supervisory staff or CBAC placement committee

Note: The CBAC Placement Committee consists of the Population Management Bureau (PMB) Captain, Community Based Alternatives to Custody (CBAC)Lieutenant, and the Community Transition Unit (CTU) Lieutenant.

The inmate shall not have:

- A. Any current or prior violent crimes except misdemeanor domestic violence
- B. Any current or prior sex crimes
- C. Any pending court dates
- D. Any holds or outstanding warrants of any kind
- E. Any extensive weapons violations/crimes committed with firearms
- F. An extensive RAP sheet
- G. Disciplinary problems
- H. No current restraining orders
- I. Committed any crime while in a custody facility (i.e.: 4573PC /4573.3Pc /4573.6PC)
- j. A court commitment excluding program participation

NOTE: ALL INMATES ARE SUBJECT TO DISCRETIONARY REVIEW BY SUPERVISORY STAFF



Revised: May 10, 2017 (JZ) Printed on 5/10/2017 at 9:33 AM

Appendix E-I

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

ABSCOND NOTICE

NAME:

AGENCY / JUDGE:

INMATE/CASE NUMBER: COURT:

OFFENSE: DIV/DEPARTMENT:

ENROLLMENT DATE: X NUMBER:

EXP. COMPLETION DATE: AREA OFFICE:

COMPLETION DATE:

OFFICER / PO:

NO OF DAYS ORDERED: Other EM Numbers

and Cases (if any):

DEFENDANT FAILED TO RETURN HOME WITHIN HIS/HER APPROVED SCHEDULE

DEFENDANT WHEREABOUTS ARE UNKNOWN AT THIS TIME

Appendix E-II

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

COMPLETION NOTICE

NAME: AGENCY / JUDGE:

<u>INMATE/CASE NUMBER:</u> <u>COURT:</u>

OFFENSE: DIV/DEPARTMENT:

ENROLLMENT DATE: X NUMBER:

EXP. COMPLETION DATE: AREA OFFICE:

COMPLETION DATE: OFFICER / PO:

NO OF DAYS ORDERED: Other EM Numbers

and Cases (if any):

ABSCOND INCIDENTS: NON-COMPLIANCE INCIDENTS:

REQUIRED DRUG/ALCOHOL TESTING COMPLETED

Appendix E-III

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

ENROLLMENT NOTICE

NAME: AGENCY / JUDGE:

INMATE/CASE NUMBER: COURT:

OFFENSE: DIV/DEPARTMENT:

ENROLLMENT DATE: X NUMBER:

EXP. COMPLETION DATE: AREA OFFICE:

COMPLETION DATE: OFFICER / PO:

NO OF DAYS ORDERED: Other EM Numbers

and Cases (if any):

- DEFENDANT WILL UNDERGO ELECTRONIC MONITORING
- † DEFENDANT WILL UNDERGO GPS MONITORING
- DEFENDANT WILL UNDERGO BREATH ALCOHOL TESTING
- DEFENDANT WILL UNDERGO DRUG TESTING
- DEFENDANT WILL UNDERGO TRANSDERMAL ALCOHOL MONITORING

Appendix E-IV

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

FAILED TO ENROLL

NAME: AGENCY / JUDGE:

<u>INMATE/CASE NUMBER:</u> <u>COURT:</u>

OFFENSE: DIV/DEPARTMENT:

ENROLLMENT DATE: X NUMBER:

EXP. COMPLETION DATE: AREA OFFICE:

COMPLETION DATE: OFFICER / PO:

NO OF MONTHS ORDERED: Other EM Numbers:

NO OF DAYS ORDERED: and Cases (if any):

DEFENDANT FAILED TO:

- ↑ KEEP HIS / HER SCHEDULED APPOINTMENT
- **†** CONTACT CONTRACTOR MONITORING
- † ENROLL BY COURT ORDERED DATE

COMMENTS

Reference Numbers Called:

Appendix E-V

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

NON-COMPLIANCE REPORT

NAME: AGENCY / JUDGE:

INMATE/CASE NUMBER: COURT:

OFFENSE: DIV/DEPARTMENT:

ENROLLMENT DATE: X NUMBER:

EXP. COMPLETION DATE: AREA OFFICE:

COMPLETION DATE: OFFICER / PO:

NO OF DAYS ORDERED: Other EM Numbers

and Cases (if any):

† This NCR issue resolved

Appendix E-VI

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

STATUS REPORT

NAME: AGENCY / JUDGE: **INMATE/CASE NUMBER:** COURT: OFFENSE: **DIV/DEPARTMENT: ENROLLMENT DATE:** X NUMBER: **EXP. COMPLETION DATE:** AREA OFFICE: **COMPLETION DATE:** OFFICER / PO: NO OF DAYS ORDERED: Other EM Numbers and Cases (if any): **COMMENTS**:

Appendix E-VII

Application No.

County of Los Angeles Probation Department Electronic Monitoring Program

TERMINATION NOTICE

NAME: AGENCY / JUDGE:

INMATE/CASE NUMBER: COURT:

OFFENSE: DIV/DEPARTMENT:

X NUMBER:

ENROLLMENT DATE:

EXP. COMPLETION DATE: AREA OFFICE:

COMPLETION DATE: OFFICER / PO:

NO OF DAYS ORDERED: Other EM Numbers

and Cases (if any):

ABSCOND INCIDENTS: NON-COMPLIANCE INCIDENTS:

CLIENT WAS TERMINATED:

↑ AS ORDERED BY THE COURT

AS ORDERED BY THE SHERIFF'S DEPARTMENT

EFFECTIVE (DATE / TIME): 08/04/2011

Appendix F

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 1 of 7

You have been placed in the Los Angeles County Electronic Monitoring Program (EMP) as an alternative to incarceration. This program uses technology to alert a central monitoring station each time you leave and enter your home, or test positive for the consumption of alcohol. The computer will also report tampering with the equipment, power outages, and loss of phone service.

On the day you begin the program, a transmitter will be fitted to your ankle and a reporting unit will be installed on your telephone. Additional equipment may be necessary if enhanced monitoring is required. This equipment can be removed only after you complete the program, unless other direction is received from the Court.

While on the electronic monitoring program or "house arrest," you are required to remain inside your home except for activities authorized by the Court, Sheriff's Dept., or Probation. It is expected that you will have full-time employment, unless prohibited by your Probation Officer, the Sheriff's Dept., or the Court. Employment must be verified in writing by someone in a supervisory position. In addition, all timecards and paycheck stubs must be submitted as further verification of employment.

At the time of enrollment, a Case Manager will establish a schedule based on your permitted activities such as employment, counseling, drug or alcohol abuse treatment, and any other permitted activities. All overtime or other schedule changes require a Case Manager's approval at least 24 hours in advance. Consideration will be given to same day requests in order to ensure compliance and successful completion of the program. Your Case Manager will also establish scheduled mandatory compliance meetings that you will have to attend at the Contractor's branch office. At these meetings, you will be required to provide documentation for all outside activities.

PROGRAM COMPLIANCE

Participation in the program is voluntary. You should be advised that once you have been enrolled in EMP, section 1203.016 of the California Penal Code states that you can be taken into custody to serve the balance of you sentence for any of the following reasons:

- A. Failure to follow program rules and/or regulations (including providing all required documentation).
- B. Failure of the equipment to perform for any reason which results in the inability to monitor you effectively.
- C. Any negative behavior resulting in the Court or Probation Officer's belief that you may not complete the program successfully.

Client	Initials	
--------	----------	--

Appendix F

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 2 of 7

If you willfully leave your residence without authorization or fail to return to your residence at the prescribed time, you can be prosecuted for escape under Penal Code section 4532.

A Non Compliance Report will be sent to the Court and/or Probation Officer for any program violation, including but not limited to:

- 1. Leaving the range of your monitoring equipment when you are scheduled to be at home.
- Returning home later than your schedule allows.
- Leaving home earlier than your schedule allows.
- 4. Missing scheduled appointments with your case manager.
- 5. Failing to provide acceptable verification of work and other Court/Probation authorized activities.
- 6. Failing to answer all telephone calls when present at your residence.
- 7. Failing to maintain electrical power either by unplugging the equipment or neglecting to pay the electricity bill.
- 8. Failing to maintain telephone/cellular service for any reason.
- 9. Failing to remove custom features from your telephone line, such as call waiting, call forwarding, answering machine voicemail, etc...
- 10. Refusing to perform alcohol or drug tests as required by the Court/Probation or testing positive for either alcohol or drugs.
- 11. If Breath Alcohol Testing (BAT) equipment is installed, failing to blow into the unit and transmit a picture when instructed by the monitoring center.
- 12. Failing to comply with any additional conditions set by the Court or the Probation Officer.
- 13. Tampering with or attempting to remove the ankle transmitter.
- 14. Tampering with or damaging any part of the electronic monitoring equipment.
- 15. Using illegal drugs while on the program.
- 16. Possession of any weapons while on the program.

Client Initials

Appendix F

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 3 of 7

PROGRAM EQUIPMENT

The transmitter fitted to your ankle, the in-home monitor attached to your telephone line, and any other equipment given to you by Contractor's Offender Services is your responsibility.

The ankle strap and transmitter are water resistant and lightweight. It must be worn under your sock, which should not interfere with your normal activities. You can shower while on the program, however, you should not submerge the transmitter in a bathtub, spa or pool.

The in-home unit is connected to the phone and electricity lines in your residence in a fashion similar to an answering machine. This unit transmits a signal through the telephone lines via a (800) phone number. You will not be charged for these calls.

This unit does not have audio recording capabilities and will not monitor your phone conversations. If you unplug either the power cord or the phone line, a Non-Compliance Report will be sent to the Court/Probation Dept. Upon program completion you must return all monitoring equipment to the Contractor's office.

Client	Initials	

Appendix F

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 4 of 7

PROGRAM SCHEDULES

Your curfew schedule is set by your Case Manager based on your work schedule and other permitted activities. All requests for schedule changes must be handled by a case manager. Schedule changes can only be made by phone **Monday through Friday from 10:00am to 4:00pm, 24 hours in advance.** It is your responsibility to plan your approved activities in advance so that last minute schedule changes do not occur.

In the event of a medical emergency, it is your responsibility to notify your Case Manager of the situation during regular business hours. For medical emergencies that occur after business hours, call the 24-hour toll free number (800) 551-4911. You will be responsible for providing written proof of the emergency to your Case Manager the following business day. You will remain in violation of the program rules until proof of any time away is received.

DRIVING PRIVILEGES

If you are driving a vehicle while on the program, you will be required to provide a valid driver's license at the time of your enrollment in the program. A participant whose license has been suspended or revoked should not operate a motor vehicle. If you are observed driving at any time while on the program, a Non Compliance report will be sent immediately to the Court or Probation Dept., which may result in termination from the program.

CLIENT GRIEVANCE PROCEDURE

If you have any questions about your treatment while on the program, you may appeal in writing to the Branch Manager. If no solution is reached at this level, you may submit your grievance to the Contractor's Offender Services, Project Director. Any further complaints must be submitted in writing to the Los Angeles County Probation Department Program Manager, Curtisa Taylor, 12021 S. Wilmington Ave., Bldg 18, STE B001A, Los Angeles, CA 90059.

Client	Initials_	
--------	-----------	--

Appendix F

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 5 of 7

CLIENT AGREEMENT

- 1. I agree not to bring my children or any children into the Contractor's office during my visits with my case manager.
- 2. I agree to admit any person or agent designated by the correctional administrator into my residence at any time for purposes of verifying my compliance with conditions of home detention.
- 3. I agree to remain within the interior premises of my residence at all times, except for the days I work, or to keep appointments for which I have received permission in advance. Any changes in employment schedule, scheduled activities, or requests for appointments will require at least 24 hour advance notice to Contractor's staff. All schedule changes must be requested during the hours of 10:00am and 4:00pm. Monday through Friday.
- 4. I agree to the use of electronic monitoring or supervising devices for the purpose of helping to very my compliance with the rules and regulations of the home detention program. The devices shall not be used to eavesdrop or record any conversation, except a conversation between me and the person supervising me, which is to be used solely for the purpose of voice identification.
- 5. I agree that the Correctional Administrator/Probation Officer may, without further order of the court, immediately retake me into custody to serve the balance of my sentence for any of the following reasons: A) Electronic monitoring or supervising devices are unable for any reason to perform their function at my designated place of home detention. B) If I for any reason no longer meet the established criteria for program participation. C) If I give the Court or Probation Dept. reason to believe that I would not complete the program successfully.
- 6. I understand that it will be necessary for monitoring devices to be installed on my telephone. I agree to maintain telephone services to my designated place of home detention while on the program. I understand that cordless telephones, call waiting, call forwarding and answering machines on the line being used for monitoring are prohibited while I am on the program and I agree to comply with this regulation.
- 7. I agree to respond to all telephone calls generated from the Electronic Monitoring Program staff and monitoring equipment when I am at home regardless of the time of day or night. Failure to respond to phone calls will result in a Non-Compliance Report.
- 8. I agree to maintain service to my designated place of home detention while on the program. I understand that generators or battery-powered devices are not acceptable and I agree to comply with this condition.

Appendix F

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 6 of 7

- 9. I agree to attend regularly scheduled office meetings, at which time I will provide verification of outside activities.
- 10. I understand that the consumption of possession of any drugs not prescribed by a medical doctor is prohibited. I agree to comply with this condition. I understand that I may be required to submit to drug/alcohol testing for the duration of EMP.
- 11. I agree that I will not violate any laws while on the electronic monitoring program.
- 12. I agree to have all firearms that are in my designated place of home detention removed prior to my participation in the program
- 13. I understand that if I am returned to custody for any reason, I will not receive any accelerated release credits and may be subjected to additional loss of good/work time.
- 14. I understand that if I willfully fail to return to my place of home detention later than the period for which I am authorized to be away, or if I make unauthorized departures, I could be prosecuted for escape under Penal Code section 4532, which can carry a sentence of up to six consecutive years in State Prison.
- 15. understand that if I willfully fail this program, I may be excluded from consideration for any other Community Based Alternatives to Custody Program.

Client Initials	
-----------------	--

COUNTY OF LOS ANGELES PROBATION DEPARTMENT ELECTRONIC MONITORING PROGRAM PARTICIPANT CONTRACT

Page 7 of 7

I have been advised that my participation in the Los Angeles County Probation Electronic Monitoring Program (EMP) is voluntary and that, if I prefer, I may serve my sentence in custody at a jail facility. The program guidelines have been explained to me and a copy given to me. I agree to comply with all program rules and regulations. I further understand that failure to follow program guidelines may result in my immediate return to custody without warrant or court order to serve the balance of my sentence.

I have read and received a copy of the aforementioned rules and regulations and agree to comply with the terms and conditions of the Electronic Monitoring Program.

Participant Name (Print)	
r articipant Name (r mr.)	Date
Participant Signature	Date
Case Manager Name	 Date
Case Manager Signature	

APPENDIX C SAMPLE CONTRACT



CONTRACT

BY AND BETWEEN

COUNTY OF LOS ANGELES

AND

(CONTRACTOR)

FOR

COMPREHENSIVE ELECTRONIC MONITORING SERVICES

PAR	RAGRA	PH <u>TITLE</u>	<u>PAGE</u>
REC	ITALS		1
1	APP	LICABLE DOCUMENTS	1
2	DEF	INITIONS	3
	2.1	Standard Definitions	3
3	WOF	RK	4
4	TER	M OF CONTRACT	5
5	CON	TRACT SUM	5
	5.1	Total Contract Sum	5
	5.2	Written Approval for Reimbursement	6
	5.3	Notification of 75% of Total Contract Sum	6
	5.4	No Payment for Services Provided Following Expiration-Termination of Contract	
	5.5	Invoices and Payments	7
	5.6	Cost of Living Adjustments (COLA's)	8
	5.7	Default Method of Payment: Direct Deposit or Electronic Funds Tran	ısfer9
6	ADM	IINISTRATION OF CONTRACT - COUNTY	9
	6.1	County Administration	9
	6.2	County's Contract Manager	9
	6.3	County's Program Manager	10
	6.4	County's Contract Monitor	10
7	ADM	IINISTRATION OF CONTRACT - CONTRACTOR	10
	7.1	Contractor Administration	10
	7.2	Contractor's Staff	11
	7.3	Approval of Contractor's Staff	11
	7.4	Contractor's Staff Identification	11
	7.5	Background and Security Investigations	11
	7.6	Confidentiality	12
8	STA	NDARD TERMS AND CONDITIONS	14
	8.1	Amendments	14
	8.2	Assignment and Delegation/Mergers or Acquisitions	14
	8.3	Authorization Warranty	15
	8.4	Budget Reductions	15
	8.5	Complaints	16

<u>PARAGRAPH</u>	<u>TITLE</u>	<u>PAGE</u>
8.6	Compliance with Applicable Law	17
8.7	Compliance with Civil Rights Laws	17
8.8	Compliance with the County's Jury Service Program	18
8.9	Conflict of Interest	19
8.10	Consideration of Hiring County Employees Targeted for Layoff or Re- Employment List	
8.11	Consideration of Hiring GAIN-START Participants	20
8.12	Contractor Responsibility and Debarment	21
8.13	Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law	23
8.14	Contractor's Warranty of Adherence to County's Child Support	
	Compliance Program	24
8.15	County's Quality Assurance Plan	24
8.16	Damage to County Facilities, Buildings or Grounds	24
8.17	Employment Eligibility Verification	25
8.18	Counterparts and Electronic Signatures and Representations	25
8.19	Fair Labor Standards	26
8.20	Force Majeure	26
8.21	Governing Law, Jurisdiction, and Venue	
8.22	Independent Contractor Status	
8.23	Indemnification	27
8.24	General Provisions for all Insurance Coverage	
8.25	Insurance Coverage	
8.26	Liquidated Damages	
8.27	Most Favored Public Entity	
8.28	Nondiscrimination and Affirmative Action	
8.29	Non Exclusivity	
8.30	Notice of Delays	
8.31	Notice of Disputes	
8.32	Notice to Employees Regarding the Federal Earned Income Credit	
8.33	Notice to Employees Regarding the Safely Surrendered Baby Law	
8.34	Notices	
8.35	Prohibition Against Inducement or Persuasion	
8.36	Public Records Act	39

PARAGRAPH	<u>TITLE</u>	<u>PAGE</u>
8.37	Publicity	40
8.38	Record Retention and Inspection-Audit Settlement	40
8.39	Recycled Bond Paper	42
8.40	Subcontracting	42
8.41	Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program	43
8.42	Termination for Convenience	43
8.43	Termination for Default	44
8.44	Termination for Improper Consideration	45
8.45	Termination for Insolvency	46
8.46	Termination for Non-Adherence of County Lobbyist Ordinance	47
8.47	Termination for Non-Appropriation of Funds	47
8.48	Validity	47
8.49	Waiver	47
8.50	Warranty Against Contingent Fees	47
8.51	Warranty of Compliance with County's Defaulted Property Tax	
	Reduction Program	48
8.52	Termination for Breach of Warranty to Maintain Compliance with	
	County's Defaulted Property Tax Reduction Program	48
8.53	Time off for Voting	48
8.54	Compliance with County's Zero Tolerance Policy on Human Traffick	ing . 49
8.55	Intentionally Omitted	49
8.56	Compliance with Fair Chance Employment Practices	49
8.57	Compliance with the County Policy of Equity	49
8.58	Prohibition from Participation in Future Solicitation(s)	50
8.59	Injury and Illness Prevention Program	50
8.60	Campaign Contribution Prohibition Following Final Decision in Contr Proceeding	
8.61	Compliance with County's Women in Technology Hiring Initiative	50
9 UNIQ	UE TERMS AND CONDITIONS	51
9.1	Health Insurance Portability and Accountability Act of 1996 (HIPAA)	51
9.2	Ownership of Materials, Software and Copyright	51
9.3	Patent, Copyright and Trade Secret Indemnification	52
9.4	Data Destruction	53

<u>PARAGRAPH</u>		<u>TITLE</u>	<u>PAGE</u>
	9.5	Contractor's Charitable Activities Compliance	53
	9.6	Intentionally Omitted	54
	9.7	Local Small Business Enterprise (LSBE) Preference Program	54
	9.8	Social Enterprise (SE) Preference Program	55
	9.9	Disabled Veteran Business Enterprise (DVBE) Preference Program	56
10	SURV	IVAL	57
SIGNATURES		S	59
		I	
		I	
		I	
		ı	
		,	
		I	
		I	
		I	
		I	
		I	
		I	
		1	
		I	
		,	
		I	
		,	
		I	
		I	
		I	
		<u>.</u>	

STANDARD EXHIBITS

(NOT ATTACHED – TO BE INCLUDED IN FINAL CONTRACT)

Α	Statement of Work (SOW)		
B, B1,	and B2 Pricing Sheets		
С	Intentionally Omitted		
D	Contractor's EEO Certification		
E	County's Administration		
F	Contractor's Administration		
G1	Contractor Acknowledgement and Confidentiality Agreement		
G2	Contractor Employee Acknowledgement and Confidentiality Agreement		
G3	${\bf Contractor\ Non-Employee\ Acknowledgement\ and\ Confidentiality\ Agreement\ }.$		
G4	Employee's Acknowledgement of Employer		
Н	Jury Service Ordinance		
I	Safely Surrendered Baby Law		
J	Intentionally Omitted		
K	Intentionally Omitted		
L	Intentionally Omitted		
М	Intentionally Omitted		
N	Business Associate Agreement under the Health Insurance Portability and Accountability Act of 1996 (HIPAA)		
0	Charitable Contributions Certification		
Р	Background Request Form		
Q	Intentionally Omitted		
R	Defaulted Property Tax Reduction Program/Form		
S	Contract Discrepancy Report		
Т	Confidentiality of CORI Information		
U	Performance Requirements Summary (PRS) Chart		
V	EQUIPMENT REQUIREMENTS		
W	EMP OPERATIONAL PROCEDURES MANUAL		

SAMPLE CONTRACT

CONTRACT BETWEEN

COUNTY OF LOS ANGELES

	AND
_	(CONTRACTOR)
	FOR
C	OMPREHENSIVE ELECTRONIC MONITORING SERVICES

This Contract ("Contract") made and entered into this _____ day of ______, 20___ by and between the County of Los Angeles, hereinafter referred to as "County" and _____, hereinafter referred to as "Contractor". ______ is located at _____.

RECITALS

WHEREAS, the County of Los Angeles Probation Department has a need for the Contractor to provide Comprehensive Electronic Monitoring Services; and

WHEREAS, the County through its Probation Officer, is authorized to Contract under California Governmental Code Section 31000.

WHEREAS, the County of Los Angeles through its Probation Officer, is authorized under California Penal Code Section 1203.14 and otherwise to engage in activities designed to monitor persons placed on probation and to prevent juvenile and adult delinquency such as contemplated by this Contract; and

WHEREAS, the Contractor is duly qualified to engage in the business of providing services as set forth hereunder and warrants that it possesses the licenses, competence, experience, preparation, organization and staffing and facilities to provide services as described in this Contract;

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for good and valuable consideration, the parties agree to the following:

1 APPLICABLE DOCUMENTS

1.1 Exhibits A, B, D, E, F, G1, G2, G3, G4, H, I, N, O, P, R, S, T, U, V, and W are attached to and form a part of this Contract. In the event of any conflict or inconsistency in the definition or interpretation of any word, responsibility, schedule, or the contents or description of any task,

deliverable, goods, service, or other work, or otherwise between the base Contract and the Exhibits, or between Exhibits, such conflict or inconsistency shall be resolved by giving precedence first to the terms and conditions of the Contract and then to the Exhibits according to the following priority.

Standard Exhibits:

- 1.1 EXHIBIT A Statement of Work
- 1.2 EXHIBIT B, B1 and B2 Pricing Sheets
- 1.3 EXHIBIT C Intentionally Omitted
- 1.4 EXHIBIT D Contractor's EEO Certification
- 1.5 EXHIBIT E County's Administration
- 1.6 EXHIBIT F Contractor's Administration
- 1.7 EXHIBIT G1 Contractor Acknowledgement and Confidentiality
 Agreement
 - EXHIBIT G2 Contractor Employee Acknowledgement and Confidentiality Agreement
 - EXHIBIT G3 Contractor Non-Employee Acknowledgement and Confidentiality Agreement
 - EXHIBIT G4 Employer's Acknowledgment of Employer
- 1.8 EXHIBIT H Jury Service Ordinance
- 1.9 EXHIBIT I Safely Surrendered Baby Law
- 1.10 EXHIBIT J Intentionally Omitted
- 1.11 EXHIBIT K Intentionally Omitted
- 1.12 EXHIBIT L Intentionally Omitted
- 1.13 EXHIBIT M Intentionally Omitted
- 1.14 EXHIBIT N Business Associate Agreement under the Health Insurance Portability and Accountability Act of 1996 (HIPPA)

- 1.15 EXHIBIT O Charitable Contributions Certification
- 1.16 EXHIBIT P Background Request Form
- 1.17 EXHIBIT Q Intentionally Omitted
- 1.18 EXHIBIT R Defaulted Property Tax Reduction Program/Form
- 1.19 EXHIBIT S Contract Discrepancy Report
- 1.20 EXHIBIT T Confidentiality of CORI Information
- 1.21 EXHIBIT U Performance Requirements Summary (PRS) Chart
- 1.22 EXHIBIT V Equipment Requirements
- 1.23 EXHIBIT W EMP Operational Procedures Manual

This Contract constitutes the complete and exclusive statement of understanding between the parties, and supersedes all previous Contracts, written and oral, and all communications between the parties relating to the subject matter of this Contract. No change to this Contract shall be valid unless prepared pursuant to Paragraph 8.1 (Amendments) and signed by both parties.

2 DEFINITIONS

2.1 Standard Definitions:

- 2.1.1 The headings herein contained are for convenience and reference only and are not intended to define the scope of any provision thereof. The following words as used herein shall be construed to have the following meaning, unless otherwise apparent from the context in which they are used.
 - 2.1.1.1 **Board of Supervisors (Board):** The Board of Supervisors of the County of Los Angeles acting as governing body.
 - 2.1.1.2 **Contract:** This agreement executed between the County and the Contractor. Included are all supplemental agreements amending or extending the service to be performed. The Contract sets forth the terms and conditions for the issuance and performance of all tasks, deliverables, services and other work.
 - 2.1.1.3 **Contractor:** The person or persons, sole proprietor, partnership, joint venture, corporation or other legal entity

- who has entered into an agreement with the County to perform or execute the work covered by this Contract.
- 2.1.1.4 **Contractor's Project Director:** Person designated by the Contractor to administer the Contract operations after the Contract award.
- 2.1.1.5 **County's Contract Manager:** Person designated by the County with authority for the County on contractual or administrative matters relating to the Contract.
- 2.1.1.6 **County's Contract Monitor:** Person designated by the County to monitor the Contract and provide reports to the County's Contract Manager and the County's Program Manager.
- 2.1.1.7 **County's Program Manager:** Person designated by the County to manage the daily operations under this Contract.
- 2.1.1.8 **Day(s):** Calendar day(s) unless otherwise specified.
- 2.1.1.9 **Fiscal Year:** The twelve (12) month period beginning July 1st and ending the following June 30th.
- 2.1.1.10 **Statement of Work:** The directions, provisions, and requirements provided herein and special provisions pertaining to the method, frequency, manner and place of performing the Contract services.
- 2.1.1.11 **Subcontract:** An agreement by the Contractor to employ a subcontractor to provide services to fulfill this Contract.
- 2.1.1.12 **Subcontractor:** Any individual, person or persons, sole proprietor, firm, partnership, joint venture, corporation, or other legal entity furnishing supplies, services of any nature, equipment, and/or materials to the Contractor in furtherance of the Contractor's performance of this Contract, at any tier, under oral or written agreement.

3 WORK

3.1 Pursuant to the provisions of this Contract, the Contractor shall fully perform, complete and deliver on time, all tasks, deliverables, services and other work as set forth in herein.

3.2 If the Contractor provides any tasks, deliverables, goods, services, or other work, other than as specified in this Contract, the same shall be deemed to be a gratuitous effort on the part of the Contractor, and the Contractor shall have no claim whatsoever against the County.

4 TERM OF CONTRACT

- 4.1 The term of this Contract shall commence following Board approval and will continue for a five (5) year period, unless terminated or extended in whole or in part, as provided in the Contract. Contingent upon available funding, the Contract may be extended by the Chief Probation Officer and the authorized official of the Contractor, by mutual written agreement, for up to two (2) additional and consecutive thirty (30) month periods for a total maximum Contract term of ten (10) years.
- 4.2 Contingent upon available funding, the term of the Contract may also be extended beyond the stated expiration date on a month-to-month basis, for a period of time not to exceed six (6) months, upon the written request of the Chief Probation Officer and the written concurrence of the Contractor. All terms of the Contract in effect at the time of extending the term shall remain in effect for the duration of the extension.

The County maintains databases that track/monitor the Contractor's performance history. Information entered into such databases may be used for a variety of purposes, including determining whether the County will exercise a Contract term extension option.

4.3 The Contractor shall notify the County of Los Angeles Probation Department when this Contract is within six (6) months of the expiration of the term as provided for hereinabove. Upon occurrence of this event, the Contractor shall send written notification to the County of Los Angeles Probation Department at the address herein provided in Exhibit E (County's Administration).

5 CONTRACT SUM

5.1 Total Contract Sum

satisfactorily perform, provide, and complete all tasks, deliverables, goods, services, and other work specified in this Contract.

The Contractor shall return to the County any unspent funds in excess of actual administrative/indirect costs under this Contract at the end of each Contract term. The Contractor must return to the County any funds received in excess of administrative/indirect costs. The Contractor agrees to be bound by applicable County unsupported and disallowed cost procedures, rules and regulations, and to repay to the County any amount, with its earned interest, which is found to violate the terms of this Contract or applicable County provisions.

5.2 Written Approval for Reimbursement

5.2.1 The Contractor shall not be entitled to payment or reimbursement for any tasks, deliverables, goods, services, or other work performed or provided, nor for any incidental or administrative expenses whatsoever incurred in or incidental to performance hereunder, except as specified herein. Assumption or takeover of any of the Contractor's duties, responsibilities, or obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever, shall not occur except with the County's express prior written approval.

5.3 Notification of 75% of Total Contract Sum

5.3.1 The Contractor shall maintain a system of record keeping that will allow the Contractor to determine when it has incurred seventy-five percent (75%) of the total Contract Sum under this Contract, as specified in Section 5.1.1 above. Upon occurrence of this event, the Contractor shall send written notification to Los Angeles County Probation Department at the address herein provided in Exhibit D (County's Administration).

5.4 No Payment for Services Provided Following Expiration- Termination of Contract

5.4.1 The Contractor shall have no claim against the County for payment of any money or reimbursement, of any kind whatsoever, for any task, deliverable, good, service, or other work performed or provided by the Contractor after the expiration or other termination of this Contract. Should the Contractor receive any such payment it shall immediately notify the County and shall immediately repay all

such funds to the County. Payment by the County for services rendered after expiration-termination of this Contract shall not constitute a waiver of County's right to recover such payment from the Contractor. This provision shall survive the expiration or other termination of this Contract.

5.5 Invoices and Payments

- 5.5.1 The Contractor shall invoice the County monthly in arrears only for providing the tasks, deliverables, goods, services, and other work specified in Statement of Work and elsewhere hereunder. The Contractor shall prepare invoices that were provided by Contractor provided during the prior month, which shall include the charges owed to the Contractor by the County under the terms of this Contract. The Contractor's invoiced amounts shall be as provided in Exhibit B1, B2 and B3 (Pricing Sheet) and the Contractor shall be paid only for the tasks, deliverables, goods, services, and other work approved in writing by the County. If the County does not approve work in writing no payment shall be due to the Contractor for that work.
- 5.5.2 The Contractor's invoices shall be priced in accordance with Exhibit B1, B2 and B3 (Pricing Sheet).
- 5.5.3 The Contractor's invoices shall contain the information set forth in Statement of Work describing the tasks, deliverables, goods, services, and other work, hours spent, and facility and/or other work for which payment is claimed, and shall tie back to the applicable Exhibit B1, B2, and B3 (Pricing Sheet) item for which the amount is billed and attach a detail invoice monthly report.
 - Each invoice needs to identify quantity provided for each applicable item in the pricing sheet and attaches a detailed listing of all services provided during the month by individual probationer/client.
- 5.5.4 The Contractor shall submit the monthly invoices to the County by the 15th calendar day of the month following the month of service.
- 5.5.5 All invoices under this Contract shall be submitted in two (2) copies to the following address:

Comprehensive Electronic Monitoring County of Los Angeles Probation Department Attention: Fiscal Department P-73 9150 East Imperial Highway Downey, California 90242

5.5.6 County Approval of Invoices

All invoices submitted by the Contractor for payment must have the written approval of the County's Program Manager prior to any payment thereof. In no event shall the County be liable or responsible for any payment prior to such written approval. Approval for payment will not be unreasonably withheld.

5.5.7 Local Small Business Enterprises – Prompt Payment Program

Certified Local Small Business Enterprises (LSBEs) will receive prompt payment for services they provide to the County departments. Prompt payment is defined as fifteen (15) calendar days after receipt of an undisputed invoice.

5.6 Cost of Living Adjustments (COLA's)

If requested by the Contractor, the Contract (hourly, daily, monthly, etc.) amount may at the sole discretion of the County, be increased annually based on the most recent published percentage change in the U.S. Department of Labor, Bureau of Labor Statistics' Consumer Price Index for Urban Consumers (CPI-U) for the Los Angeles-Long Beach-Anaheim Area for the twelve (12) month period preceding the Contract commencement anniversary date, which will be the effective date for any Cost of Living Adjustment (COLA). However, any increase must not exceed the general salary movement granted to County employees as determined by the Chief Executive Officer as of each July 1 for the prior twelve (12) month period. Furthermore, should fiscal circumstances ultimately prevent the Board from approving any increase in County employee salaries, no COLA will be granted. Where the County decides to grant a COLA pursuant to this paragraph for living wage contracts, it may, in its sole discretion exclude the cost of labor (including the cost of wages and benefits paid to employees providing services under this Contract) from the base upon which a COLA is calculated, unless the Contractor can show that their labor cost will actually increase. Further, before any COLA increase will take effect and become part of this Contract, it will require a written amendment to this Contract first, that has been formally approved and executed by the parties. To request a COLA, Contractor must submit a written request along with appropriate justification to the Contract Analyst 60 days prior to exercising the additional option year periods identified in Paragraph 4.0.

5.7 Default Method of Payment: Direct Deposit or Electronic Funds Transfer

- 5.7.1 The County, at its sole discretion, has determined that the most efficient and secure default form of payment for goods and/or services provided under an agreement/contract with the County shall be Electronic Funds Transfer (EFT) or direct deposit, unless an alternative method of payment is deemed appropriate by the Auditor-Controller (A-C).
- 5.7.2 The Contractor shall submit a direct deposit authorization request via the website https://directdeposit.lacounty.gov with banking and vendor information, and any other information that the A-C determines is reasonably necessary to process the payment and comply with all accounting, record keeping, and tax reporting requirements.
- 5.7.3 Any provision of law, grant, or funding agreement requiring a specific form or method of payment other than EFT or direct deposit shall supersede this requirement with respect to those payments.
- 5.7.4 At any time during the duration of the Contract, a Contractor may submit a written request for an exemption to this requirement. Such request must be based on specific legal, business, or operational needs and explain why the payment method designated by the A-C is not feasible and an alternative is necessary. The A-C, in consultation with the contracting department(s), shall decide whether to approve exemption requests.

6 ADMINISTRATION OF CONTRACT - COUNTY

6.1 County Administration

A listing of all County Administration referenced in the following subparagraphs are designated in Exhibit E (County's Administration). The County will notify the Contractor in writing of any change in the names or addresses shown.

6.2 County's Contract Manager

- 6.2.1 The role of the County's Contract Manager may include:
 - 6.2.1.1 Coordinating with the Contractor and ensuring the Contractor's performance of the Contract; however, in no

event shall the Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby; and

6.2.1.2 Upon request of the Contractor, providing direction to the Contractor, as appropriate in areas relating to County policy, information requirements, and procedural requirements; however, in no event, shall the Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

6.3 County's Program Manager

- 6.3.1 The role of the County's Program Manager is authorized to include:
 - 6.3.1.1 Meeting with the Contractor's Project Director on a regular basis; and
 - 6.3.1.2 Inspecting all tasks, deliverables, goods, services, or other work provided by or on behalf of the Contractor; however, in no event shall the Contractor's obligation to fully satisfy all of the requirements of this Contract be relieved, excused or limited thereby.

The County's Program Manager is not authorized to make any changes in any of the terms and conditions of this Contract and is not authorized to further obligate the County in any respect whatsoever.

6.4 County's Contract Monitor

The County's Contract Monitor is responsible for the monitoring of the Contract and the Contractor. The County's Contract Monitor provides reports to the County's Contract Manager and the County's Program Manager.

7 ADMINISTRATION OF CONTRACT - CONTRACTOR

7.1 Contractor Administration

A listing of all the Contractor's Administration referenced in the following paragraphs is designated in Exhibit F (Contractor's Administration). The Contractor will notify the County in writing of any change in the names or addresses shown.

7.2 Contractor's Staff

- 7.2.1 The Contractor shall have a Project Director pursuant to Section 6.3 (Project Director) of Exhibit A (Statement of Work).
- 7.2.2 The Contractor shall be responsible for providing competent staff pursuant to Section 6.4 (Personnel) of Exhibit A (Statement of Work).

7.3 Approval of Contractor's Staff

The County has the absolute right to approve or disapprove all of the Contractor's staff performing work hereunder and any proposed changes in the Contractor's staff, including, but not limited to, the Contractor's Project Director.

7.4 Contractor's Staff Identification

The Contract shall provide, at the Contractor's expense, all staff providing services under this Contract with a photo identification badge.

7.5 Background and Security Investigations

Background and security investigations of the Contractor's staff are required as a condition of beginning and continuing work under this Contract. The cost of background checks is the responsibility of the Contractor. The Contractor shall be responsible for the ongoing implementation and monitoring of Subparagraphs 7.5.1 through 7.5.6 of this Contract. On at least a quarterly basis, the Contractor shall report, in writing, monitoring results to the County, indicating compliance or problem areas. Elements of the monitoring report shall receive prior written approval from the County.

7.5.1 The Contractor shall submit the names of the Contractor's or the subcontractor's employees to the County's Program Manager prior to the employee starting work on this Contract. The County appointments will schedule to conduct background investigation/record checks based on fingerprints of the Contractor's or the subcontractor's employees. The County shall have the right to conduct background investigations of the Contractor's or the subcontractor's employees at any time. The Contractor's or the subcontractor's employees shall not begin work on this Contract before receiving written notification of clearance from the County.

- 7.5.2 No personnel employed by the Contractor or the subcontractor for this service having access to Probation information or records shall have a criminal conviction record or pending criminal trial unless such information has been fully disclosed to the County and employment of the employee for this service is approved in writing by the County.
- 7.5.3 The County reserves the right, in its sole discretion, to preclude the Contractor or the subcontractor from employment or continued employment of any individual performing services under this Contract.
- 7.5.4 No Contractor or subcontractor staff providing services under this Contract shall be on active probation or parole.
- 7.5.5 The Contractor or the subcontractor staff performing services under this Contract shall be under a continuing obligation to disclose any prior or subsequent criminal conviction record or any pending criminal trial to the County.
- 7.5.6 Because the County is charged by the State for checking the criminal records of the Contractor's or the subcontractor's employees; the County will bill the Contractor to recover these expenses. The current amount is forty-nine (\$49.00) dollars per record check, which is subject to change by the State.

7.6 Confidentiality

The Contractor shall be responsible for safeguarding all County information provided for use by the Contractor.

- 7.6.1 The Contractor shall maintain the confidentiality of all records and information in accordance with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures relating to confidentiality, including, without limitation, County policies concerning information technology security and the protection of confidential records and information.
- 7.6.2 The Contractor shall inform all of its officers, employees, agents and subcontractors providing services hereunder of the confidentiality provisions of this Contract.
 - 7.6.2.1 The Contractor shall sign and adhere to the provisions of Exhibit G1 (Contractor Acknowledgement and Confidentiality Agreement).

- 7.6.2.2 The Contractor shall require each employee performing services covered by this Contract to sign and adhere to the provisions of Exhibit G2 (Contractor Employee Acknowledgement and Confidentiality Agreement)
- 7.6.2.3 The Contractor shall require each non-employee performing services covered by this Contract to sign and adhere to the provisions of Exhibit G3 (Contractor Non-Employee Acknowledgement and Confidentiality Agreement)
- 7.6.3 The Contractor shall indemnify, defend, and hold harmless the County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs and expenses, including, without limitation, defense costs and legal, accounting and other expert, consulting, or professional fees, arising from, connected with, or related to any failure by the Contractor, its officers, employees, agents, or subcontractors, to comply with this Paragraph 7.6 (Confidentiality), as determined by the County in its sole judgment. Any legal defense pursuant to the Contractor indemnification obligations under this Paragraph 7.6 (Confidentiality) shall be conducted by the Contractor and performed by counsel selected by the Contractor and approved by the County. Notwithstanding the preceding sentence, the County shall have the right to participate in any such defense at its sole cost and expense, except that in the event the Contractor fails to provide the County with a full and adequate defense, as determined by the County in its sole judgment, the County shall be entitled to retain its own counsel, including, without limitation, County Counsel, and reimbursement from the Contractor for all such costs and expenses incurred by the County in doing so. The Contractor shall not have the right to enter into any settlement, agree to any injunction, or make any admission, in each case, on behalf of the County without the County's prior written approval.

7.6.4 Confidentiality of Adult and Juvenile Records

By state law (California Welfare and Institutions Code sections 827 and 828, and Penal Code sections 1203.05, 1203.09, and 11140 through 11144) all adult and juvenile records and Probation case information provided to the Contractor is confidential and no such information shall be disclosed except those authorized employees of the County of Los Angeles Probation Department and law enforcement agencies.

- 7.6.5 The Contractor's employees shall be given copies of all cited code sections, and a CORI form to sign, as provided in Exhibit T (Confidentiality of CORI Information) regarding confidentiality of the information in adult and juvenile records. The Contractor shall retain original CORI forms and forward copies to the County's Program Manager within five (5) business days of start of employment.
- 7.6.6 <u>Violations:</u> The Contractor agrees to inform all of its employees, agents, subcontractors, and partners of the above provision and that any person knowingly and intentionally violating the provisions of said state law is guilty of a misdemeanor.

8 STANDARD TERMS AND CONDITIONS

8.1 Amendments

- 8.1.1 For any change which affects the scope of work, term, Contract sum, payments, or any term or condition included under this Contract, an amendment to the Contract shall be prepared and executed by the Contractor and by the Chief Probation Officer or his/her designee.
- 8.1.2 The County's Board of Supervisors or Chief Executive Officer or designee may require the addition and/or change of certain terms and conditions in the Contract during the term of this Contract. The County reserves the right to add and/or change such provisions as required by the County's Board of Supervisors or Chief Executive Officer. To implement such changes, an Amendment to the Contract shall be prepared and executed by the Contractor and by the Chief Probation Officer or his/her designee.
- 8.1.3 The Chief Probation Officer or his/her designee, may at his/her sole discretion, authorize extensions of time as defined in Paragraph 4 (Term of Contract). The Contractor agrees that such extensions of time shall not change any other term or condition of this Contract during the period of such extensions. To implement an extension of time, an Amendment to the Contract shall be prepared and executed by the Contractor and by the Chief Probation Officer or his/her designee.

8.2 Assignment and Delegation/Mergers or Acquisitions

8.2.1 The Contractor shall notify the County of any pending acquisitions/mergers of its company unless otherwise legally prohibited from doing so. If the Contractor is restricted from legally

notifying the County of pending acquisitions/mergers, then it should notify the County of the actual acquisitions/mergers as soon as the law allows and provide to the County the legal framework that restricted it from notifying the County prior to the actual acquisitions/mergers.

- 8.2.2 The Contractor shall not assign, exchange, transfer, or delegate its rights or duties under this Contract, whether in whole or in part, without the prior written consent of the County, in its discretion, and any attempted assignment, delegation, or otherwise transfer of its rights or duties, without such consent shall be null and void. For purposes of this paragraph, County consent shall require a written Amendment to the Contract, which is formally approved and executed by the parties. Any payments by the County to any approved delegate or assignee on any claim under this Contract shall be deductible, at the County's sole discretion, against the claims, which the Contractor may have against the County.
- 8.2.3 Any assumption, assignment, delegation, or takeover of any of the Contractor's duties, responsibilities, obligations, or performance of same by any person or entity other than the Contractor, whether through assignment, subcontract, delegation, merger, buyout, or any other mechanism, with or without consideration for any reason whatsoever without the County's express prior written approval, shall be a material breach of the Contract which may result in the termination of this Contract. In the event of such termination, the County shall be entitled to pursue the same remedies against the Contractor as it could pursue in the event of default by the Contractor.

8.3 Authorization Warranty

The Contractor represents and warrants that the person executing this Contract for the Contractor is an authorized agent who has actual authority to bind the Contractor to each and every term, condition, and obligation of this Contract and that all requirements of the Contractor have been fulfilled to provide such actual authority.

8.4 Budget Reductions

In the event that the County's Board of Supervisors adopts, in any fiscal year, a County Budget which provides for reductions in the salaries and benefits paid to the majority of County employees and imposes similar reductions with respect to County Contracts, the County reserves the right to reduce its payment obligation under this Contract correspondingly for that fiscal year and any subsequent fiscal year during the term of this

Contract (including any extensions), and the services to be provided by the Contractor under this Contract shall also be reduced correspondingly. The County's notice to the Contractor regarding said reduction in payment obligation shall be provided within thirty (30) calendar days of the Board's approval of such actions. Except as set forth in the preceding sentence, the Contractor shall continue to provide all of the services set forth in this Contract.

8.5 Complaints

8.5.1 The Contractor shall develop, maintain and operate procedures for receiving, investigating and responding to complaints.

8.5.2 Complaint Procedures

- 8.5.2.1 Within fifteen (15) business days after the Contract effective date, the Contractor shall provide the County with the Contractor's policy for receiving, investigating and responding to user complaints.
- 8.5.2.2 The County will review the Contractor's policy and provide the Contractor with approval of said plan or with requested changes.
- 8.5.2.3 If the County requests changes in the Contractor's policy, the Contractor shall make such changes and resubmit the plan within five (5) business days for County approval.
- 8.5.2.4 If, at any time, the Contractor wishes to change the Contractor's policy, the Contractor shall submit proposed changes to the County for approval before implementation.
- 8.5.2.5 The Contractor shall preliminarily investigate all complaints and notify the County's Program Manager of the status of the investigation within five (5) business days of receiving the complaint.
- 8.5.2.6 When complaints cannot be resolved informally, a system of follow-through shall be instituted which adheres to formal plans for specific actions and strict time deadlines.

8.5.2.7 Copies of all written responses shall be sent to the County's Program Manager within three (3) business days of mailing to the complainant.

8.6 Compliance with Applicable Law

- 8.6.1 In the performance of this Contract, the Contractor shall comply with all applicable Federal, State and local laws, rules, regulations, ordinances, directives, guidelines, policies and procedures, and all provisions required thereby to be included in this Contract are hereby incorporated herein by reference.
- The Contractor shall indemnify, defend, and hold harmless the 8.6.2 County, its officers, employees, and agents, from and against any and all claims, demands, damages, liabilities, losses, costs, and expenses, including, without limitation, defense costs and legal, accounting and other expert, consulting or professional fees, arising from, connected with, or related to any failure by the Contractor, its officers, employees, agents, or the subcontractors, to comply with any such laws, rules, regulations, ordinances, directives, guidelines, policies, or procedures, as determined by the County in its sole judgment. Any legal defense pursuant to the Contractor's indemnification obligations under Paragraph 8.6 (Compliance with Applicable Law) shall be conducted by the Contractor and performed by counsel selected by the Contractor and approved by the County. Notwithstanding the preceding sentence, the County shall have the right to participate in any such defense at its sole cost and expense, except that in the event the Contractor fails to provide the County with a full and adequate defense, as determined by the County in its sole judgment, the County shall be entitled to retain its own counsel, including, without limitation, County Counsel. reimbursement from the Contractor for all such costs and expenses incurred by the County in doing so. The Contractor shall not have the right to enter into any settlement, agree to any injunction or other equitable relief, or make any admission, in each case, on behalf of the County without the County's prior written approval.

8.7 Compliance with Civil Rights Laws

The Contractor hereby assures that it will comply with Subchapter VI of the Civil Rights Act of 1964, 42 USC Sections 2000 (e) (1) through 2000 (e) (17), to the end that no person shall, on the grounds of race, creed, color, sex, religion, ancestry, age, condition of physical handicap, marital status, political affiliation, or national origin, be excluded from participation

in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract. The Contractor shall comply with Exhibit D (Contractor's EEO Certification).

8.8 Compliance with the County's Jury Service Program

8.8.1 **Jury Service Program:**

This Contract is subject to the provisions of the County's ordinance entitled Contractor Employee Jury Service ("Jury Service Program") as codified in Sections 2.203.010 through 2.203.090 of the Los Angeles County Code, a copy of which is attached as Exhibit H (Jury Duty Ordinance) and incorporated by reference into and made a part of this Contract.

8.8.2 Written Employee Jury Service Policy

- 8.8.2.1 Unless the Contractor has demonstrated to the County's satisfaction either that the Contractor is not a "Contractor" as defined under the Jury Service Program (Section 2.203.020 of the County Code) or that the Contractor qualifies for an exception to the Jury Service Program (Section 2.203.070 of the County Code), the Contractor shall have and adhere to a written policy that provides that its Employees shall receive from the Contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that Employees deposit any fees received for such jury service with the Contractor or that the Contractor deduct from the Employee's regular pay the fees received for jury service.
- 8.8.2.2 For purposes of this paragraph, "Contractor" means a person, partnership, corporation or other entity which has a Contract with the County or a subcontract with a County Contractor and has received or will receive an aggregate sum of fifty thousand dollars (\$50,000) or more in any twelve (12) month period under one or more County Contracts or subcontracts. "Employee" means any California resident who is a full-time employee of the Contractor. "Full-time" means forty (40) hours or more worked per week, or a lesser number of hours if: 1) the lesser number is a recognized industry standard as determined by the County, or 2) the Contractor has a long-standing practice that defines the lesser number of

hours as full-time. Full-time employees providing short-term, temporary services of ninety (90) days or less within a twelve (12) month period are not considered full-time for purposes of the Jury Service Program. If the Contractor uses any subcontractor to perform services for the County under the Contract, the subcontractor shall also be subject to the provisions of this paragraph. The provisions of this paragraph shall be inserted into any such subcontract agreement and a copy of the Jury Service Program shall be attached to the agreement.

- If the Contractor is not required to comply with the Jury 8.8.2.3 Service Program when the Contract commences, the Contractor shall have a continuing obligation to review the applicability of its "exception status" from the Jury Service Program, and the Contractor shall immediately notify the County if the Contractor at any time either comes within the Jury Service Program's definition of "Contractor" or if the Contractor no longer qualifies for an exception to the Jury Service Program. In either event, the Contractor shall immediately implement a written policy consistent with the Jury Service Program. The County may also require, at any time during the Contract and at its sole discretion, that the Contractor demonstrate, to the County's satisfaction that the Contractor either continues to remain outside of the Jury Service Program's definition of "Contractor" and/or that the Contractor continues to qualify for an exception to the Program.
- 8.8.2.4 The Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, the County may, in its sole discretion, terminate the Contract and/or bar the Contractor from the award of future County Contracts for a period of time consistent with the seriousness of the breach.

8.9 Conflict of Interest

8.9.1 No County employee whose position with the County enables such employee to influence the award of this Contract or any competing Contract, and no spouse or economic dependent of such employee, shall be employed in any capacity by the Contractor or have any other direct or indirect financial interest in this Contract. No officer or employee of the Contractor who may financially

benefit from the performance of work hereunder shall in any way participate in the County's approval, or ongoing evaluation, of such work, or in any way attempt to unlawfully influence the County's approval or ongoing evaluation of such work.

8.9.2 The Contractor shall comply with all conflict of interest laws, ordinances, and regulations now in effect or hereafter to be enacted during the term of this Contract. The Contractor warrants that it is not now aware of any facts that create a conflict of interest. If the Contractor hereafter becomes aware of any facts that might reasonably be expected to create a conflict of interest, it shall immediately make full written disclosure of such facts to the County. Full written disclosure shall include, but is not limited to, identification of all persons implicated and a complete description of all relevant circumstances. Failure to comply with the provisions of this paragraph shall be a material breach of this Contract.

8.10 Consideration of Hiring County Employees Targeted for Layoff or Re-Employment List

Should the Contractor require additional or replacement personnel after the effective date of this Contract to perform the services set forth herein, the Contractor shall give first consideration for such employment openings to qualified, permanent County employees who are targeted for layoff or qualified, former County employees who are on a re-employment list during the life of this Contract.

8.11 Consideration of Hiring GAIN-START Participants

8.11.1 Should the Contractor require additional or replacement personnel after the effective date of this Contract, the Contractor shall give consideration for any such employment openings to participants in the County's Department of Public Social Services Greater Avenues for Independence (GAIN) Program or Skills and Training to Achieve Readiness for Tomorrow (START) Program who meet the Contractor's minimum qualifications for the open position. For this purpose, consideration shall mean that the Contractor will interview qualified candidates. The County will refer GAIN-START participants by job category to the Contractor. The Contractors shall job openings iob requirements report all with to: GAINSTART@DPSS.LACOUNTY.GOV and BSERVICES@OPPORTUNITY.LACOUNTY.GOV and DPSS will refer qualified GAIN-START iob candidates.

8.11.2 In the event that both laid-off County employees and GAIN-START participants are available for hiring, County employees shall be given first priority.

8.12 Contractor Responsibility and Debarment

8.12.1 **Responsible Contractor**

A responsible Contractor is a Contractor who has demonstrated the attribute of trustworthiness, as well as quality, fitness, capacity, and experience to satisfactorily perform the Contract. It is the County's policy to conduct business only with responsible Contractors.

8.12.2 Chapter 2.202 of the County Code

The Contractor is hereby notified that, in accordance with Chapter 2.202 of the County Code, if the County acquires information concerning the performance of the Contractor on this or other Contracts which indicates that the Contractor is not responsible, the County may, in addition to other remedies provided in the Contract, debar the Contractor from bidding or proposing on, or being awarded, and/or performing work on County Contracts for a specified period of time, which generally will not exceed five (5) years but may exceed five (5) years or be permanent if warranted by the circumstances, and terminate any or all existing Contracts the Contractor may have with the County.

8.12.3 Non-responsible Contractor

The County may debar a Contractor if the Board of Supervisors finds, in its discretion, that the Contractor has done any of the following: 1) violated a term of a Contract with the County or a nonprofit corporation created by the County, 2) committed an act or omission which negatively reflects on the Contractor's quality, fitness or capacity to perform a Contract with the County, any other public entity, or a nonprofit corporation created by the County, or engaged in a pattern or practice which negatively reflects on same, 3) committed an act or offense which indicates a lack of business integrity or business honesty, or 4) made or submitted a false claim against the County or any other public entity.

8.12.4 Contractor Hearing Board

- 8.12.4.1 If there is evidence that the Contractor may be subject to debarment, the Department will notify the Contractor in writing of the evidence which is the basis for the proposed debarment and will advise the Contractor of the scheduled date for a debarment hearing before the Contractor Hearing Board.
- 8.12.4.2 The Contractor Hearing Board will conduct a hearing where evidence on the proposed debarment is The Contractor and/or the Contractor's presented. representative shall be given an opportunity to submit evidence at that hearing. After the hearing, the Contractor Hearing Board shall prepare a tentative decision, which shall proposed contain recommendation regarding whether the Contractor should be debarred, and, if so, the appropriate length of time of the debarment. The Contractor and the Department shall be provided an opportunity to object to the tentative proposed decision prior to its presentation to the Board of Supervisors.
- 8.12.4.3 After consideration of any objections, or if no objections are submitted, a record of the hearing, the proposed decision, and any other recommendation of the Contractor Hearing Board shall be presented to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.
- 8.12.4.4 If a Contractor has been debarred for a period longer than five (5) years, that Contractor may after the debarment has been in effect for at least five (5) years, submit a written request for review of the debarment determination to reduce the period of debarment or terminate the debarment. The County may, in its discretion, reduce the period of debarment or terminate the debarment if it finds that the Contractor has adequately demonstrated one or more of the following:

 1) elimination of the grounds for which the debarment was imposed; 2) a bona fide change in ownership or management; 3) material evidence discovered after debarment was imposed; or 4) any other reason that is in the best interests of the County.

- 8.12.4.5 The Contractor Hearing Board will consider a request for review of a debarment determination only where 1) the Contractor has been debarred for a period longer than five (5) years; 2) the debarment has been in effect for at least five (5) years; and 3) the request is in writing, states one or more of the grounds for reduction of the debarment period or termination of the debarment, and includes supporting documentation. Upon receiving an appropriate request, the Contractor Hearing Board will provide notice of the hearing on the request. At the hearing, the Contractor Hearing Board shall conduct a hearing where evidence on the proposed reduction of debarment period or termination of debarment is presented. This hearing shall be conducted and the request for review decided by the Contractor Hearing Board pursuant to the same procedures as for a debarment hearing.
- 8.12.4.6 The Contractor Hearing Board's proposed decision shall contain a recommendation on the request to reduce the period of debarment or terminate the debarment. The Contractor Hearing Board shall present its proposed decision and recommendation to the Board of Supervisors. The Board of Supervisors shall have the right to modify, deny, or adopt the proposed decision and recommendation of the Contractor Hearing Board.

8.12.5 **Subcontractors of Contractor**

These terms shall also apply to subcontractors of the County Contractors.

8.13 Contractor's Acknowledgement of County's Commitment to Safely Surrendered Baby Law

The Contractor acknowledges that the County places a high priority on the implementation of the Safely Surrendered Baby Law. The Contractor understands that it is the County's policy to encourage all County Contractors to voluntarily post the County's "Safely Surrendered Baby Law" poster, in Exhibit I, in a prominent position at the Contractor's place of business. The Contractor will also encourage its subcontractors, if any, to post this poster in a prominent position in the subcontractor's place of business. Information and posters for printing are available at https://lacounty.gov/residents/family-services/child-safety/safe-surrender/.

8.14 Contractor's Warranty of Adherence to County's Child Support Compliance Program

- 8.14.1 The Contractor acknowledges that the County has established a goal of ensuring that all individuals who benefit financially from the County through Contracts are in compliance with their court-ordered child, family and spousal support obligations in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.
- 8.14.2 As required by the County's Child Support Compliance Program (County Code Chapter 2.200) and without limiting the Contractor's duty under this Contract to comply with all applicable provisions of law, the Contractor warrants that it is now in compliance and shall during the term of this Contract maintain in compliance with employment and wage reporting requirements as required by the Federal Social Security Act (42 USC Section 653a) and California Unemployment Insurance Code Section 1088.5, and shall implement all lawfully served Wage and Earnings Withholding Orders or Child Support Services Department Notices of Wage and Earnings Assignment for Child, Family or Spousal Support, pursuant to Code of Civil Procedure Section 706.031 and Family Code Section 5246(b).

8.15 County's Quality Assurance Plan

The County or its agent(s) will monitor the Contractor's performance under this Contract on not less than an annual basis. Such monitoring will include assessing the Contractor's compliance with all Contract terms and conditions and performance standards. The Contractor deficiencies which the County determines are significant or continuing and that may place performance of the Contract in jeopardy if not corrected will be reported to the Board of Supervisors and listed in the appropriate Contractor performance database. The report to the Board will include improvement/corrective action measures taken by the County and the Contractor. If improvement does not occur consistent with the corrective action measures, the County may terminate this Contract or impose other penalties as specified in this Contract.

8.16 Damage to County Facilities, Buildings or Grounds

8.16.1 The Contractor shall repair, or cause to be repaired, at its own cost, any and all damage to County facilities, buildings, or grounds caused by the Contractor or employees or agents of the Contractor. Such repairs shall be made immediately after the

- Contractor has become aware of such damage, but in no event later than thirty (30) days after the occurrence.
- 8.16.2 If the Contractor fails to make timely repairs, the County may make any necessary repairs. All costs incurred by the County, as determined by the County, for such repairs shall be repaid by the Contractor by cash payment upon demand.

8.17 Employment Eligibility Verification

- 8.17.1 The Contractor warrants that it fully complies with all Federal and State statutes and regulations regarding the employment of aliens and others and that all its employees performing work under this Contract meet the citizenship or alien status requirements set forth in Federal and State statutes and regulations. Contractor shall obtain from all employees performing work hereunder. all verification and other documentation of employment eligibility status required by Federal and State statutes and regulations including, but not limited to, the Immigration Reform and Control Act of 1986, (P.L. 99-603), or as they currently exist and as they may be hereafter amended. The Contractor shall retain all such documentation for all covered employees for the period prescribed by law.
- 8.17.2 The Contractor shall indemnify, defend, and hold harmless, the County, its agents, officers, and employees from employer sanctions and any other liability which may be assessed against the Contractor or the County or both in connection with any alleged violation of any Federal or State statutes or regulations pertaining to the eligibility for employment of any persons performing work under this Contract.

8.18 Counterparts and Electronic Signatures and Representations

This Contract may be executed in two or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same Contract. The facsimile, email or electronic signature of the Parties shall be deemed to constitute original signatures, and facsimile or electronic copies hereof shall be deemed to constitute duplicate originals.

The County and the Contractor hereby agree to regard electronic representations of original signatures of authorized officers of each party, when appearing in appropriate places on the Amendments prepared pursuant to Paragraph 8.1 (Amendments) and received via communications facilities (facsimile, email or electronic signature),

as legally sufficient evidence that such legally binding signatures have been affixed to Amendments to this Contract.

8.19 Fair Labor Standards

The Contractor shall comply with all applicable provisions of the Federal Fair Labor Standards Act and shall indemnify, defend, and hold harmless the County and its agents, officers, and employees from any and all liability, including, but not limited to, wages, overtime pay, liquidated damages, penalties, court costs, and attorneys' fees arising under any wage and hour law, including, but not limited to, the Federal Fair Labor Standards Act, for work performed by the Contractor's employees for which the County may be found jointly or solely liable.

8.20 Force Majeure

- 8.20.1 Neither party shall be liable for such party's failure to perform its obligations under and in accordance with this Contract, if such failure arises out of fires, floods, epidemics, quarantine restrictions, other natural occurrences, strikes, lockouts (other than a lockout by such party or any of such party's subcontractors), freight embargoes, or other similar events to those described above, but in every such case the failure to perform must be totally beyond the control and without any fault or negligence of such party (such events are referred to in this paragraph as "force majeure events").
- 8.20.2 Notwithstanding the foregoing, a default by a subcontractor of the Contractor shall not constitute a force majeure event, unless such default arises out of causes beyond the control of both the Contractor and such subcontractor, and without any fault or negligence of either of them. In such case, the Contractor shall not be liable for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the Contractor to meet the required performance schedule. As used in this subparagraph. and "subcontractors" the term "subcontractor" mean subcontractors at any tier.
- 8.20.3 In the event the Contractor's failure to perform arises out of a force majeure event, the Contractor agrees to use commercially reasonable best efforts to obtain goods or services from other sources, if applicable, and to otherwise mitigate the damages and reduce the delay caused by such force majeure event.

8.21 Governing Law, Jurisdiction, and Venue

This Contract shall be governed by, and construed in accordance with, the laws of the State of California. The Contractor agrees and consents to the exclusive jurisdiction of the courts of the State of California for all purposes regarding this Contract and further agrees and consents that venue of any action brought hereunder shall be exclusively in the County of Los Angeles.

8.22 Independent Contractor Status

- 8.22.1 This Contract is by and between the County and the Contractor and is not intended, and shall not be construed, to create the relationship of agent, servant, employee, partnership, joint venture, or association, as between the County and the Contractor. The employees and agents of one party shall not be, or be construed to be, the employees or agents of the other party for any purpose whatsoever.
- 8.22.2 The Contractor shall be solely liable and responsible for providing to, or on behalf of, all persons performing work pursuant to this Contract all compensation and benefits. The County shall have no liability or responsibility for the payment of any salaries, wages, unemployment benefits, disability benefits, Federal, State, or local taxes, or other compensation, benefits, or taxes for any personnel provided by or on behalf of the Contractor.
- 8.22.3 The Contractor understands and agrees that all persons performing work pursuant to this Contract are, for purposes of Workers' Compensation liability, solely employees of the Contractor and not employees of the County. The Contractor shall be solely liable and responsible for furnishing any and all Workers' Compensation benefits to any person as a result of any injuries arising from or connected with any work performed by or on behalf of the Contractor pursuant to this Contract.
- 8.22.4 The Contractor shall adhere to the provisions stated in Paragraph 7.6 (Confidentiality).

8.23 Indemnification

The Contractor shall indemnify, defend and hold harmless the County, its Special Districts, elected and appointed officers, employees, agents and volunteers (County Indemnitees) from and against any and all liability, including but not limited to demands, claims, actions, fees, costs and expenses (including attorney and expert witness fees), arising from and/or

relating to this Contract, except for such loss or damage arising from the sole negligence or willful misconduct of the County indemnitees.

8.24 General Provisions for all Insurance Coverage

8.24.1 Without limiting the Contractor's indemnification of the County, and in the performance of this Contract and until all of its obligations pursuant to this Contract have been met, the Contractor shall provide and maintain at its own expense insurance coverage satisfying the requirements specified in Paragraphs 8.24 (General Provisions for all Insurance Coverage) and 8.25 (Insurance Coverage) of this Contract. These minimum insurance coverage terms, types and limits (the "Required Insurance") also are in addition to and separate from any other contractual obligation imposed upon the Contractor pursuant to this Contract. The County in no way warrants that the Required Insurance is sufficient to protect the Contractor for liabilities which may arise from or relate to this Contract.

8.24.2 Evidence of Coverage and Notice to County

- 8.24.2.1 Certificate(s) of insurance coverage (Certificate) satisfactory to the County, and a copy of an Additional Insured endorsement confirming the County and its Agents (defined below) has been given Insured status under the Contractor's General Liability policy, shall be delivered to the County at the address shown below and provided prior to commencing services under this Contract.
- 8.24.2.2 Renewal Certificates shall be provided to the County not less than ten (10) days prior to the Contractor's policy expiration dates. The County reserves the right to obtain complete, certified copies of any required Contractor and/or subcontractor insurance policies at any time.
- 8.24.2.3 Certificates shall identify all Required Insurance coverage types and limits specified herein, reference this Contract by name or number, and be signed by an authorized representative of the insurer(s). The Insured party named on the Certificate shall match the name of the Contractor identified as the contracting party in this Contract. Certificates shall provide the full name of each insurer providing coverage, its NAIC (National Association of Insurance Commissioners) identification

number, its financial rating, the amounts of any policy deductibles or self-insured retentions exceeding fifty thousand dollars (\$50,000), and list any County required endorsement forms.

- 8.24.2.4 Neither the County's failure to obtain, nor the County's receipt of, or failure to object to a non-complying insurance certificate or endorsement, or any other insurance documentation or information provided by the Contractor, its insurance broker(s) and/or insurer(s), shall be construed as a waiver of any of the Required Insurance provisions.
- 8.24.2.5 Certificates and copies of any required endorsements shall be sent to:

Oscar Rivas, Contract Analyst
County of Los Angeles Probation Department
Contracts & Grants Management Division
9150 East Imperial Highway, Room D-29
Downey, CA 90242

E-mail address: Oscar.Rivas@probation.lacounty.gov Fax #: (562) 658-2307

8.24.2.6 The Contractor also shall promptly report to the County any injury or property damage accident or incident, including any injury to a Contractor employee occurring on County property, and any loss, disappearance, destruction, misuse, or theft of County property, monies or securities entrusted to the Contractor. The Contractor also shall promptly notify the County of any third party claim or suit filed against the Contractor or any of its subcontractors which arises from or relates to this Contract, and could result in the filing of a claim or lawsuit against the Contractor and/or the County.

8.24.3 Additional Insured Status and Scope of Coverage

The County of Los Angeles, its Special Districts, Elected Officials, Officers, Agents, employees and volunteers (collectively County and its Agents) shall be provided additional insured status under the Contractor's General Liability policy with respect to liability arising out of the Contractor's ongoing and completed operations performed on behalf of the County. The County and its Agents additional insured status shall apply with respect to liability and defense of suits arising out of the Contractor's acts or omissions,

whether such liability is attributable to the Contractor or to the County. The full policy limits and scope of protection also shall apply to the County and its Agents as an additional insured, even if they exceed the County's minimum Required Insurance specifications herein. Use of an automatic additional insured endorsement form is acceptable providing it satisfies the Required Insurance provisions herein.

8.24.4 Cancellation of or Changes in Insurance

The Contractor shall provide the County with, or the Contractor's insurance policies shall contain a provision that the County shall receive, written notice of cancellation or any change in Required Insurance, including insurer, limits of coverage, term of coverage or policy period. The written notice shall be provided to the County at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. Failure to provide written notice of cancellation or any change in Required Insurance may constitute a material breach of the Countract, in the sole discretion of the County, upon which the County may suspend or terminate this Contract.

8.24.5 Failure to Maintain Insurance

The Contractor's failure to maintain or to provide acceptable evidence that it maintains the Required Insurance shall constitute a material breach of the Contract, upon which the County immediately may withhold payments due to the Contractor, and/or suspend or terminate this Contract. The County, at its sole discretion, may obtain damages from the Contractor resulting from said breach. Alternatively, the County may purchase the Required Insurance, and without further notice to the Contractor, deduct the premium cost from sums due to the Contractor or pursue Contractor reimbursement.

8.24.6 Insurer Financial Ratings

Coverage shall be placed with insurers acceptable to the County with A.M. Best ratings of not less than A:VII unless otherwise approved by the County.

8.24.7 Contractor's Insurance Shall Be Primary

The Contractor's insurance policies, with respect to any claims related to this Contract, shall be primary with respect to all other sources of coverage available to the Contractor. Any County

maintained insurance or self-insurance coverage shall be in excess of and not contribute to any Contractor coverage.

8.24.8 Waivers of Subrogation

To the fullest extent permitted by law, the Contractor hereby waives its rights and its insurer(s)' rights of recovery against the County under all the Required Insurance for any loss arising from or relating to this Contract. The Contractor shall require its insurers to execute any waiver of subrogation endorsements which may be necessary to effect such waiver.

8.24.9 Subcontractor Insurance Coverage Requirements

The Contractor shall include all subcontractors as insureds under the Contractor's own policies, or shall provide the County with each subcontractor's separate evidence of insurance coverage. The Contractor shall be responsible for verifying each subcontractor complies with the Required Insurance provisions herein, and shall require that each subcontractor name the County and the Contractor as additional insureds on the subcontractor's General Liability policy. The Contractor shall obtain the County's prior review and approval of any subcontractor request for modification of the Required Insurance.

8.24.10 Deductibles and Self-Insured Retentions (SIRs)

The Contractor's policies shall not obligate the County to pay any portion of any Contractor deductible or SIR. The County retains the right to require the Contractor to reduce or eliminate policy deductibles and SIRs as respects the County, or to provide a bond guaranteeing the Contractor's payment of all deductibles and SIRs, including all related claims investigation, administration and defense expenses. Such bond shall be executed by a corporate surety licensed to transact business in the State of California.

8.24.11 Claims Made Coverage

If any part of the Required Insurance is written on a claims made basis, any policy retroactive date shall precede the effective date of this Contract. The Contractor understands and agrees it shall maintain such coverage for a period of not less than three (3) years following Contract expiration, termination or cancellation.

8.24.12 Application of Excess Liability Coverage

The Contractors may use a combination of primary, and excess insurance policies which provide coverage as broad as ("follow form" over) the underlying primary policies, to satisfy the Required Insurance provisions.

8.24.13 Separation of Insureds

All liability policies shall provide cross-liability coverage as would be afforded by the standard ISO (Insurance Services Office, Inc.) separation of insureds provision with no insured versus insured exclusions or limitations.

8.24.14 Alternative Risk Financing Programs

The County reserves the right to review, and then approve, Contractor use of self-insurance, risk retention groups, risk purchasing groups, pooling arrangements and captive insurance to satisfy the Required Insurance provisions. The County and its Agents shall be designated as an Additional Covered Party under any approved program.

8.24.15 County Review and Approval of Insurance Requirements

The County reserves the right to review and adjust the Required Insurance provisions, conditioned upon the County's determination of changes in risk exposures.

8.25 Insurance Coverage

8.25.1 Commercial General Liability insurance (providing scope of coverage equivalent to ISO policy form CG 00 01), naming the County and its Agents as an additional insured, with limits of not less than:

General Aggregate: \$2 million

Products/Completed Operations Aggregate: \$1 million

Personal and Advertising Injury: \$1 million

Each Occurrence: \$1 million

There can be no exclusion for the use of animals in the performance of this Contract.

- **8.25.2** Automobile Liability insurance (providing scope of coverage equivalent to ISO policy form CA 00 01) with limits of not less than \$1 million for bodily injury and property damage, in combined or equivalent split limits, for each single accident. Insurance shall cover liability arising out of the Contractor's use of autos pursuant to this Contract, including owned, leased, hired, and/or nonowned autos, as each may be applicable.
- Workers Compensation and Employers' Liability insurance or 8.25.3 qualified self-insurance satisfying statutory requirements, which includes Employers' Liability coverage with limits of not less than \$1 million per accident. If the Contractor will provide leased employees, or, is an employee leasing or temporary staffing firm or a professional employer organization (PEO), coverage also shall include an Alternate Employer Endorsement (providing scope of coverage equivalent to ISO policy form WC 00 03 01 A) naming the County as the Alternate Employer. The written notice shall be provided to the County at least ten (10) days in advance of cancellation for non-payment of premium and thirty (30) days in advance for any other cancellation or policy change. If applicable to the Contractor's operations, coverage also shall be arranged to satisfy the requirements of any federal workers or workmen's compensation law or any federal occupational disease law.

8.25.4 Unique Insurance Coverage

8.25.4.3 Sexual Misconduct Liability

Insurance covering actual or alleged claims for sexual misconduct and/or molestation with limits of not less than \$1 million per claim and \$1 million aggregate, and claims for negligent employment, investigation, supervision, training or retention of, or failure to report to proper authorities, a person(s) who committed any act of abuse, molestation, harassment, mistreatment or maltreatment of a sexual nature.

8.25.4.2 Intentionally Omitted

8.25.4.3 Property Coverage

Contractors given exclusive use of County owned or leased property must carry property coverage at least as broad as that provided by the ISO special causes of loss (ISO policy form CP 10 30) form. The County and its Agents must be named as an Additional Insured and

Loss Payee on Contractor's insurance as its interests may appear. Automobiles and mobile equipment must be insured for their actual cash value. Real property and all other personal property must be insured for their full replacement value.

8.25.4.4 Intentionally Omitted

8.25.4.3 Technology Errors & Omissions Insurance

Insurance for liabilities arising from errors, omissions, or negligent acts in rendering or failing to render computer or information technology services and technology products. Coverage for violation of software copyright should be included. Technology services should at a minimum include (1) systems analysis; (2) systems programming; (3) data processing; (4) systems integration; (5) outsourcing including outsourcing development and design; systems (6) consulting, development and modification; (7) training services relating to computer software or hardware; (8) management, repair and maintenance of computer products, networks and systems; (9) marketing, selling, distributing, servicing, installing and maintaining computer hardware or software; (10) data entry, modification. verification. maintenance. storage, retrieval or preparation of data output, and any other services provided by the vendor with limits of not less than \$10 million.

8.25.4.3 Cyber Liability Insurance

The Contractor must secure and maintain cyber liability insurance coverage with limits of \$ 2,000,000 Million per occurrence and in the aggregate during the term of the Contract, including coverage for: network security liability; privacy liability; privacy regulatory proceeding, defense, response, expenses and fines; technology professional liability (errors and omissions); privacy breach expense reimbursement (liability arising from the loss or disclosure of County Information no matter how it occurs); system breach; denial or loss of service; introduction, implantation, or spread of malicious software code; unauthorized access to or use of computer systems; and Data/Information loss and business interruption; any other liability or risk that

arises out of the Contract. The Contractor must add the County as an additional insured to its cyber liability insurance policy and provide to the County certificates of insurance evidencing the foregoing upon the County's request. The procuring of the insurance described herein, or delivery of the certificates of insurance described herein, will not be construed as a limitation upon the Contractor's liability or as full performance of its indemnification obligations hereunder. No exclusion/restriction for unencrypted portable devices/media may be on the policy.

8.25.4.3 Intentionally Omitted

8.26 Liquidated Damages

- 8.26.1 If, in the judgment of the Chief Probation Officer, or his/her designee, the Contractor is deemed to be non-compliant with the terms and obligations assumed hereby, the Chief Probation Officer, or his/her designee, at his/her option, in addition to, or in lieu of, other remedies provided herein, may withhold the entire monthly payment or deduct pro rata from the Contractor's invoice for work not performed. A description of the work not performed and the amount to be withheld or deducted from payments to the Contractor from the County, will be forwarded to the Contractor by the Chief Probation Officer, or his/her designee, in a written notice describing the reasons for said action.
- 8.26.2 If the Chief Probation Officer, or his/her designee, determines that there are deficiencies in the performance of this Contract that the Chief Probation Officer, or his/her designee, deems are correctable by the Contractor over a certain time span, the Chief Probation Officer, or his/her designee, will provide a written notice to the Contractor to correct the deficiency within specified time frames. Should the Contractor fail to correct deficiencies within said time frame, the Chief Probation Officer, or his/her designee, may:
 - (a) Deduct from the Contractor's payment, pro rata, those applicable portions of the Monthly Contract Sum; and/or
 - (b) Deduct liquidated damages. The parties agree that it will be impracticable or extremely difficult to fix the extent of actual damages resulting from the failure of the Contractor to correct a deficiency within the specified time frame. The parties hereby agree that under the current circumstances a

reasonable estimate of such damages is one hundred dollars (\$100) per day per infraction, or as specified in the Exhibit U (Performance Requirements Summary (PRS) Chart) hereunder, and that the Contractor shall be liable to the County for liquidated damages in said amount. Said amount shall be deducted from the County's payment to the Contractor; and/or

- (c) Upon giving five (5) days' notice to the Contractor for failure to correct the deficiencies, the County may correct any and all deficiencies and the total costs incurred by the County for completion of the work by an alternate source, whether it be County forces or separate private Contractor, will be deducted and forfeited from the payment to the Contractor from the County, as determined by the County.
- 8.26.3 The action noted in Subparagraph 8.26.2 shall not be construed as a penalty, but as adjustment of payment to the Contractor to recover the County cost due to the failure of the Contractor to complete or comply with the provisions of this Contract.
- 8.26.4 This paragraph shall not, in any manner, restrict or limit the County's right to damages for any breach of this Contract provided by law or as specified in the PRS or Subparagraph 8.26.2, and shall not, in any manner, restrict or limit the County's right to terminate this Contract as agreed to herein.

8.27 Most Favored Public Entity

If the Contractor's prices decline or should the Contractor at any time during the term of this Contract provide the same goods or services under similar quantity and delivery conditions to the State of California or any county, municipality, or district of the State at prices below those set forth in this Contract, then such lower prices shall be immediately extended to the County.

8.28 Nondiscrimination and Affirmative Action

8.28.1 The Contractor certifies and agrees that all persons employed by it, its affiliates, subsidiaries, or holding companies are and shall be treated equally without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations.

- 8.28.2 The Contractor shall certify to, and comply with, the provisions of Exhibit D (Contractor's EEO Certification).
- 8.28.3 The Contractor shall take affirmative action to ensure that applicants are employed, and that employees are treated during employment, without regard to race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, in compliance with all applicable Federal and State anti-discrimination laws and regulations. Such action shall include, but is not limited to: employment, upgrading, demotion, transfer, recruitment or recruitment advertising, layoff or termination, rates of pay or other forms of compensation, and selection for training, including apprenticeship.
- 8.28.4 The Contractor certifies and agrees that it will deal with its subcontractors, bidders, or vendors without regard to or because of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation.
- 8.28.5 The Contractor certifies and agrees that it, its affiliates, subsidiaries, or holding companies shall comply with all applicable Federal and State laws and regulations to the end that no person shall, on the grounds of race, color, religion, ancestry, national origin, sex, age, physical or mental disability, marital status, or political affiliation, be excluded from participation in, be denied the benefits of, or be otherwise subjected to discrimination under this Contract or under any project, program, or activity supported by this Contract.
- 8.28.6 The Contractor shall allow County representatives access to the Contractor's employment records during regular business hours to verify compliance with the provisions of this Paragraph 8.28 (Nondiscrimination and Affirmative Action) when so requested by the County.
- 8.28.7 If the County finds that any provisions of this Paragraph 8.28 (Nondiscrimination and Affirmative Action) have been violated, such violation shall constitute a material breach of this Contract upon which the County may terminate or suspend this Contract. While the County reserves the right to determine independently that the anti-discrimination provisions of this Contract have been violated, in addition, a determination by the California Fair Employment and Housing Commission or the Federal Equal Employment Opportunity Commission that the Contractor has violated Federal or State anti-discrimination laws or regulations

shall constitute a finding by the County that the Contractor has violated the anti-discrimination provisions of this Contract.

8.28.8 The parties agree that in the event the Contractor violates any of the anti-discrimination provisions of this Contract, the County shall, at its sole option, be entitled to the sum of five hundred dollars (\$500) for each such violation pursuant to California Civil Code Section 1671 as liquidated damages in lieu of terminating or suspending this Contract.

8.29 Non Exclusivity

Nothing herein is intended nor shall be construed as creating any exclusive arrangement with the Contractor. This Contract shall not restrict County from acquiring similar, equal or like goods and/or services from other entities or sources.

8.30 Notice of Delays

Except as otherwise provided under this Contract, when either party has knowledge that any actual or potential situation is delaying or threatens to delay the timely performance of this Contract, that party shall, within one (1) business day, give notice thereof, including all relevant information with respect thereto, to the other party.

8.31 Notice of Disputes

The Contractor shall bring to the attention of the County's Program Manager and/or the County's Contract Manager any dispute between the County and the Contractor regarding the performance of services as stated in this Contract. If the County's Program Manager and/or the County's Contract Manager is not able to resolve the dispute, the Chief Probation Officer, or designee shall resolve it.

8.32 Notice to Employees Regarding the Federal Earned Income Credit

The Contractor shall notify its employees, and shall require each subcontractor to notify its employees, that they may be eligible for the Federal Earned Income Credit under the federal income tax laws. Such notice shall be provided in accordance with the requirements set forth in Internal Revenue Service Notice No. 1015.

8.33 Notice to Employees Regarding the Safely Surrendered Baby Law

The Contractor shall notify and provide to its employees, and shall require each subcontractor to notify and provide to its employees, information

regarding the Safely Surrendered Baby Law, its implementation in Los Angeles County, and where and how to safely surrender a baby. The information is set forth in Exhibit I (Safely Surrendered Baby Law) of this Contract. Additional information is available at https://lacounty.gov/residents/family-services/child-safety/safe-surrender/.

8.34 Notices

All notices or demands required or permitted to be given or made under this Contract shall be in writing and shall be hand delivered with signed receipt or mailed by first-class registered or certified mail, postage prepaid, addressed to the parties as identified in Exhibits E (County's Administration) and F (Contractor's Administration). Addresses may be changed by either party giving ten (10) days prior written notice thereof to the other party. The Chief Probation Officer or his/her designee shall have the authority to issue all notices or demands required or permitted by the County under this Contract.

8.35 Prohibition Against Inducement or Persuasion

Notwithstanding the above, the Contractor and the County agree that, during the term of this Contract and for a period of one year thereafter, neither party shall in any way intentionally induce or persuade any employee of one party to become an employee or agent of the other party. No bar exists against any hiring action initiated through a public announcement.

8.36 Public Records Act

8.36.1 Any documents submitted by the Contractor; all information obtained in connection with the County's right to audit and inspect the Contractor's documents, books, and accounting records pursuant to Paragraph 8.38 (Record Retention and Inspection-Audit Settlement) of this Contract; as well as those documents which were required to be submitted in response to the Request for Proposals (RFP) used in the solicitation process for this Contract. become the exclusive property of the County. All such documents become a matter of public record and shall be regarded as public Exceptions will be those elements in the California Government Code Section 7921 et seg. (Public Records Act) and which are marked "trade secret", "confidential", or "proprietary". The County shall not in any way be liable or responsible for the disclosure of any such records including, without limitation, those so marked, if disclosure is required by law, or by an order issued by a court of competent jurisdiction.

8.36.2 In the event the County is required to defend an action on a Public Records Act request for any of the aforementioned documents, information, books, records, and/or contents of a proposal marked "trade secret", "confidential", or "proprietary", the Contractor agrees to defend and indemnify the County from all costs and expenses, including reasonable attorney's fees, in action or liability arising under the Public Records Act.

8.37 Publicity

- 8.37.1 The Contractor shall not disclose any details in connection with this Contract to any person or entity except as may be otherwise provided hereunder or required by law. However, in recognizing the Contractor's need to identify its services and related clients to sustain itself, the County shall not inhibit the Contractor from publishing its role under this Contract within the following conditions:
 - 8.37.1.1 The Contractor shall develop all publicity material in a professional manner; and
 - 8.37.1.2 During the term of this Contract, the Contractor shall not, and shall not authorize another to, publish or disseminate any commercial advertisements, press releases, feature articles, or other materials using the name of the County without the prior written consent of the County's Program Manager. The County shall not unreasonably withhold written consent.
- 8.37.2 The Contractor may, without the prior written consent of the County, indicate in its proposals and sales materials that it has been awarded this Contract with the County of Los Angeles, provided that the requirements of this Paragraph 8.37 (Publicity) shall apply.

8.38 Record Retention and Inspection-Audit Settlement

8.38.1 The Contractor shall maintain accurate and complete financial records of its activities and operations relating to this Contract in accordance with generally accepted accounting principles. The Contractor shall also maintain accurate and complete employment and other records relating to its performance of this Contract. The Contractor agrees that the County, or its authorized representatives, shall have access to and the right to examine, audit, excerpt, copy, or transcribe any pertinent transaction, activity, or record relating to this Contract. All such material,

including, but not limited to, all financial records, bank statements, cancelled checks or other proof of payment, timecards, sign-in/sign-out sheets and other time and employment records, and proprietary data and information, shall be kept and maintained by the Contractor and shall be made available to the County during the term of this Contract and for a period of five (5) years thereafter unless the County's written permission is given to dispose of any such material prior to such time. All such material shall be maintained by the Contractor at a location in Los Angeles County, provided that if any such material is located outside Los Angeles County, then, at the County's option, the Contractor shall pay the County for travel, per diem, and other costs incurred by the County to examine, audit, excerpt, copy, or transcribe such material at such other location.

- 8.38.2 In the event that an audit of the Contractor is conducted specifically regarding this Contract by any Federal or State auditor, or by any auditor or accountant employed by the Contractor or otherwise, then the Contractor shall file a copy of such audit report with the County's Auditor-Controller within thirty (30) days of the Contractor's receipt thereof, unless otherwise provided by applicable Federal or State law or under this Contract. Subject to applicable law, the County shall make a reasonable effort to maintain the confidentiality of such audit report(s).
- 8.38.3 Failure on the part of the Contractor to comply with any of the provisions of this Paragraph 8.38 shall constitute a material breach of this Contract upon which the County may terminate or suspend this Contract.
- 8.38.4 If, at any time during the term of this Contract or within five (5) years after the expiration or termination of this Contract, representatives of the County conduct an audit of the Contractor regarding the work performed under this Contract, and if such audit finds that the County's dollar liability for any such work is less than payments made by the County to the Contractor, then the difference shall be either: a) repaid by the Contractor to the County by cash payment upon demand or b) at the sole option of the County's Auditor-Controller, deducted from any amounts due to the Contractor from the County, whether under this Contract or otherwise. If such audit finds that the County's dollar liability for such work is more than the payments made by the County to the Contractor, then the difference shall be paid to the Contractor by the County by cash payment, provided that in no event shall the County's maximum obligation for this Contract exceed the funds appropriated by the County for the purpose of this Contract.

8.38.5 Intentionally Omitted

8.39 Recycled Bond Paper

Consistent with the Board of Supervisors' policy to reduce the amount of solid waste deposited at the County landfills, the Contractor agrees to use recycled-content paper to the maximum extent possible on this Contract.

8.40 Subcontracting

- 8.40.1 The requirements of this Contract may not be subcontracted by the Contractor **without the advance approval of the County**. Any attempt by the Contractor to subcontract without the prior consent of the County may be deemed a material breach of this Contract.
- 8.40.2 If the Contractor desires to subcontract, the Contractor shall provide the following information promptly at the County's request:
 - 8.40.2.1 A description of the work to be performed by the subcontractor;
 - 8.40.2.2 A draft copy of the proposed subcontract; and
 - 8.40.2.3 Other pertinent information and/or certifications requested by the County.
- 8.40.3 The Contractor shall indemnify, defend, and hold the County harmless with respect to the activities of each and every subcontractor in the same manner and to the same degree as if such subcontractor(s) were the Contractor employees.
- 8.40.4 The Contractor shall remain fully responsible for all performances required of it under this Contract, including those that the Contractor has determined to subcontract, notwithstanding the County's approval of the Contractor's proposed subcontract.
- 8.40.5 The County's consent to subcontract shall not waive the County's right to prior and continuing approval of any and all personnel, including subcontractor employees, providing services under this Contract. The Contractor is responsible to notify its subcontractors of this County right.
- 8.40.6 The County's Contract Manager is authorized to act for and on behalf of the County with respect to approval of any subcontract and subcontractor employees. After approval of the subcontract by

the County, the Contractor shall forward a fully executed subcontract to the County for their files.

- 8.40.7 The Contractor shall be solely liable and responsible for all payments or other compensation to all subcontractors and their officers, employees, agents, and successors in interest arising through services performed hereunder, notwithstanding the County's consent to subcontract.
- 8.40.8 The Contractor shall obtain certificates of insurance, which establish that the subcontractor maintains all the programs of insurance required by the County from each approved subcontractor. Before any subcontractor employee may perform any work hereunder, the Contractor shall ensure delivery of all such documents to:

Oscar Rivas, Contract Analyst
County of Los Angeles Probation Department
Contracts & Grants Management Division
9150 East Imperial Highway, Room D-29
Downey, CA 90242

E-mail address: Oscar.Rivas@probation.lacounty.gov

Fax#: (562) 658-2307

8.41 Termination for Breach of Warranty to Maintain Compliance with County's Child Support Compliance Program

Failure of the Contractor to maintain compliance with the requirements set forth in Paragraph 8.14 (Contractor's Warranty of Adherence to County's Child Support Compliance Program) shall constitute default under this Contract. Without limiting the rights and remedies available to the County under any other provision of this Contract, failure of the Contractor to cure such default within ninety (90) calendar days of written notice shall be grounds upon which the County may terminate this Contract pursuant to Paragraph 8.43 (Termination for Default) and pursue debarment of the Contractor, pursuant to County Code Chapter 2.202.

8.42 Termination for Convenience

8.42.1 This Contract may be terminated, in whole or in part, from time to time, when such action is deemed by the County, in its sole discretion, to be in its best interest. Termination of work hereunder shall be effected by notice of termination to the Contractor specifying the extent to which performance of work is terminated and the date upon which such termination becomes effective. The

- date upon which such termination becomes effective shall be no less than ten (10) days after the notice is sent.
- 8.42.2 After receipt of a notice of termination and except as otherwise directed by the County, the Contractor shall:
 - 8.42.2.1 Stop work under this Contract on the date and to the extent specified in such notice, and
 - 8.42.2.2 Complete performance of such part of the work as shall not have been terminated by such notice.
- 8.42.3 All material including books, records, documents, or other evidence bearing on the costs and expenses of the Contractor under this Contract shall be maintained by the Contractor in accordance with Paragraph 8.38 (Record Retention and Inspection-Audit Settlement).

8.43 Termination for Default

- 8.43.1 The County may, by written notice to the Contractor, terminate the whole or any part of this Contract, if, in the judgment of the County's Contract Manager:
 - 8.43.1.1 The Contractor has materially breached this Contract; or
 - 8.43.1.2 The Contractor fails to timely provide and/or satisfactorily perform any task, deliverable, service, or other work required either under this Contract; or
 - 8.43.1.3 The Contractor fails to demonstrate a high probability of timely fulfillment of performance requirements under this Contract, or of any obligations of this Contract and in either case, fails to demonstrate convincing progress toward a cure within five (5) working days (or such longer period as the County may authorize in writing) after receipt of written notice from the County specifying such failure.
- 8.43.2 In the event that the County terminates this Contract in whole or in part as provided in Subparagraph 8.43.1, the County may procure, upon such terms and in such manner as the County may deem appropriate, goods and services similar to those so terminated. The Contractor shall be liable to the County for any and all excess costs incurred by the County, as determined by the County, for such similar goods and services. The Contractor shall continue the

- performance of this Contract to the extent not terminated under the provisions of this paragraph.
- 8.43.3 Except with respect to defaults of any subcontractor, the Contractor shall not be liable for any such excess costs of the type identified in Subparagraph 8.43.2 if its failure to perform this Contract arises out of causes beyond the control and without the fault or negligence of the Contractor. Such causes may include, but are not limited to: acts of God or of the public enemy, acts of the County in either its sovereign or contractual capacity, acts of Federal or State governments in their sovereign capacities, fires, floods, epidemics, quarantine restrictions, strikes, freight embargoes, and unusually severe weather: but in every case, the failure to perform must be beyond the control and without the fault or negligence of the Contractor. If the failure to perform is caused by the default of a subcontractor, and if such default arises out of causes beyond the control of both the Contractor and the subcontractor, and without the fault or negligence of either of them, the Contractor shall not be liable for any such excess costs for failure to perform, unless the goods or services to be furnished by the subcontractor were obtainable from other sources in sufficient time to permit the Contractor to meet the required performance schedule. As used in this paragraph, the term "subcontractor(s)" means subcontractor(s) at any tier.
- 8.43.4 If, after the County has given notice of termination under the provisions of Paragraph 8.43 (Termination for Default) it is determined by the County that the Contractor was not in default under the provisions of Paragraph 8.43 (Termination for Default) or that the default was excusable under the provisions of Subparagraph 8.43.3, the rights and obligations of the parties shall be the same as if the notice of termination had been issued pursuant to Paragraph 8.42 (Termination for Convenience).
- 8.43.5 The rights and remedies of the County provided in this Paragraph 8.43 (Termination for Default) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.44 Termination for Improper Consideration

8.44.1 The County may, by written notice to the Contractor, immediately terminate the right of the Contractor to proceed under this Contract if it is found that consideration, in any form, was offered or given by the Contractor, either directly or through an intermediary, to any County officer, employee, or agent with the intent of securing this

Contract or securing favorable treatment with respect to the award, amendment, or extension of this Contract or the making of any determinations with respect to the Contractor's performance pursuant to this Contract. In the event of such termination, the County shall be entitled to pursue the same remedies against the Contractor as it could pursue in the event of default by the Contractor.

- 8.44.2 The Contractor shall immediately report any attempt by a County officer or employee to solicit such improper consideration. The report shall be made either to the County manager charged with the supervision of the employee or to the Los Angeles County Fraud Hotline at (800) 544-6861 or https://fraud.lacounty.gov/.
- 8.44.3 Among other items, such improper consideration may take the form of cash, discounts, services, the provision of travel or entertainment, or tangible gifts.

8.45 Termination for Insolvency

- 8.45.1 The County may terminate this Contract forthwith in the event of the occurrence of any of the following:
 - 8.45.1.1 Insolvency of the Contractor. The Contractor shall be deemed to be insolvent if it has ceased to pay its debts for at least sixty (60) days in the ordinary course of business or cannot pay its debts as they become due, whether or not a petition has been filed under the Federal Bankruptcy Code and whether or not the Contractor is insolvent within the meaning of the Federal Bankruptcy Code;
 - 8.45.1.2 The filing of a voluntary or involuntary petition regarding the Contractor under the Federal Bankruptcy Code;
 - 8.45.1.3 The appointment of a Receiver or Trustee for the Contractor; or
 - 8.45.1.4 The execution by the Contractor of a general assignment for the benefit of creditors.
- 8.45.2 The rights and remedies of the County provided in this Paragraph 8.45 (Termination for Insolvency) shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.46 Termination for Non-Adherence of County Lobbyist Ordinance

The Contractor, and each County Lobbyist or the County Lobbying firm as defined in County Code Section 2.160.010 retained by the Contractor, shall fully comply with the County's Lobbyist Ordinance, County Code Chapter 2.160. Failure on the part of the Contractor or any County Lobbyist or the County Lobbying firm retained by the Contractor to fully comply with the County's Lobbyist Ordinance shall constitute a material breach of this Contract, upon which the County may in its sole discretion, immediately terminate or suspend this Contract.

8.47 Termination for Non-Appropriation of Funds

Notwithstanding any other provision of this Contract, the County shall not be obligated for the Contractor's performance hereunder or by any provision of this Contract during any of the County's future fiscal years unless and until the County's Board of Supervisors appropriates funds for this Contract in the County's Budget for each such future fiscal year. In the event that funds are not appropriated for this Contract, then this Contract shall terminate as of June 30 of the last fiscal year for which funds were appropriated. The County shall notify the Contractor in writing of any such non-allocation of funds at the earliest possible date.

8.48 Validity

If any provision of this Contract or the application thereof to any person or circumstance is held invalid, the remainder of this Contract and the application of such provision to other persons or circumstances shall not be affected thereby.

8.49 Waiver

No waiver by the County of any breach of any provision of this Contract shall constitute a waiver of any other breach or of such provision. Failure of the County to enforce at any time, or from time to time, any provision of this Contract shall not be construed as a waiver thereof. The rights and remedies set forth in this Paragraph 8.49 shall not be exclusive and are in addition to any other rights and remedies provided by law or under this Contract.

8.50 Warranty Against Contingent Fees

8.50.1 The Contractor warrants that no person or selling agency has been employed or retained to solicit or secure this Contract upon any Contract or understanding for a commission, percentage, brokerage, or contingent fee, excepting bona fide employees or

bona fide established commercial or selling agencies maintained by the Contractor for the purpose of securing business.

8.50.2 For breach of this warranty, the County shall have the right to terminate this Contract and, at its sole discretion, deduct from the Contract price or consideration, or otherwise recover, the full amount of such commission, percentage, brokerage, or contingent fee.

8.51 Warranty of Compliance with County's Defaulted Property Tax Reduction Program

The Contractor acknowledges that the County has established a goal of ensuring that all individuals and businesses that benefit financially from the County through Contract are current in paying their property tax obligations (secured and unsecured roll) in order to mitigate the economic burden otherwise imposed upon the County and its taxpayers.

Unless the Contractor qualifies for an exemption or exclusion, the Contractor warrants and certifies that to the best of its knowledge it is now in compliance, and during the term of this Contract will maintain compliance, with Los Angeles County Code Chapter 2.206.

8.52 Termination for Breach of Warranty to Maintain Compliance with County's Defaulted Property Tax Reduction Program

Failure of the Contractor to maintain compliance with the requirements set forth in Paragraph 8.51 (Warranty of Compliance with County's Defaulted Property Tax Reduction Program) shall constitute default under this Contract. Without limiting the rights and remedies available to the County under any other provision of this Contract, failure of the Contractor to cure such default within ten (10) days of notice shall be grounds upon which the County may terminate this Contract and/or pursue debarment of the Contractor, pursuant to County Code Chapter 2.206.

8.53 Time Off for Voting

The Contractor shall notify its employees, and shall require each subcontractor to notify and provide to its employees, information regarding the time off for voting law (Elections Code Section 14000). Not less than ten (10) days before every statewide election, every Contractor and subcontractors shall keep posted conspicuously at the place of work, if practicable, or elsewhere where it can be seen as employees come or go to their place of work, a notice setting forth the provisions of Section 14000.

8.54 Compliance with County's Zero Tolerance Policy on Human Trafficking

The Contractor acknowledges that the County has established a Zero Tolerance Policy on Human Trafficking prohibiting the Contractors from engaging in human trafficking.

If a Contractor or member of the Contractor's staff is convicted of a human trafficking offense, the County shall require that the Contractor or member of the Contractor's staff be removed immediately from performing services under the Contract. The will not be under any obligation to disclose confidential information regarding the offenses other than those required by law.

Disqualification of any member of Contractor's staff pursuant to this paragraph shall not relieve Contractor of its obligation to complete all work in accordance with the terms and conditions of this Contract.

8.55 Intentionally Omitted

8.56 Compliance with Fair Chance Employment Practices

The Contractor, and its subcontractors, must comply with fair chance employment hiring practices set forth in California Government Code Section 12952. The Contractor's violation of this paragraph of the Contract may constitute a material breach of the Contract. In the event of such material breach, the County may, in its sole discretion, terminate the Contract.

8.57 Compliance with the County Policy of Equity

The Contractor acknowledges that the County takes its commitment to preserving the dignity and professionalism of the workplace very seriously, set forth in the County Policy Equity as of (https://ceop.lacounty.gov/). The Contractor further acknowledges that the County strives to provide a workplace free from discrimination. harassment, retaliation and inappropriate conduct based on a protected characteristic, and which may violate the CPOE. The Contractor, its employees and subcontractors acknowledge and certify receipt and understanding of the CPOE. Failure of the Contractor, its employees or its subcontractors to uphold the County's expectations of a workplace free from harassment and discrimination, including inappropriate conduct based on a protected characteristic, may subject the Contractor to termination of contractual agreements as well as civil liability.

8.58 Prohibition from Participation in Future Solicitation(s)

A Proposer, or a Contractor or its subsidiary or subcontractor ("Proposer/Contractor"), is prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has provided advice or consultation for the solicitation. A Proposer/Contractor is also prohibited from submitting a bid or proposal in a County solicitation if the Proposer/Contractor has developed or prepared any of the solicitation materials on behalf of the County. A violation of this provision shall result in the disqualification of the Proposer/Contractor from participation in the County solicitation or the termination or cancellation of any resultant County contract. This provision shall survive the expiration, or other termination of this Agreement.

8.59 Injury and Illness Prevention Program

The Contractor will be required to comply with the State of California's Cal OSHA's regulations. California Code of Regulations Title 8 Section 3203 requires all California employers to have a written, effective Injury and Illness Prevention Program (IIPP) that addresses hazards pertaining to the particular workplace covered by the program.

8.60 Campaign Contribution Prohibition Following Final Decision in Contract Proceeding

Pursuant to <u>Government Code Section 84308</u>, Contractor and its Subcontractors, are prohibited from making a contribution of more than \$250 to a County officer for twelve (12) months after the date of the final decision in the proceeding involving this Contract. Failure to comply with the provisions of <u>Government Code Section 84308</u> and of this paragraph, may be a material breach of this Contract as determined in the sole discretion of the County.

8.61 Compliance with County's Women in Technology Hiring Initiative

At the direction of the Board, the County has established a "Women in Technology" (WIT) Hiring Initiative focused on recruiting, training, mentoring and preparing all genders, including women, at-risk youth, and underrepresented populations (program participants) for County Information Technology (IT) careers. In support of the subject initiative, IT contractors currently offering certification, training, and/or mentoring programs must make such program(s) available to WIT program participants, if feasible. Contractors must report such programs available to: <a href="https://www.withub.com/withub.c

9 UNIQUE TERMS AND CONDITIONS

9.1 Health Insurance Portability and Accountability Act of 1996 (HIPAA)

The County is subject to the Administrative Simplification requirements and prohibitions of the Health Insurance Portability and Accountability Act of 1996, Public Law 104-191 (HIPAA), and regulations promulgated thereunder, including the Privacy, Security, Breach Notification, and Enforcement Rules at 45 Code of Federal Regulations (C.F.R.) Parts 160 and 164 (collectively, the "HIPAA Rules"). Under this Contract, the Contractor provides services to the County and the Contractor creates, has access to, receives, maintains, or transmits Protected Health Information as defined in Exhibit N (Business Associate Under Health Insurance Portability and Accountability Act of 1996 ("HIPAA")) in order to provide those services. The County and the Contractor therefore agree to the terms of Exhibit N (Business Associate Under Health Insurance Portability and Accountability Act of 1996 ("HIPAA")).

9.2 Ownership of Materials, Software and Copyright

- 9.2.1 County will be the sole owner of all right, title and interest, including copyright, in and to all software, plans, diagrams, facilities, and tools (hereafter "materials") which are originated or created through the Contractor's work pursuant to this Contract. The Contractor, for valuable consideration herein provided, must execute all documents necessary to assign and transfer to, and vest in the County all of the Contractor's right, title and interest in and to such original materials, including any copyright, patent and trade secret rights which arise pursuant to the Contractor's work under this Contract.
- 9.2.2 During the term of this Contract and for five (5) years thereafter, the Contractor must maintain and provide security for all of the Contractor's working papers prepared under this Contract. County will have the right to inspect, copy and use at any time during and subsequent to the term of this Contract, any and all such working papers and all information contained therein.
- 9.2.3 Any and all materials, software and tools which are developed or were originally acquired by the Contractor outside the scope of this Contract, which the Contractor desires to use hereunder, and which the Contractor considers to be proprietary or confidential, must be specifically identified by the Contractor to the County's Project Manager as proprietary or confidential, and must be plainly and prominently marked by the Contractor as "Proprietary" or "Confidential" on each appropriate page of any document containing such material.

- 9.2.4 The County will use reasonable means to ensure that the Contractor's proprietary and/or confidential items are safeguarded and held in confidence. The County agrees not to reproduce, distribute or disclose to non-County entities any such proprietary and/or confidential items without the prior written consent of the Contractor.
- 9.2.5 Notwithstanding any other provision of this Contract, the County will not be obligated to the Contractor in any way under subparagraph 9.2.4 for any of the Contractor's proprietary and/or confidential items which are not plainly and prominently marked with restrictive legends as required by subparagraph 9.2.3 or for any disclosure which the County is required to make under any state or federal law or order of court.

9.3 Patent, Copyright and Trade Secret Indemnification

- 9.3.1 The Contractor must indemnify, hold harmless and defend County from and against any and all liability, damages, costs, and expenses, including, but not limited to, defense costs and attorneys' fees, for or by reason of any actual or alleged infringement of any third party's patent or copyright, or any actual or alleged unauthorized trade secret disclosure, arising from or related to the operation and utilization of the Contractor's work under this Contract. County will inform the Contractor as soon as practicable of any claim or action alleging such infringement or unauthorized disclosure, and will support the Contractor's defense and settlement thereof.
- 9.3.2 In the event any equipment, part thereof, or software product becomes the subject of any complaint, claim, or proceeding alleging infringement or unauthorized disclosure, such that County's continued use of such item is formally restrained, enjoined, or subjected to a risk of damages, the Contractor, at its sole expense, and providing that County's continued use of the system is not materially impeded, must either:
 - Procure for County all rights to continued use of the questioned equipment, part, or software product; or
 - Replace the questioned equipment, part, or software product with a non-questioned item; or
 - Modify the questioned equipment, part, or software so that it is free of claims.

9.3.3 The Contractor will have no liability if the alleged infringement or unauthorized disclosure is based upon a use of the questioned product, either alone or in combination with other items not supplied by the Contractor, in a manner for which the questioned product was not designed nor intended.

9.4 Data Destruction

Contractor(s) that have maintained, processed, or stored the County data and/or information, implied or expressed, have the sole responsibility to certify that the data and information have been appropriately destroyed consistent with the National Institute of Standards and Technology (NIST) Special Publication SP 800-88 titled Guidelines for Media Sanitization. Available at:

http://csrc.nist.gov/publications/PubsDrafts.html#SP-800-88 Rev.%201

The data and/or information may be stored on purchased, leased, or rented electronic storage equipment (e.g., printers, hard drives) and electronic devices (e.g., servers, workstations) that are geographically located within the County, or external to the County's boundaries. The County must receive within ten (10) business days, a signed document from Contractor(s) that certifies and validates the data and information were placed in one or more of the following stored states: unusable, unreadable, and/or indecipherable.

Contractor(s) must certify that any County data stored on purchased, leased, or rented electronic storage equipment and electronic devices, including, but not limited to printers, hard drives, servers, and/or workstations are destroyed consistent with the current National Institute of Standard and Technology (NIST) Special Publication SP-800-88, Guidelines for Media Sanitization. Contractor(s) must provide County with written certification, within ten (10) business days of removal of any electronic storage equipment and devices that validates that any and all County data was destroyed and is unusable, unreadable, and/or indecipherable.

9.5 Contractor's Charitable Activities Compliance

The Supervision of Trustees and Fundraisers for Charitable Purposes Act regulates entities receiving or raising charitable contributions. The "Nonprofit Integrity Act of 2004" (SB 1262, Chapter 919) (SB 1262, Chapter 919) increased Charitable Purposes Act requirements. By requiring Contractors to complete the Charitable Contributions Certification, Exhibit O, the County seeks to ensure that all County contractors which receive or raise charitable contributions comply with

California law in order to protect the County and its taxpayers. A Contractor which receives or raises charitable contributions without complying with its obligations under California law commits a material breach subjecting it to either contract termination or debarment proceedings or both. (County Code Chapter 2.202)

9.6 Intentionally Omitted

9.7 Local Small Business Enterprise (LSBE) Preference Program

- 9.7.1 This Contract is subject to the provisions of the County's ordinance entitled LSBE Preference Program, as codified in Chapter 2.204 of the Los Angeles County Code.
- 9.7.2 The Contractor shall not knowingly and with the intent to defraud, fraudulently obtain, retain, attempt to obtain or retain, or aid another in fraudulently obtaining or retaining or attempting to obtain or retain certification as a LSBE.
- 9.7.3 The Contractor shall not willfully and knowingly make a false statement with the intent to defraud, whether by affidavit, report, or other representation, to a County official or employee for the purpose of influencing the certification or denial of certification of any entity as a LSBE.
- 9.7.4 If the Contractor has obtained certification as a LSBE by reason of having furnished incorrect supporting information or by reason of having withheld information, and which knew, or should have known, the information furnished was incorrect or the information withheld was relevant to its request for certification, and which by reason of such certification has been awarded this Contract to which it would not otherwise have been entitled, shall:
 - 1. Pay to the County any difference between the Contract amount and what the County's costs would have been if the Contract had been properly awarded;
 - 2. In addition to the amount described in subdivision (1), be assessed a penalty in an amount of not more than ten (10) percent of the amount of the Contract; and
 - 3. Be subject to the provisions of Chapter 2.202 of the Los Angeles County Code (Determinations of Contractor Non-responsibility and Contractor Debarment).

The above penalties shall also apply to any business that has previously obtained proper certification, however, as a result of a change in their status would no longer be eligible for certification, and fails to notify the State and the Department of Consumer and Business Affairs of this information prior to responding to a solicitation or accepting a Contract award.

9.8 Social Enterprise (SE) Preference Program

- 9.8.1 This Contract is subject to the provisions of the County's ordinance entitled SE Preference Program, as codified in Chapter 2.205 of the Los Angeles County Code.
- 9.8.2 The Contractor shall not knowingly and with the intent to defraud, fraudulently obtain, retain, attempt to obtain or retain, or aid another in fraudulently obtaining or retaining or attempting to obtain or retain certification as a SE.
- 9.8.3 The Contractor shall not willfully and knowingly make a false statement with the intent to defraud, whether by affidavit, report, or other representation, to a County official or employee for the purpose of influencing the certification or denial of certification of any entity as a SE.
- 9.8.4 If the Contractor has obtained the County certification as a SE by reason of having furnished incorrect supporting information or by reason of having withheld information, and which knew, or should have known, the information furnished was incorrect or the information withheld was relevant to its request for certification, and which by reason of such certification has been awarded this Contract to which it would not otherwise have been entitled, the Contractor shall:
 - 1. Pay to the County any difference between the Contract amount and what the County's costs would have been if the Contract had been properly awarded;
 - 2. In addition to the amount described in subdivision (1) above, the Contractor will be assessed a penalty in an amount of not more than ten percent (10%) of the amount of the Contract; and
 - 3. Be subject to the provisions of Chapter 2.202 of the Los Angeles County Code (Determinations of Contractor Non-responsibility and Contractor Debarment).

The above penalties shall also apply to any entity that has previously obtained proper certification, however, as a result of a change in their status would no longer be eligible for certification, and fails to notify the Department of Consumer and Business Affairs of this information prior to responding to a solicitation or accepting a Contract award.

9.9 Disabled Veteran Business Enterprise (DVBE) Preference Program

- 9.9.1 This Contract is subject to the provisions of the County's ordinance entitled DVBE Preference Program, as codified in Chapter 2.211 of the Los Angeles County Code.
- 9.9.2 The Contractor shall not knowingly and with the intent to defraud, fraudulently obtain, retain, attempt to obtain or retain, or aid another in fraudulently obtaining or retaining or attempting to obtain or retain certification as a DVBE.
- 9.9.3 The Contractor shall not willfully and knowingly make a false statement with the intent to defraud, whether by affidavit, report, or other representation, to a County official or employee for the purpose of influencing the certification or denial of certification of any entity as a DVBE.
- 9.9.4 If the Contractor has obtained certification as a DVBE by reason of having furnished incorrect supporting information or by reason of having withheld information, and which knew, or should have known, the information furnished was incorrect or the information withheld was relevant to its request for certification, and which by reason of such certification has been awarded this Contract to which it would not otherwise have been entitled, the Contractor shall:
 - 1. Pay to the County any difference between the Contract amount and what the County's costs would have been if the Contract had been properly awarded;
 - 2. In addition to the amount described in subdivision (1) above, the Contractor will be assessed a penalty in an amount of not more than 10 percent of the amount of the Contract; and
 - 3. Be subject to the provisions of Chapter 2.202 of the Los Angeles County Code (Determinations of Contractor Non-responsibility and Contractor Debarment).

Notwithstanding any other remedies in this Contract, the above penalties shall also apply to any business that has previously obtained proper certification, however, as a result of a change in their status would no longer be eligible for certification, and fails to notify the State and the Department of Consumer and Business Affairs of this information prior to responding to a solicitation or accepting a Contract award.

10 SURVIVAL

In addition to any terms and conditions of this Agreement that expressly survive expiration or termination of this Agreement by their terms, the following provisions shall survive the expiration or termination of this Agreement for any reason:

Paragraph 1 (Applicable Documents)

Paragraph 2 (Definitions)

Paragraph 3 (Work)

Paragraph 5.4 (No Payment for Services Provided Following

Expiration/Termination of Agreement)

Paragraph 7.6 (Confidentiality)

Paragraph 8.1 (Amendments)

Paragraph 8.2 (Assignment and Delegation/Mergers or Acquisitions)

Paragraph 8.6.2

Paragraph 8.19 (Fair Labor Standards)

Paragraph 8.20 (Force Majeure)

Paragraph 8.21 (Governing Law, Jurisdiction, and Venue)

Paragraph 8.23 (Indemnification)

Paragraph 8.24 (General Provisions for all Insurance Coverage)

Paragraph 8.25 (Insurance Coverage)

Paragraph 8.26 (Liquidated Damages)

```
Paragraph 8.34 (Notices)
Paragraph 8.38 (Record Retention and Inspection/Audit Settlement)
Paragraph 8.42 (Termination for Convenience)
Paragraph 8.43 (Termination for Default)
Paragraph 8.48 (Validity)
Paragraph 8.49 (Wavier)
Paragraph 8.58 (Prohibition from Participation in Future Solicitation(s))
Paragraph 10 (Survival)
```

IN WITNESS WHEREOF, County and Contractor have caused this Contract to be executed on their behalf by their authorized representatives, the day, month, and year first above written. The person signing on behalf of the Contractor warrants that he or she is authorized to bind the Contractor and attest under penalty of perjury to the truth and authenticity of representations made and documents submitted and incorporated as part of this Contract.

COUNTY OF LOS ANGELES PROBATION DEPARTMENT	
By: GUILLERMO VIERA ROSA CHIEF PROBATION OFFICER	
	(CONTRACTOR NAME)
	Ву:
	Name (Typed or Printed)
	Title
APPROVED AS TO FORM:	
DAWYN HARRISON COUNTY COUNSEL	
By: JASON C. CARNEVALE DEPUTY COUNTY COUNSEL	

APPENDIX D

REQUIRED FORMS

FOR

REQUEST FOR PROPOSALS (RFP)

APPENDIX D REQUIRED FORMS

TABLE OF CONTENTS

<u>EXHIE</u>	<u>PAGE</u>
BUSI	NESS FORMS
1	Proposer's Organization Questionnaire/Affidavit
1a	Community Business Enterprise (CBE) Information
2	Prospective Contractor References
3	Prospective Contractor List of Contracts6
4	Prospective Contractor List of Terminated Contracts
5	Certification of No Conflict of Interest
6	Familiarity with the County Lobbyist Ordinance Certification9
7	Request for Preference Program Consideration
8	Proposer's EEO Certification
9	Attestation of Willingness to Consider GAIN-START Participants
10	Contractor Employee Jury Service Program Certification Form and Application for Exception
COST	FORMS
11	Pricing Sheets
12	Certification of Independent Price Determination and Acknowledgement of RFP Restrictions
13-14	Intentionally Omitted
LIVIN	G WAGE FORMS
15-18	Intentionally Omitted
CERT	TIFICATIONS
19	Charitable Contributions Certification
20	Certification of Compliance with the County's Defaulted Property Tax Reduction Program
21	Zero Tolerance Policy on Human Trafficking Certification
22	Intentionally Omitted
23	Compliance with Fair Chance Employment Hiring Practices Certification 36
24	Contribution and Agent Declaration Form

REQUIRED FORMS - EXHIBIT 1 PROPOSER'S ORGANIZATION QUESTIONNAIRE/AFFIDAVIT

PAGE 1 OF 3

Please complete, sign and date this form. The person signing the form must be authorized to sign on behalf of the Proposer and to bind the applicant in a Contract.

1. Is your fir	n a corporation or limited liability com	pany (LLC)?	☐ Yes ☐ No
•	npiete. ne (found in Articles of Incorporation)		
-	no (round in 7 thoros of moorporation)		
	m is a limited partnership or a sole		
3. Is your fir	n doing business under one or more	DBA's?	☐ Yes ☐ No
If yes, co	nplete:		
Name	C	County of Registration	Year became DBA
4. Is your fir	m wholly/majority owned by, or a sub	sidiary of another firm?	 ☐ Yes ☐ No
If yes, co	nplete:		
Name of	parent firm:		
	corporation or registration of parent f		
5. Has your	firm done business as other names w	vithin last five (5) years?	☐ Yes ☐ No
If yes, co	nplete:		
Name		Yea	r of Name Change
	m involved in any pending acquisit		
□ Yes □	No If yes, provide information:		

REQUIRED FORMS - EXHIBIT 1 PROPOSER'S ORGANIZATION QUESTIONNAIRE/AFFIDAVIT

PAGE 2 OF 3

The Proposer acknowledges and certifies that the firm meets and will comply with the Proposer's Minimum Mandatory Qualifications as stated in Paragraph 3 of this Request for Proposal, as listed below.

		ropriate boxes:	
Yes	□ No	Subparagraph 3.1.1	The Proposer must submit a proposal by April 10, 2025, 12:00 p.m., PT .
Yes	□ No	Subparagraph 3.1.2	The Proposer must demonstrate that the agency has a minimum of three (3) years' experience within the last five (5) years providing services in the area of electronic monitoring, including GPS technology, transdermal alcohol monitoring and breath alcohol monitoring for a population of five hundred (500) or more participants.
Yes	□ No	Subparagraph 3.1.3	The Proposer must have an administrative business office located within the United States of America. The address to the administrative business office must be included in Exhibit 1 (Proposer's Organization Questionnaire/Affidavit) of Appendix D (Required Forms).
			The address to the administrative business office located within the United States of America must be listed here:
			Address:
			Address: City:

REQUIRED FORMS - EXHIBIT 1 PROPOSER'S ORGANIZATION QUESTIONNAIRE/AFFIDAVIT

PAGE 3 OF 3

The Proposer further acknowledges that if any false, misleading, incomplete, or deceptively unresponsive statements in connection with this proposal are made, the proposal may be rejected. The evaluation and determination in this area shall be at the Chief Probation Officer's sole judgment and his/her judgment shall be final.

<u>DECLARATION</u>: I DECLARE UNDER PENALTY OF PERJURY UNDER THE LAWS OF THE STATE OF CALIFORNIA THAT THE ABOVE INFORMATION IS TRUE AND ACCURATE.

PROPOSER NAME:			COUNTY WEBVEN NUMBER:	
ADDRESS:				
PHONE NUMBER:	E-MAIL:			
INTERNAL REVENUE SERVIONUMBER:	CE EMPLOYER IDENTIFICATION	CALIFORNIA BUSINESS LICENSE NUMBER:		
PROPOSER OFFICIAL NAME	AND TITLE (PRINT):			
SIGNATURE			DATE	

REQUIRED FORMS - EXHIBIT 1a

		CO	MMUNITY BU	JSINESS EN	TE	RPRISE (CBE) INFORM	IATION				
TITLE REFERENCE						TITLE REFERENCE					
Total Number of Employees (including						2 CERTIFICATION AS MINORITY, WOMEN, DISADVANTAGED, DISABLED VETERAN, AND LESBIAN, GAY, BISEXUAL, TRANSGENDER, QUEER, AND QUESTIONING-OWNED (LGBTQQ) BUSINESS ENTERPRISE				eteran or queer, nterprise	
owners): Race/Ethnic Composition of into the following categories:	,	nake-up of Own	ers/Partners/Assoc	iate Partners							
Race/Ethnic Composition Owners/Partners/ Associate Partners Percentage of how ownership of the firm is distributed			Agency Name	Minority	Women	Disadvantaged	Disabled Veteran	LGBTQQ			
	Male	Female	Male	Female							
Black/African American			%	%							
Hispanic/Latino			%	%							
Asian or Pacific Islander			%	%							
American Indian			%	%							
Filipino			%	%							
White			%	%							

REQUIRED FORMS - EXHIBIT 2 PROSPECTIVE CONTRACTOR REFERENCES

Contractor's Name:	

List only three (3) references where the same or similar scope of services were provided in order to meet the Minimum Requirements stated in this solicitation. If more than three (3) references are provided, only the first three (3) listed will be contacted. All dates must be in **MM/DD/YYYY** format.

1. Name of Firm	Address of Firm	Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	Type of Service End: / /	e	Dollar Amt.
2. Name of Firm	Address of Firm	Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	Type of Service End: / /	e	Dollar Amt.
3. Name of Firm	Address of Firm	Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	Type of Service End: / /	ce	Dollar Amt.

REQUIRED FORMS - EXHIBIT 3 PROSPECTIVE CONTRACTOR LIST OF CONTRACTS

Contractor's Name:	

List of all public entities for which the Contractor has provided service within the last three (3) years. All dates must be in **MM/DD/YYYY** format. Use additional sheets if necessary.

1. Name of Firm	Address of Firm		Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	End:	Type of	Service	Dollar Amt.
2. Name of Firm	Address of Firm		Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	End:	Type of	Service	Dollar Amt.
3. Name of Firm	Address of Firm		Contact Name/Title	Telephone # ()	Fax # ()
Name or Contract No.	Term of Contract Start: / /	End:	Type of	Service	Dollar Amt.
4. Name of Firm	Address of Firm		Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	End:	Type of	Service	Dollar Amt.
5. Name of Firm	Address of Firm		Contact Name/title	Telephone #	Fax # ()
Name or Contract No.	Term of Contract Start: / /	End:	Type of	Service	Dollar Amt.

REQUIRED FORMS - EXHIBIT 4 PROSPECTIVE CONTRACTOR LIST OF TERMINATED CONTRACTS

Contractor's Name:	

List of all contracts that have been terminated within the past three (3) years.

1. Name of Firm	Address of Firm	Contact Name/Title	Telephone # ()	Fax # ()
Name or Contract No.	Reason for Termination:			
2. Name of Firm	Address of Firm	Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Reason for Termination:			
3. Name of Firm	Address of Firm	Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Reason for Termination:			
4. Name of Firm	Address of Firm	Contact Name/Title	Telephone #	Fax # ()
Name or Contract No.	Reason for Termination:			

REQUIRED FORMS - EXHIBIT 5 CERTIFICATION OF NO CONFLICT OF INTEREST

The Los Angeles County Code, Section 2.180.010, provides as follows:

CONTRACTS PROHIBITED

Notwithstanding any other section of this Code, the County shall not Contract with, and shall reject any proposals submitted by, the persons or entities specified below, unless the Board of Supervisors finds that special circumstances exist which justify the approval of such Contract:

- 1. Employees of the County or of public agencies for which the Board of Supervisors is the governing body;
- 2. Profit-making firms or businesses in which employees described in number 1 serve as officers, principals, partners, or major shareholders;
- 3. Persons who, within the immediately preceding 12 months, came within the provisions of number 1, and who:
 - a. Were employed in positions of substantial responsibility in the area of service to be performed by the Contract; or
 - b. Participated in any way in developing the Contract or its service specifications; and
- 4. Profit-making firms or businesses in which the former employees, described in number 3, serve as officers, principals, partners, or major shareholders.

The Contracts submitted to the Board of Supervisors for approval or ratification shall be accompanied by an assurance by the submitting department, district or agency that the provisions of this section have not been violated.

Proposer Name		
Proposer Official Title		
Official's Signature		

REQUIRED FORMS - EXHIBIT 6 FAMILIARITY WITH THE COUNTY LOBBYIST ORDINANCE CERTIFICATION

The Proposer certifies that:

, , ,	oposer certifies that.
1)	it is familiar with the terms of the County of Los Angeles Lobbyist Ordinance, Los Angeles Code Chapter 2.160;
2)	that all persons acting on behalf of the Proposer organization have and will comply with it during the proposal process; and
3)	it is not on the County's Executive Office's List of Terminated Registered Lobbyists.

Signature: _____ Date: _____

REQUIRED FORMS - EXHIBIT 7 REQUEST FOR PREFERENCE PROGRAM CONSIDERATION

<u>INSTRUCTIONS:</u> Businesses requesting preference consideration must complete and return this form for proper consideration of the proposal. Businesses may request consideration for one or more preference programs. Check all certifications that apply.*

I MEET ALL OF THE REQUIREMENTS AND REQUEST THIS PROPOSAL BE CONSIDERED FOR THE PREFERENCE PROGRAM(S) SELECTED BELOW. A COPY OF THE CERTIFICATION LETTER ISSUED BY THE DEPARTMENT OF CONSUMER AND BUSINESS AFFAIRS (DCBA) IS ATTACHED.

mall Business Enterprise	e (LSBE) Program Prefe	erence			
		its principal place of business			
Certified as a LSBE with other certifying agencies under DCBA's inclusion policy that has its principal place of business located in Los Angeles County and has revenues and employee size that meet the State's Department of General Services requirements; and					
BE by the DCBA.					
Enterprise (SE) Program	Preference				
•	• •	•			
business by the DCBA.					
d Veterans Business En	erprise (DVBE) Prograi	m Preference			
State of California, or					
☐ Certified by U.S. Department of Veterans Affairs as a DVBE; or					
Certified as a DVBE with other certifying agencies under DCBA's inclusion policy that meets th criteria set forth by: the State of California as a DVBE or is verified as a service-disabled veterar owned small business by the Veterans Administration: and					
BE by the DCBA.					
ANY OF THE ABOVE E BE COMBINED WITH 6) IN RESPONSE TO ANY ARE UNDER PENALTY O ABOVE INFORMATION I	LISTED PREFERENCE ANY OTHER COUNTY COUNTY SOLICITATION F PERJURY UNDER TH	E PROGRAMS PRICE OI TY PROGRAM TO EXCEEI ON. HE LAWS OF THE STATE O			
	County Webven No.				
	Title:				
	Date:				
Approved	Disapproved	Date			
	State of California as a smangeles County for at least of SBE with other certifying of business located in Los ate's Department of General BBE by the DCBA. Enterprise (SE) Program has been in operation for a Transitional Workforce of California, or Department of Veterans Advisor the State of California a by: the State of Cali	of business located in Los Angeles County and has a te's Department of General Services requirements a BE by the DCBA. Enterprise (SE) Program Preference has been in operation for at least one year provice a Transitional Workforce or providing social, environal transitional transi			

REQUIRED FORMS - EXHIBIT 8 PROPOSER'S EEO CERTIFICATION

Co	mpany Name		
Ad	dress		
Int	ernal Revenue Service Employer Identification Number		
	GENERAL		
ag wil or	accordance with provisions of the County Code of the County rees that all persons employed by such firm, its affiliates, so I be treated equally by the firm without regard to or because sex and in compliance with all anti-discrimination laws of the differnia.	ubsidiaries, or holdine of race, religion, ar	ng companies are and ncestry, national origin,
	CERTIFICATION	YES	NO
1.	Proposer has written policy statement prohibiting discrimination in all phases of employment.		
2.	Proposer periodically conducts a self-analysis or utilization analysis of its work force.		
	Proposer has a system for determining if its employment practices are discriminatory against protected groups.		
4.	When problem areas are identified in employment practices Proposer has a system for taking reasonable corrective action to include establishment of goal and/or timetables.	, 	
Si	gnature	Date	
Na	ame and Title of Signer (please print)		

REQUIRED FORMS - EXHIBIT 9 ATTESTATION OF WILLINGNESS TO CONSIDER GAIN-START PARTICIPANTS

As a threshold requirement for consideration for contract award, Proposer shall demonstrate a proven record for hiring GAIN-START participants or shall attest to a willingness to consider GAIN-START participants for any future employment opening if they meet the minimum qualifications for that opening. Additionally, Proposer shall attest to a willingness to provide employed GAIN-START participants access to the Proposer's employee mentoring program, if available, to assist these individuals in obtaining permanent employment and/or promotional opportunities.

To report all job openings with job requirements to obtain qualified GAIN-START participants as potential employment candidates, Contractor shall email: GAINSTART@DPSS.LACOUNTY.GOV and BSERVICES@OPPORTUNITY.LACOUNTY.GOV

Proposers unable to meet this requirement shall not be considered for contract award.

Proposer shall complete all of the following information, sign where indicated below, and return this form with their proposal.

	•	•			
A.	Proposer has a	proven record of	f hiring GAIN-ST	TART participants.	
	☐ YES (sub	ject to verification	n by County)	□ NO	
B.	GAIN-START p meets the minir	articipants for an	y future employ ns for the openi	ob openings and job requirements to conside yment openings if the GAIN-START participar ing. "Consider" means that Proposer is willin	nt
	☐ YES	□no			
C.		ling to provide e ram, if available.	employed GAIN-	I-START participants access to its employee	∋-
	☐ YES	□ №	□ N/A (Pro	rogram not available)	
Pro	pposer's Organiza	ation:			
Sig	nature:				-
Pri	nt Name:				_
Titl	e:			Date:	_
Tel	ephone No:		-	_ Fax No:	-

REQUIRED FORMS - EXHIBIT 10 COUNTY OF LOS ANGELES CONTRACTOR EMPLOYEE JURY SERVICE PROGRAM CERTIFICATION FORM AND APPLICATION FOR EXCEPTION

The County's solicitation for this Request for Proposals is subject to the County of Los Angeles Contractor Employee Jury Service Program (Program), Los Angeles County Code, Chapter 2.203. All proposers, whether a Contractor or subcontractor, must complete this form to either certify compliance or request an exception from the Program requirements. Upon review of the submitted form, the County department will determine, in its sole discretion, whether the Proposer is given an exemption from the Program.

Company Name:			
Company Address:			
City:	State:	Zip Code:	
Telephone Number:			
Solicitation For	Services		

If you believe the Jury Service Program does not apply to your business, check the appropriate box in Part I (attach documentation to support your claim); or, complete Part II to certify compliance with the Program. Whether you complete Part I or Part II, please sign and date this form below.

Part I: Jury Service Program is Not Applicable to My Business

- My business does not meet the definition of "contractor," as defined in the Program, as it has not received an aggregate sum of \$50,000 or more in any 12-month period under one or more County Contracts or subcontracts (this exception is not available if the contract itself will exceed \$50,000). I understand that the exception will be lost and I must comply with the Program if my revenues from the County exceed an aggregate sum of \$50,000 in any 12-month period.
- My business is a small business as defined in the Program. It 1) has ten or fewer employees; <u>and</u>, 2) has annual gross revenues in the preceding twelve months which, if added to the annual amount of this contract, are \$500,000 or less; <u>and</u>, 3) is not an affiliate or subsidiary of a business dominant in its field of operation, as defined below. I understand that the exception will be lost and I must comply with the Program if the number of employees in my business and my gross annual revenues exceed the above limits.
 - **"Dominant in its field of operation"** means having more than ten employees and annual gross revenues in the preceding twelve months, which, if added to the annual amount of the contract awarded, exceed \$500,000.
 - "Affiliate or subsidiary of a business dominant in its field of operation" means a business which is at least 20 percent owned by a business dominant in its field of operation, or by partners, officers, directors, majority stockholders, or their equivalent, of a business dominant in that field of operation.
- My business is subject to a Collective Bargaining Agreement (attach agreement) that expressly provides that it supersedes all provisions of the Program.

OR

Part II: Certification of Compliance

My business <u>has</u> and adheres to a written policy that provides, on an annual basis, no less than five days of regular pay for actual jury service for full-time employees of the business who are also California residents, or my company <u>will have</u> and adhere to such a policy prior to award of the Contract.

I declare under penalty of perjury under the laws of the State of California that the information stated above is true and correct.

nt Name:	Title:
nature:	Date:

Page 1 of 7

GLOBAL POSITIONING SYSTEM MONITORING PROGRAM AND JUVENILE ELECTRONIC MONITORING PROGRAM

The undersigned offers to provide all labor and supplies necessary to provide Comprehensive Electronic Monitoring Services as set forth RFP# 6402501.

Said work shall be done for the period prescribed and in the manner set forth in said specifications, and compensation therefore shall be on a sliding scale, fixed-fee basis as provided upon the hereinafter proposed sliding scale, fixed rates. I agree that if the County Board of Supervisors accepts my proposal, I will commence services immediately following contract execution.

I agree to provide the specified services for the County of Los Angeles - Probation Department in accordance with the attached specifications for the following submitted compensation,

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING CELLULAR RADIO FREQUENCY (CRF), AS FOLLOWS:

1. Monitoring and support services only

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 – 300	 951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 – 450	 1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

Page 2 of 7

2. Monitoring and support services including installation and removal

1 – 50	701 – 750	 1401 – 1450
51 – 100	751 – 800	 1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 – 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 – 650	 1301 – 1350	
651 – 700	 1351 – 1400	

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING RADIO FREQUENCY (RF), AS FOLLOWS:

1. Monitoring and support services only

1 – 50	701 – 750	1401 – 1450
51 – 100	751 – 800	1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 – 300	951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

Page 3 of 7

1 – 50	701 – 750	 1401 – 1450
51 – 100	751 – 800	 1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 – 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 – 650	 1301 – 1350	
651 – 700	 1351 – 1400	

Page 4 of 7

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING GPS - PASSIVE, AS FOLLOWS:

1. Monitoring and support services only

1 – 50	 701 – 750	 1401 – 1450
51 – 100	751 – 800	1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 - 250	 901 – 950	 1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	1701 – 1750
351 - 400	 1051 – 1100	 1751 – 1800
401 – 450	1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

1 – 50	701 – 750	1401 – 1450
51 – 100	751 – 800	1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 – 300	951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 - 450	1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

Page 5 of 7

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING GPS - INTERMEDIATE, AS FOLLOWS:

1. Monitoring and support services only

1 - 50		701 – 750	1401 – 1450
51 – 100		751 – 800	1451 – 1500
101 – 150		801 - 850	 1501 – 1550
151 – 200		851 – 900	 1551 – 1600
201 - 250		901 – 950	 1601 – 1650
251 - 300		951 – 1000	 1651 – 1700
301 - 350		1001 – 1050	 1701 – 1750
351 – 400		1051 – 1100	 1751 – 1800
401 - 450		1101 – 1150	 1801 – 1850
451 – 500		1151 – 1200	 1851 – 1900
501 – 550		1201 – 1250	 1901 – 1950
551 – 600		1251 – 1300	 1951 – 2000
601 - 650		1301 – 1350	
651 – 700		1351 – 1400	

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	851 – 900	 1551 – 1600
201 - 250	901 – 950	 1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 - 400	 1051 – 1100	 1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

Page 6 of 7

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING GPS – CONTINUOUS (ACTIVE), AS FOLLOWS:

1. Monitoring and support services only

1 – 50	701 – 750	1401 – 1450
51 – 100	751 – 800	1451 – 1500
101 – 150	801 – 850	1501 – 1550
151 – 200	851 – 900	 1551 – 1600
201 – 250	901 – 950	1601 – 1650
251 – 300	951 – 1000	1651 – 1700
301 – 350	1001 – 1050	 1701 – 1750
351 – 400	1051 – 1100	1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	1151 – 1200	1851 – 1900
501 – 550	1201 – 1250	1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	1451 – 1500
101 – 150	 801 – 850	1501 – 1550
151 - 200	851 – 900	1551 – 1600
201 – 250	901 – 950	1601 – 1650
251 – 300	 951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 - 400	1051 – 1100	1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	1201 – 1250	1901 – 1950
551 – 600	1251 – 1300	1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

Page 7 of 7

Respectfully submitted,
Printed Name
Title
Signature
Date:
Address:
City:
Phone:

Page 1 of 7

Adult Electronic Monitoring Program

The undersigned offers to provide all labor and supplies necessary to provide Comprehensive Electronic Monitoring Services as set forth RFP# 6402501.

Said work shall be done for the period prescribed and in the manner set forth in said specifications, and compensation therefore shall be on a sliding scale, fixed-fee basis as provided upon the hereinafter proposed sliding scale, fixed rates. I agree that if the County Board of Supervisors accepts my proposal, I will commence services immediately following contract execution.

I agree to provide the specified services for the County of Los Angeles - Probation Department in accordance with the attached specifications that includes but not limited to:

- All staff and facilities
- Installation and removal of devices
- Case management as specified by Adult Probation's Procedure Manual, Technical Exhibit 7 of Appendix B (Statement of Work, Technical Exhibits)
 - Case Management meetings weekly or twice a month depending on risk level and behavior
 - Monitoring of whereabouts and incident reporting
 - Other as specified.
- Court Reminders to participants
- Letters and emails to participants

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING CELLULAR RADIO FREQUENCY (CRF), AS FOLLOWS:

1. Monitoring and support services only

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 - 250	 901 – 950	 1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 – 450	 1101 – 1150	 1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 - 700	 1351 – 1400	

Page 2 of 7

1 – 50	 701 – 750	 1401 – 1450
51 – 100	751 – 800	 1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 - 300	951 – 1000	1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 - 450	1101 – 1150	1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	1251 – 1300	1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING RADIO FREQUENCY (RF), AS FOLLOWS:

1. Monitoring and support services only

1 - 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	901 – 950	 1601 – 1650
251 – 300	951 – 1000	 1651 – 1700
301 - 350	1001 – 1050	 1701 – 1750
351 – 400	1051 – 1100	 1751 – 1800
401 - 450	 1101 – 1150	 1801 – 1850
451 – 500	1151 – 1200	 1851 – 1900
501 – 550	1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	1301 – 1350	
651 – 700	 1351 – 1400	

1 – 50	701 – 750	1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	801 - 850	1501 – 1550
151 – 200	851 – 900	 1551 – 1600
201 - 250	901 – 950	1601 – 1650
251 – 300	 951 – 1000	 1651 – 1700
301 - 350	1001 – 1050	 1701 – 1750
351 – 400	1051 – 1100	 1751 – 1800
401 - 450	1101 – 1150	1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	1251 – 1300	 1951 – 2000
601 - 650	1301 – 1350	
651 – 700	 1351 – 1400	

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING GPS - PASSIVE, AS FOLLOWS:

1. Monitoring and support services only

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 – 300	 951 – 1000	 1651 – 1700
301 – 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 – 450	1101 – 1150	1801 – 1850
451 – 500	1151 – 1200	1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

1 – 50	 701 – 750	1401 – 1450
51 – 100	 751 – 800	1451 – 1500
101 – 150	 801 – 850	1501 – 1550
151 – 200	 851 – 900	1551 – 1600
201 – 250	 901 – 950	1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 - 350	 1001 – 1050	 1701 – 1750
351 - 400	 1051 – 1100	 1751 – 1800
401 – 450	1101 – 1150	1801 – 1850
451 – 500	 1151 – 1200	1851 – 1900
501 – 550	 1201 – 1250	1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING GPS - INTERMEDIATE, AS FOLLOWS:

1. Monitoring and support services only

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	 851 – 900	 1551 – 1600
201 – 250	 901 – 950	 1601 – 1650
251 – 300	 951 – 1000	 1651 – 1700
301 – 350	 1001 – 1050	 1701 – 1750
351 – 400	 1051 – 1100	 1751 – 1800
401 – 450	1101 – 1150	1801 – 1850
451 – 500	1151 – 1200	1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	 1251 – 1300	 1951 – 2000
601 - 650	 1301 – 1350	
651 – 700	 1351 – 1400	

1 – 50	 701 – 750	1401 – 1450
51 – 100	 751 – 800	1451 – 1500
101 – 150	 801 – 850	 1501 – 1550
151 – 200	 851 – 900	1551 – 1600
201 – 250	901 – 950	1601 – 1650
251 - 300	 951 – 1000	 1651 – 1700
301 - 350	1001 – 1050	1701 – 1750
351 - 400	1051 – 1100	1751 – 1800
401 - 450	1101 – 1150	1801 – 1850
451 – 500	 1151 – 1200	 1851 – 1900
501 – 550	 1201 – 1250	 1901 – 1950
551 – 600	1251 – 1300	1951 – 2000
601 - 650	1301 – 1350	
651 - 700	 1351 – 1400	

REQUIRED FORMS - EXHIBIT 11-B PRICING SHEET

Page 6 of 7

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR THE REQUIRED SERVICES USING GPS – CONTINUOUS (ACTIVE), AS FOLLOWS:

1. Monitoring and support services only

1 – 50	 701 – 750	 1401 – 1450
51 – 100	 751 – 800	 1451 – 1500
101 – 150	801 – 850	 1501 – 1550
151 – 200	851 – 900	 1551 – 1600
201 – 250	901 – 950	 1601 – 1650
251 – 300	951 – 1000	 1651 – 1700
301 – 350	1001 – 1050	 1701 – 1750
351 – 400	1051 – 1100	 1751 – 1800
401 – 450	 1101 – 1150	 1801 – 1850
451 – 500	1151 – 1200	 1851 – 1900
501 – 550	1201 – 1250	 1901 – 1950
551 – 600	1251 – 1300	 1951 – 2000
601 - 650	1301 – 1350	
651 – 700	 1351 – 1400	

	• • •		•	
1 – 50		701 – 750		1401 – 1450
51 – 100		751 – 800		1451 – 1500
101 – 150		801 – 850		1501 – 1550
151 – 200		851 – 900		1551 – 1600
201 - 250		901 – 950		1601 – 1650
251 - 300		951 – 1000		1651 – 1700
301 - 350		1001 – 1050		1701 – 1750
351 - 400		1051 – 1100		1751 – 1800
401 - 450		1101 – 1150		1801 – 1850
451 – 500		1151 – 1200		1851 – 1900
501 – 550		1201 – 1250		1901 – 1950
551 – 600		1251 – 1300		1951 – 2000
601 - 650		1301 – 1350		
651 – 700		1351 – 1400		

REQUIRED FORMS - EXHIBIT 11-B PRICING SHEET

Page 7 of 7

Respectfully submitted,
Printed Name
Title
Signature
Date:
Address:
City:
Phone:

REQUIRED FORMS - EXHIBIT 11-C PRICING SHEET

The undersigned offers to provide all labor and supplies necessary to provide Comprehensive Electronic Monitoring Services as set forth RFP# 6402501.

Said work shall be done for the period prescribed and in the manner set forth in said specifications, and compensation therefore shall be on a sliding scale, fixed-fee basis as provided upon the hereinafter proposed sliding scale, fixed rates. I agree that if the County Board of Supervisors accepts my proposal, I will commence services immediately following contract execution.

I agree to provide the specified services for the County of Los Angeles - Probation Department in accordance with the attached specifications for the following submitted compensation,

I PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR TRANSDERMAL ALCOHOL MONITORING, AS FOLLOWS:

1 – 50		701 – 750	1401 – 1450
51 – 100		751 – 800	 1451 – 1500
101 – 150		801 – 850	 1501 – 1550
151 – 200		851 – 900	 1551 – 1600
201 - 250		901 – 950	 1601 – 1650
251 - 300		951 – 1000	 1651 – 1700
301 - 350		1001 – 1050	 1701 – 1750
351 – 400		1051 – 1100	 1751 – 1800
401 – 450		1101 – 1150	 1801 – 1850
451 – 500		1151 – 1200	 1851 – 1900
501 – 550		1201 – 1250	 1901 – 1950
551 – 600		1251 – 1300	 1951 – 2000
601 - 650		1301 – 1350	
651 – 700		1351 – 1400	

Respectfully submitted,

Printed Name
Title
Signature
Date:
Address:
City:
Phone:

REQUIRED FORMS - EXHIBIT 11-D PRICING SHEET

The undersigned offers to provide all labor and supplies necessary to provide Comprehensive Electronic Monitoring Services as set forth RFP# 6402501.

Said work shall be done for the period prescribed and in the manner set forth in said specifications, and compensation therefore shall be on a sliding scale, fixed-fee basis as provided upon the hereinafter proposed sliding scale, fixed rates. I agree that if the County Board of Supervisors accepts my proposal, I will commence services immediately following contract execution.

I agree to provide the specified services for the County of Los Angeles - Probation Department in accordance with the attached specifications for the following submitted compensation,

II PROPOSE A SLIDING SCALE DAILY FIXED RATE/FEE FOR BREATH ALCOHOL REMOTE TESTING MONITORING, AS FOLLOWS:

1 – 50		701 – 750	 1401 – 1450
51 – 100		751 – 800	 1451 – 1500
101 – 150		801 – 850	 1501 – 1550
151 - 200		851 – 900	 1551 – 1600
201 - 250		901 – 950	 1601 – 1650
251 - 300		951 – 1000	 1651 – 1700
301 - 350		1001 – 1050	 1701 – 1750
351 – 400		1051 – 1100	 1751 – 1800
401 – 450		1101 – 1150	 1801 – 1850
451 – 500		1151 – 1200	 1851 – 1900
501 – 550		1201 – 1250	 1901 – 1950
551 – 600		1251 – 1300	 1951 – 2000
601 - 650		1301 – 1350	
651 – 700		1351 – 1400	

Respectfully submitted,

Printed Name
Title
Signature
Date:
Address:
City:
Phone:

REQUIRED FORMS - EXHIBIT 12 CERTIFICATION OF INDEPENDENT PRICE DETERMINATION AND ACKNOWLEDGEMENT OF RFP RESTRICTIONS

A.	By submission of this Proposal, Proposer certifies that the prices quoted herein have bee arrived at independently without consultation, communication, or agreement with any other Proposer or competitor for the purpose of restricting competition.		
В.	List all names and telephone numb	per of person legally authorized to commit the Proposer.	
	NAME	PHONE NUMBER	
	NOTE: Persons signing on behalf authorized to bind the Cor	of the Contractor will be required to warrant that they are stractor.	
C.		partners, subcontractors, or others having any right or eeds thereof. If not applicable, state "NONE".	
D.	preparation, or selection process a	as not participated as a consultant in the development, associated with this RFP. Proposer understands that, if it the Proposer did participate as a consultant in this RFP s proposal.	
Nan	ne of Firm		
Prin	t Name of Signer	Title	
Sigr	nature	Date	

REQUIRED FORMS – EXHIBITS 13-18

INTENTIONALLY OMITTED

REQUIRED FORMS - EXHIBIT 19 CHARITABLE CONTRIBUTIONS CERTIFICATION

Com	npany Name
Add	ress
Inter	rnal Revenue Service Employer Identification Number
Calif	fornia Registry of Charitable Trusts "CT" number (if applicable)
Sup	Nonprofit Integrity Act (SB 1262, Chapter 919) added requirements to California's ervision of Trustees and Fundraisers for Charitable Purposes Act which regulates be receiving and raising charitable contributions.
Che	ck the Certification below that is applicable to your company.
	Proposer or Contractor has examined its activities and determined that it does not now receive or raise charitable contributions regulated under California's Supervision of Trustees and Fundraisers for Charitable Purposes Act. It Proposer engages in activities subjecting it to those laws during the term of a County contract, it will timely comply with them and provide County a copy of its initial registration with the California State Attorney General's Registry of Charitable Trusts when filed.
	OR
	Proposer or Contractor is registered with the California Registry of Charitable Trusts under the CT number listed above and is in compliance with its registration and reporting requirements under California law. Attached is a copy of its most recent filing with the Registry of Charitable Trusts as required by Title 11 California Code of Regulations, sections 300-301 and Government Code sections 12585-12586.
Sign	nature Date
Plea	ase Print Name and Title of Signer

REQUIRED FORMS - EXHIBIT 20 CERTIFICATION OF COMPLIANCE WITH THE COUNTY'S DEFAULTED PROPERTY TAX REDUCTION PROGRAM

	Company Name:				
	Company Address:				
	City:	State:		Zip Code:	
	Telephone Number:	Email	address:		
	Solicitation/Contract For		Services		
Th	ne Proposer/Bidder/Contractor	certifies that:			
	It is familiar with the terms of the County of Los Angeles Defaulted Property Tax Reduction Program, Los Angeles County Code Chapter 2.206; AND				
	To the best of its knowledge, after a reasonable inquiry, the Proposer/Bidder/Contractor is not in default, as that term is defined in Los Angeles County Code Section 2.206.020.E, on any Los Angeles County property tax obligation; AND				
	The Proposer/Bidder/Contractor agrees to comply with the County's Defaulted Property Tax Reduction Program during the term of any awarded Contract.				
		- OR -			
	I am exempt from the County of Los Angeles Defaulted Property Tax Reduction Program, pursuant to Los Angeles County Code Section 2.206.060, for the following reason:				
					
	I declare under penalty of perjury under the laws of the State of California that the information stated above is true and correct.				
	Print Name:		Title:		
	Signature:		Date:		

REQUIRED FORMS - EXHIBIT 21 ZERO TOLERANCE POLICY ON HUMAN TRAFFICKING CERTIFICATION

Company Name:		
Company Address:		
City:	State:	Zip Code:
Telephone Number:	Email address:	:
Solicitation/Contract for		Services
PROPOSER C	CERTIFICATIO	N
Los Angeles County has taken significant steps to protect victims of human trafficking by establishing a zero tolerance policy on human trafficking that prohibits contractors found to have engaged in human trafficking from receiving contract awards or performing services under a County contract.		
Proposer acknowledges and certifies compliance with Section 8.54 (Compliance with County's Zero Tolerance Policy on Human Trafficking) of the proposed Contract and agrees that proposer or a member of his staff performing work under the proposed Contract will be in compliance. Proposer further acknowledges that noncompliance with the County's Zero Tolerance Policy on Human Trafficking may result in rejection of any proposal, or cancellation of any resultant Contract, at the sole judgment of the County.		
I declare under penalty of perjury under the laws of the State of California that the information herein is true and correct and that I am authorized to represent this company.		
Print Name:		Title:
Signature:	- 1	Date:

REQUIRED FORM – EXHIBIT 22

INTENTIONALLY OMITTED

REQUIRED FORMS - EXHIBIT 23 COMPLIANCE WITH FAIR CHANCE EMPLOYMENT HIRING PRACTICES CERTIFICATION

Company Name:				
Company Address:				
City:	State:	Zip Code:		
Telephone Number:	Email addres	s:		
Solicitation/Contract for		Services		
PROPOSER/CONTR	PROPOSER/CONTRACTOR CERTIFICATION			
The Los Angeles County Board of Supervisors approved a Fair Chance Employment Policy in an effort to remove job barriers for individuals with criminal records. The policy requires businesses that contract with the County to comply with fair chance employment hiring practices set forth in California Government Code Section 12952, Employment Discriminations Conviction History (California Government Code Section 12952), effective January 1, 2018.				
Proposer/Contractor acknowledges and certifies compliance with fair chance employment hiring practices set forth in California Government Code Section 12952 and agrees that proposer/contractor and staff performing work under the Contract will be in compliance Proposer/Contractor further acknowledges that noncompliance with fair chance employment practices set forth in California Government Code Section 12952 may result in rejection of any proposal, or termination of any resultant Contract, at the sole judgment of the County.				
I declare under penalty of perjury under the laws of the State of California that the information herein is true and correct and that I am authorized to represent this company.				
Print Name:		Title:		
Signature:		Date:		

Page 1 of 5

This form must be completed separately by all bidders/proposers, including all prime contractors and subcontractors, and by all applicants for licenses, permits, and other entitlements for use issued by the County of Los Angeles ("County").

Pursuant to the Levine Act (<u>Government Code Section 84308</u>), a member of the Board of Supervisors, other elected County officials (the Sheriff, Assessor, and the District Attorney), and other County employees and/or officers ("County Officers") are disqualified and not able to participate in a proceeding involving contracts, franchises, licenses, permits and other entitlements for use if the County Officer received more than \$250 in contributions in the past 12 months from the bidder, proposer or applicant, any paid agent of the bidder, proposer, or applicant, or any financially interested participant who actively supports or opposes a particular decision in the proceeding.

State law requires you to disclose information about contributions made by you, your company, and lobbyists and agents paid to represent you. Failure to complete the form in its entirety may result in significant delays in the processing of your application and potential disqualification from the procurement or application process.

You must fully answer the applicable questions below. You ("Declarant"), or your company, if applicable, including all entities identified below (collectively, "Declarant Company") must also answer the questions below. The term "employee(s)" shall be defined as employees, officers, partners, owners, or directors of Declarant Company.

An affirmative response to any questions will not automatically cause the disqualification of your bid/proposal, or the denial of your application for a license, permit or other entitlement. However, failure to answer questions completely, in good faith, or providing materially false answers may subject a bidder/proposer to disqualification from the procurement.

This material is intended for use by bidders/proposers, including all prime contractors and subcontractors, and by all applicants for licenses, permits, and other entitlements for use issued by the County of Los Angeles and does not constitute legal advice. If you have questions about the Levine Act and how it applies to you, you should call your lawyer or contact the Fair Political Practices Commission for further guidance.

Page 2 of 5

Complete each section below. State "none" if applicable.

A. <u>COMPANY OR APPLICANT INFORMATION</u>

- 1) Declarant Company or Applicant Name:
 - a)If applicable, identify all subcontractors that have been or will be named in your bid or proposal:
 - b)If applicable, variations and acronyms of Declarant Company's name used within the past 12 months:
 - c)Identify all entities or individuals who have the authority to make decisions for you or Declarant Company about making contributions to a County Officer, regardless of whether you or Declarant Company have actually made a contribution:

[IF A COMPANY, ANSWER QUESTIONS 2 - 3]

- 2) Identify <u>only</u> the Parent(s), Subsidiaries and Related Business Entities that Declarant Company has controlled or directed, or been controlled or directed by. "Controlled or directed" means shared ownership, 50% or greater ownership, or shared management and control between the entities.
 - a) Parent(s):
 - b) Subsidiaries:
 - c) Related Business Entities:
- 3) If Declarant Company is a closed corporation (non-public, with under 35 shareholders), identify the majority shareholder.
- 4) Identify all entities (proprietorships, firms, partnerships, joint ventures, syndicates, business trusts, companies, corporations, limited liability companies, associations, committees, and any other organization or group of persons acting in concert) whose contributions you or Declarant Company have the authority to direct or control.

Page 3 of 5

5) Identify any individuals such as employees, agents, attorneys, law firms, lobbyists, and lobbying firms who are or who will act on behalf of you or Declarant Company and who will receive compensation to communicate with a County Officer regarding the award or approval of **this** contract or project, license, permit, or other entitlement for use.

(Do **not** list individuals and/or firms who, as part of their profession, either (1) submit to the County drawings or submissions of an architectural, engineering, or similar nature, **or** (2) provide purely technical data or analysis, **and** who will not have any other type of communication with a County agency, employee, or officer.)

6) If you or Declarant Company are a 501(c)(3) non-profit organization, identify the compensated officers of your organization and the compensated members of your board.

B. CONTRIBUTIONS

1) Have you or the Declarant Company solicited or directed your employee(s) or agent(s) to make contributions, whether through fundraising events, communications, or any other means, to a County Officer in the past 12 months? If so, provide details of each occurrence, including the date.

Date (contribution solicited, or directed)	Recipient Name (elected official)	Amount

^{*}Please attach an additional page, if necessary.

2) Disclose all contributions made by you or any of the <u>entities and individuals</u> <u>identified</u> in Section A to a County office in the past 12 months.

Date (contribution solicited, or directed)	Recipient Name (elected official)	Amount

^{*}Please attach an additional page, if necessary.

Page 4 of 5

C. DECLARATION

Signature

DECLARATION
By signing this Contribution and Agent Declaration form, you (Declarant), or you and the Declarant Company, if applicable, attest that you have read the entirety of the Contribution Declaration and the statements made herein are true and correct to the best of your knowledge and belief. (Only complete the one section that applies.)
There areadditional pages attached to this Contribution Declaration Form.
COMPANY BIDDERS OR APPLICANTS
I, (Authorized Representative), on behalf of (Declarant Company), at which I am employed as (Title), attest that after having made or caused to be made a reasonably diligent investigation regarding the Declarant Company, the foregoing responses, and the explanation on the attached page(s), if any, are correct to the best of my knowledge and belief. Further, I understand that failure to answer the questions in good faith or providing materially false answers may subject Declarant Company to consequences, including disqualification of its bid/proposal or delays in the processing of the requested contract, license, permit, or other entitlement.
IMPORTANT NOTICE REGARDING FUTURE AGENTS AND FUTURE CONTRIBUTIONS:
By signing this Contribution and Agent Declaration form, you also agree that, if Declarant Company hires an agent, such as, but not limited to, an attorney or lobbyist during the course of these proceedings and will compensate them for communicating with the County about this contract, project, permit, license, or other entitlement for use, you agree to inform the County of the identity of the agent or lobbyist and the date of their hire. You also agree to disclose to the County any future contributions made to members of the County Board of Supervisors, another elected County officer (the Sheriff, Assessor, and the District Attorney), or any other County officer or employee by the Declarant Company, or, if applicable, any of the Declarant Company's proposed subcontractors, agents, lobbyists, and employees who have communicated or will communicate with the County about this contract, license, permit, or other entitlement after the date of signing this disclosure form, and within 12 months following the approval, renewal, or extension of the requested contract, license, permit, or entitlement for use.

Date

Page 5 of 5

INDIVIDUAL BIDDERS OR APPLICANTS			
declare that the foregoing responses and the explanation the attached sheet(s), if any, are correct to the best of my knowledge and belief. Further, I understand that failure to answer the questions in good faith of providing materially false answers may subject me to consequences, including disqualification of my bid/proposal or delays in the processing of the requested license, permit, or other entitlement.			
IMPORTANT NOTICE REGARDING FUTURE ACCONTRIBUTIONS:	GENTS AND FUTURE		
If I hire an agent or lobbyist during the course of the compensate them for communicating with the County at permit, license, or other entitlement for use, I agree to identity of the agent or lobbyist and the date of their hire to the County any future contributions made to member Supervisors, another elected County official (the Sh District Attorney), or any other County officer or emplosuch as, but not limited to, a lobbyist or attorney represafter the date of signing this disclosure form, and within approval, renewal, or extension of the requested contentitlement for use.	bout this contract, project, inform the County of the e. I also agree to disclose its of the County Board of heriff, Assessor, and the oyee by me, or an agent senting me, that are made in 12 months following the		
Signature	Date		

APPENDICES E - N

APPENDICES E - N

TABLE OF CONTENTS

<u>APPENDIX</u>		PAGE
E.	RFP Transmittal to Request a Solicitation Requirements Review	1
F.	County of Los Angeles Policy on Doing Business with Small Business	2
G.	Jury Service Ordinance	3
H.	Listing of Contractors Debarred in Los Angeles County	6
l.	IRS Notice 1015	7
J - L.	Intentionally Omitted	8
M.	Background and Resources: California Charities Regulation	9
N.	Defaulted Property Tax Reduction Program	11

REQUEST FOR PROPOSALS (RFP) TRANSMITTAL TO REQUEST A SOLICITATION REQUIREMENTS REVIEW

Proposer requesting a Solicitation Requirements Review must submit this form to the County within the timeframe identified in the solicitation document.

Proposer Name:	Date of Request:			
Solicitation Title:	Solicitation No.:			
A Solicitation Requirements Review is being requested because the Proposer asserts that they are being unfairly disadvantaged for the following reason(s): (check all that apply)				
☐ Application of Minimum Requirements				
☐ Application of Evaluation Criteria				
☐ Application of Business Requirements				
 Due to unclear instructions, the process may result in the best possible responses 	e County not receiving the			
For each area contested, Proposer must explain in detail the factual reasons for the requested review. (Attach supporting documentation.)				
Request submitted by:				
(Name) (Title)			
For County use only				
Date Transmittal Received by County: Date Solicitation Released:				
Reviewed by:				

COUNTY OF LOS ANGELES POLICY ON DOING BUSINESS WITH SMALL BUSINESS

Forty-two percent of businesses in Los Angeles County have five or fewer employees. Only about four percent of businesses in the area exceed 100 employees. According to the Los Angeles Times and local economists, it is not large corporations, but these small companies that are generating new jobs and helping move Los Angeles County out of its worst recession in decades.

WE RECOGNIZE...

The importance of small business to the County. . .

- in fueling local economic growth
- providing new jobs
- creating new local tax revenues
- offering new entrepreneurial opportunity to those historically under-represented in business

The County can play a positive role in helping small business grow. . .

- as a multi-billion dollar purchaser of goods and services
- as a broker of intergovernmental cooperation among numerous local jurisdictions
- by greater outreach in providing information and training
- by simplifying the bid/proposal process
- by maintaining selection criteria which are fair to all
- by streamlining the payment process

WE THEREFORE SHALL:

- 1. Constantly seek to streamline and simplify our processes for selecting our vendors and for conducting business with them.
- Maintain a strong outreach program, fully-coordinated among our departments and districts, as well as other participating governments to: a) inform and assist the local business community in competing to provide goods and services; b) provide for ongoing dialogue with and involvement by the business community in implementing this policy.
- 3. Continually review and revise how we package and advertise solicitations, evaluate and select prospective vendors, address subcontracting and conduct business with our vendors, in order to: a) expand opportunity for small business to compete for our business; and b) to further opportunities for all businesses to compete regardless of size.
- Insure that staff who manage and carry out the business of purchasing goods and services are well trained, capable and highly motivated to carry out the letter and spirit of this policy.

Title 2 ADMINISTRATION Chapter 2.203.010 through 2.203.090 CONTRACTOR EMPLOYEE JURY SERVICE

Page 1 of 3

2.203.010 Findings.

The board of supervisors makes the following findings. The county of Los Angeles allows its permanent, full-time employees unlimited jury service at their regular pay. Unfortunately, many businesses do not offer or are reducing or even eliminating compensation to employees who serve on juries. This creates a potential financial hardship for employees who do not receive their pay when called to jury service, and those employees often seek to be excused from having to serve. Although changes in the court rules make it more difficult to excuse a potential juror on grounds of financial hardship, potential jurors continue to be excused on this basis, especially from longer trials. This reduces the number of potential jurors and increases the burden on those employers, such as the county of Los Angeles, who pay their permanent, full-time employees while on juror duty. For these reasons, the county of Los Angeles has determined that it is appropriate to require that the businesses with which the county contracts possess reasonable jury service policies. (Ord. 2002-0015 § 1 (part), 2002)

2.203.020 Definitions.

The following definitions shall be applicable to this chapter:

- A. "Contractor" means a person, partnership, corporation or other entity which has a contract with the county or a subcontract with a county contractor and has received or will receive an aggregate sum of \$50,000 or more in any 12-month period under one or more such contracts or subcontracts.
- B. "Employee" means any California resident who is a full-time employee of a contractor under the laws of California.
- C. "Contract" means any agreement to provide goods to, or perform services for or on behalf of, the county but does not include:
 - 1. A contract where the board finds that special circumstances exist that justify a waiver of the requirements of this chapter; or
 - 2. A contract where federal or state law or a condition of a federal or state program mandates the use of a particular contractor; or
 - 3. A purchase made through a state or federal contract; or
 - 4. A monopoly purchase that is exclusive and proprietary to a specific manufacturer, distributor, or reseller, and must match and inter-member with existing supplies, equipment or systems maintained by the county pursuant to the Los Angeles County Purchasing Policy and Procedures Manual, Section P-3700 or a successor provision; or
 - 5. A revolving fund (petty cash) purchase pursuant to the Los Angeles County Fiscal Manual, Section 4.4.0 or a successor provision; or
 - 6. A purchase card purchase pursuant to the Los Angeles County Purchasing Policy and Procedures Manual, Section P-2810 or a successor provision; or
 - 7. A non-agreement purchase with a value of less than \$5,000 pursuant to the Los Angeles County Purchasing Policy and Procedures Manual, Section A-0300 or a successor provision; or
 - 8. A bona fide emergency purchase pursuant to the Los Angeles County Purchasing Policy and Procedures Manual, Section PP-1100 or a successor provision

Title 2 ADMINISTRATION Chapter 2.203.010 through 2.203.090 CONTRACTOR EMPLOYEE JURY SERVICE

Page 2 of 3

- D. "Full time" means 40 hours or more worked per week, or a lesser number of hours if:
 - 1. The lesser number is a recognized industry standard as determined by the chief administrative officer,
 - 2. The contractor has a long-standing practice that defines the lesser number of hours as full time.
- E. "County" means the county of Los Angeles or any public entities for which the board of supervisors is the governing body. (Ord. 2002-0040 § 1, 2002: Ord. 2002-0015 § 1 (part), 2002)

2.203.030 Applicability.

This chapter shall apply to contractors who enter into contracts that commence after July 11, 2002. This chapter shall also apply to contractors with existing contracts which are extended into option years that commence after July 11, 2002. Contracts that commence after May 28, 2002, but before July 11, 2002, shall be subject to the provisions of this chapter only if the solicitations for such contracts stated that the chapter would be applicable. (Ord. 2002-0040 § 2, 2002: Ord. 2002-0015 § 1 (part), 2002)

2.203.040 Contractor Jury Service Policy.

A contractor shall have and adhere to a written policy that provides that its employees shall receive from the contractor, on an annual basis, no less than five days of regular pay for actual jury service. The policy may provide that employees deposit any fees received for such jury service with the contractor or that the contractor deduct from the employees' regular pay the fees received for jury service. (Ord. 2002-0015 § 1 (part), 2002)

2.203.050 Other Provisions.

- A. Administration. The chief administrative officer shall be responsible for the administration of this chapter. The chief administrative officer may, with the advice of county counsel, issue interpretations of the provisions of this chapter and shall issue written instructions on the implementation and ongoing administration of this chapter. Such instructions may provide for the delegation of functions to other county departments.
- B. Compliance Certification. At the time of seeking a contract, a contractor shall certify to the county that it has and adheres to a policy consistent with this chapter or will have and adhere to such a policy prior to award of the contract. (Ord. 2002-0015 § 1 (part), 2002)

2.203.060 Enforcement and Remedies.

For a contractor's violation of any provision of this chapter, the county department head responsible for administering the contract may do one or more of the following:

- 1. Recommend to the board of supervisors the termination of the contract; and/or,
- 2. Pursuant to chapter 2.202, seek the debarment of the contractor. (Ord. 2002-0015 § 1 (part), 2002)

Title 2 ADMINISTRATION Chapter 2.203.010 through 2.203.090 CONTRACTOR EMPLOYEE JURY SERVICE

Page 3 of 3

2.203.070. Exceptions.

- A. Other Laws. This chapter shall not be interpreted or applied to any contractor or to any employee in a manner inconsistent with the laws of the United States or California.
- B. Collective Bargaining Agreements. This chapter shall be superseded by a collective bargaining agreement that expressly so provides.
- C. Small Business. This chapter shall not be applied to any contractor that meets all of the following:
 - 1. Has ten or fewer employees during the contract period; and,
 - 2. Has annual gross revenues in the preceding twelve months which, if added to the annual amount of the contract awarded, are less than \$500,000; and,
 - 3. Is not an affiliate or subsidiary of a business dominant in its field of operation.

"Dominant in its field of operation" means having more than ten employees and annual gross revenues in the preceding twelve months which, if added to the annual amount of the contract awarded, exceed \$500,000.

"Affiliate or subsidiary of a business dominant in its field of operation" means a business which is at least 20 percent owned by a business dominant in its field of operation, or by partners, officers, directors, majority stockholders, or their equivalent, of a business dominant in that field of operation. (Ord. 2002-0015 § 1 (part), 2002)

2.203.090. Severability.

If any provision of this chapter is found invalid by a court of competent jurisdiction, the remaining provisions shall remain in full force and effect. (Ord. 2002-0015 § 1 (part), 2002)

LISTING OF CONTRACTORS DEBARRED IN LOS ANGELES COUNTY

List of Debarred Contractors in Los Angeles County may be obtained by going to the following website.

https://doingbusiness.lacounty.gov/listing-of-contractors-debarred-in-los-angeles-county/

IRS NOTICE 1015



Notice 1015

(Rev. December 2023)

Have You Told Your Employees About the Earned Income Credit (EIC)?

What Is the EIC?

The EIC is a refundable tax credit for certain workers.

Which Employees Must I Notify About the EIC?

You must notify each employee who worked for you at any time during the year and from whose wages you did not withhold income tax. However, you do not have to notify any employee who claimed exemption from withholding on Form W-4, Employee's Withholding Certificate.

Note: You are encouraged to notify all employees whose wages for 2023 are less than \$63,398 that they may be eligible for the EIC.

How and When Must I Notify My Employees?

You must give the employee one of the following.

- The IRS Form W-2, Wage and Tax Statement, which has the required information about the EIC on the back of Copy B.
- A substitute Form W-2 with the same EIC information on the back of the employee's copy that is on Copy B of the IRS Form W-2.
- Notice 797, Possible Federal Tax Refund Due to the Earned Income Credit (FIC).
- Your written statement with the same wording as Notice 797.

If you give an employee a Form W-2 on time, no further notice is necessary if the Form W-2 has the required information about the EIC on the back of the employee's copy. If you give an employee a substitute Form W-2, but it does not have the required information, you

must notify the employee within 1 week of the date the substitute Form W-2 is given. If Form W-2 is required but is not given on time, you must give the employee Notice 797 or your written statement by the date Form W-2 is required to be given. If Form W-2 is not required, you must notify the employee by February 5, 2024.

You must hand the notice directly to the employee or send it by first-class mail to the employee's last known address. You will not meet the notification requirements by posting Notice 797 on an employee bulletin board or sending it through office mail. However, you may want to post the notice to help inform all employees of the EIC. You can download copies of the notice at www.irs.gov/FormsPubs. Or you can go to www.irs.gov/FormsPubs. Or you can

How Will My Employees Know if They Can Claim the EIC?

The basic requirements are covered in Notice 797. For more detailed information, the employee needs to see Pub. 596, Earned Income Credit (EIC), or the Instructions for Forms 1040 and 1040-SR.

How Do My Employees Claim the EIC?

Eligible employees claim the EIC on their 2023 tax return. Even an employee who has no tax withheld from wages and owes no tax may claim the EIC and ask for a refund, but they must file a tax return to do so. For example, if an employee has no tax withheld in 2023 and owes no tax but is eligible for a credit of \$800, they must file a 2023 tax return to get the \$800 refund.

Notice **1015** (Rev. 12-2023) Cat. No. 20599I

APPENDICES J – L INTENTIONALLY OMITTED

BACKGROUND AND RESOURCES: CALIFORNIA CHARITIES REGULATION

Page 9 of 2

There is a keen public interest in preventing misuse of charitable contributions. California's "Supervision of Trustees and Fundraisers for Charitable Purposes Act" regulates those raising and receiving charitable contributions. The "Nonprofit Integrity Act of 2004" (SB 1262, Chapter 919) tightened Charitable Purposes Act requirements for charitable organization administration and fundraising.

The Charitable Purposes Act rules cover California public benefit corporations, unincorporated associations, and trustee entities. They may include similar foreign corporations doing business or holding property in California. Generally, an organization is subject to the registration and reporting requirements of the Charitable Purposes Act if it is a California nonprofit public benefit corporation or is tax exempt under Internal Revenue Code § 501(c)(3), and not exempt from reporting under Government Code § 12583. Most educational institutions, hospitals, cemeteries, and religious organizations are exempt from Supervision of Trustees Act requirements.

Key new Charitable Purposes Act requirements affect executive compensation, fund-raising practices and documentation. Charities with over \$2 million of revenues (excluding grants and service-contract funds a governmental entity requires to be accounted for) have new audit requirements. Charities required to have audits must also establish an audit committee whose members have no material financial interest in any entity doing business with the charity.

Organizations or persons that receive or raise charitable contributions are likely to be subject to the Charitable Purposes Act. A Proposer on Los Angeles County contracts must determine if it is subject to the Charitable Purposes Act and certify either that:

- It is not presently subject to the Act, but will comply if later activities make it subject, or,
- If subject, it is currently in compliance.

RESOURCES

The following references to resources are offered to assist Proposers who engage in charitable contributions activities. Each Proposer, however, is ultimately responsible to research and determine its own legal obligations and properly complete its compliance certification (Exhibit 19).

In California, supervision of charities is the responsibility of the Attorney General, whose website, http://oag.ca.gov/ contains much information helpful to regulated charitable organizations.

1. LAWS AFFECTING NONPROFITS

The "Supervision of Trustees and Fundraisers for Charitable Purposes Act" is found at California Government Code §§ 12580 through 12599.7. Implementing regulations are found at Title 11, California Code of Regulations, §§ 300 through 312. In California, charitable solicitations ("advertising") are governed by Business & Professions Code §§ 17510 through 17510.95. Regulation of nonprofit corporations is found at Title 11, California Code of Regulations, §§ 999.1 through 999.5. (Amended regulations are pending.) Links to all of these rules are at: http://oag.ca.gov/charities/laws

BACKGROUND AND RESOURCES: CALIFORNIA CHARITIES REGULATION

Page 2 of 2

2. SUPPORT FOR NONPROFIT ORGANIZATIONS

Several organizations offer both complimentary and fee-based assistance to nonprofits, including in Los Angeles, the *Center for Nonprofit Management*, 606 S. Olive St #2450, Los Angeles, CA 90014 (213) 623-7080 http://www.cnmsocal.org/., and statewide, the *California Association of Nonprofits*, http://www.calnonprofits.org/. Both organizations' websites offer information about how to establish and manage a charitable organization.

The above information, including the organizations listed, provided under this subsection of this Appendix N is for informational purposes only. Nothing contained in this sub-section shall be construed as an endorsement by the County of Los Angeles of such organizations.

Page 1 of 4

- 2.206.010 Findings and declarations.
- 2.206.020 Definitions.
- 2.206.030 Applicability.
- 2.206.040 Required solicitation and Contract language.
- 2.206.050 Administration and compliance certification.
- 2.206.060 Exclusions/Exemptions.
- 2.206.070 Enforcement and remedies.
- 2.206.080 Severability.

2.206.010 Findings and declarations.

The Board of Supervisors finds that significant revenues are lost each year as a result of taxpayers who fail to pay their tax obligations on time. The delinquencies impose an economic burden upon the County and its taxpayers. Therefore, the Board of Supervisors establishes the goal of ensuring that individuals and businesses that benefit financially from the Contracts with the County fulfill their property tax obligation. (Ord. No. 2009-0026 § 1 (part), 2009.)

2.206.020 Definitions.

The following definitions shall be applicable to this chapter:

- A. "Contractor" shall mean any person, firm, corporation, partnership, or combination thereof, which submits a bid or proposal or enters into a Contract or agreement with the County.
- B. "County" shall mean the County of Los Angeles or any public entities for which the Board of Supervisors is the governing body.
- C. "County Property Taxes" shall mean any property tax obligation on the County's secured or unsecured roll; except for tax obligations on the secured roll with respect to property held by a Contractor in a trust or fiduciary capacity or otherwise not beneficially owned by the Contractor.
- D. "Department" shall mean the County department, entity, or organization responsible for the solicitation and/or administration of the Contract.
- E. "Default" shall mean any property tax obligation on the secured roll that has been deemed defaulted by operation of law pursuant to California Revenue and Taxation Code section 3436; or any property tax obligation on the unsecured roll that remains unpaid on the applicable delinquency date pursuant to California Revenue and Taxation Code section 2922; except for any property tax obligation dispute pending before the Assessment Appeals Board.
- F. "Solicitation" shall mean the County's process to obtain bids or proposals for goods and services.
- G. "Treasurer-Tax Collector" shall mean the Treasurer and Tax Collector of the County of Los Angeles. (Ord. No. 2009-0026 § 1 (part), 2009.)

Page 2 of 4

2.206.030 Applicability.

This chapter shall apply to all solicitations issued 60 days after the effective date of the ordinance codified in this chapter. This chapter shall also apply to all new, renewed, extended, and/or amended Contracts entered into 60 days after the effective date of the ordinance codified in this chapter. (Ord. No. 2009-0026 § 1 (part), 2009.)

2.206.040 Required solicitation and Contract language.

All solicitations and all new, renewed, extended, and/or amended Contracts shall contain language which:

- Requires any Contractor to keep County Property Taxes out of Default status at all times during the term of an awarded Contract;
- B. Provides that the failure of the Contractor to comply with the provisions in this chapter may prevent the Contractor from being awarded a new Contract; and
- C. Provides that the failure of the Contractor to comply with the provisions in this chapter may constitute a material breach of an existing Contract, and failure to cure the breach within 10 days of notice by the County by paying the outstanding County Property Tax or making payments in a manner agreed to and approved by the Treasurer-Tax Collector, may subject the Contract to suspension and/or termination. (Ord. No. 2009-0026 § 1 (part), 2009.)

2.206.050 Administration and compliance certification.

- A. The Treasurer-Tax Collector shall be responsible for the administration of this chapter. The Treasurer-Tax Collector shall, with the assistance of the Chief Executive Officer, Director of Internal Services, and County Counsel, issue written instructions on the implementation and ongoing administration of this chapter. Such instructions may provide for the delegation of functions to other departments.
- B. The Contractor shall be required to certify, at the time of submitting any bid or proposal to the County, or entering into any new Contract, or renewal, extension or amendment of an existing Contract with the County, that it is in compliance with this chapter is not in Default on any County Property Taxes or is current in payments due under any approved payment arrangement. (Ord. No. 2009-0026 § 1 (part), 2009.)

2.206.060 Exclusions/Exemptions.

- A. This chapter shall not apply to the following Contracts:
 - 1. Chief Executive Office delegated authority agreements under \$50,000;

Page 3 of 4

- 2. A Contract where federal or state law or a condition of a federal or state program mandates the use of a particular Contractor;
- 3. A purchase made through a state or federal Contract;
- 4. A Contract where state or federal monies are used to fund service related programs, including but not limited to voucher programs, foster care, or other social programs that provide immediate direct assistance;
- 5. Purchase orders under a master agreement, where the Contractor was certified at the time the master agreement was entered into and at any subsequent renewal, extension and/or amendment to the master agreement.
- 6. Purchase orders issued by Internal Services Department under \$100,000 that is not the result of a competitive bidding process.
- 7. Program agreements that utilize Board of Supervisors' discretionary funds;
- 8. National Contracts established for the purchase of equipment and supplies for and by the National Association of Counties, U.S. Communities Government Purchasing Alliance, or any similar related group purchasing organization;
- A monopoly purchase that is exclusive and proprietary to a specific manufacturer, distributor, reseller, and must match and inter-member with existing supplies, equipment or systems maintained by the County pursuant to the Los Angeles Purchasing Policy and Procedures Manual, section P-3700 or a successor provision;
- 10. A revolving fund (petty cash) purchase pursuant to the Los Angeles County Fiscal Manual, section 4.6.0 or a successor provision;
- 11. A purchase card purchase pursuant to the Los Angeles County Purchasing Policy and Procedures Manual, section P-2810 or a successor provision;
- A non-agreement purchase worth a value of less than \$5,000 pursuant to the Los Angeles County Purchasing Policy and Procedures Manual, section A-0300 or a successor provision; or
- A bona fide emergency purchase pursuant to the Los Angeles County Purchasing Policy and Procedures Manual section P-0900 or a successor provision;
- 14. Other Contracts for mission critical goods and/or services where the Board of Supervisors determines that an exemption is justified.
- B. Other laws. This chapter shall not be interpreted or applied to any Contractor in a manner inconsistent with the laws of the United States or California. (Ord. No. 2009-0026 § 1 (part), 2009.)

Page 4 of 4

2.206.070 Enforcement and remedies.

- A. The information furnished by each Contractor certifying that it is in compliance with this chapter shall be under penalty of perjury.
- B. No Contractor shall willfully and knowingly make a false statement certifying compliance with this chapter for the purpose of obtaining or retaining a County Contract.
- C. For Contractor's violation of any provision of this chapter, the County department head responsible for administering the Contract may do one or more of the following:
 - 1. Recommend to the Board of Supervisors the termination of the Contract; and/or,
 - 2. Pursuant to chapter 2.202, seek the debarment of the Contractor; and/or,
 - 3. Recommend to the Board of Supervisors that an exemption is justified pursuant to Section 2.206.060.A.14 of this chapter or payment deferral as provided pursuant to the California Revenue and Taxation Code. (Ord. No. 2009-0026 § 1 (part), 2009.)

2.206.080 Severability.

If any provision of this chapter is found invalid by a court of competent jurisdiction, the remaining provisions shall remain in full force and effect. (Ord. No. 2009-0026 § 1 (part), 2009.